# Western University Scholarship@Western

**Electronic Thesis and Dissertation Repository** 

7-11-2017 12:00 AM

# Batch and Fed Batch Cultivation and Harvesting of Nannochloropsis Gaditana for Environmental Applications

Roopa P. Devasya, The University of Western Ontario

Supervisor: Dr Amarjeet Bassi, *The University of Western Ontario* A thesis submitted in partial fulfillment of the requirements for the Doctor of Philosophy degree in Chemical and Biochemical Engineering © Roopa P. Devasya 2017

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Chemical Engineering Commons

#### **Recommended Citation**

Devasya, Roopa P., "Batch and Fed Batch Cultivation and Harvesting of Nannochloropsis Gaditana for Environmental Applications" (2017). *Electronic Thesis and Dissertation Repository*. 4670. https://ir.lib.uwo.ca/etd/4670

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

#### Abstract

The exhaustion of fossil fuels and climate change are two significant issues of our times. Microalgae are eukaryotic phototrophs or prokaryotic cyanobacteria (blue green algae), which are able to capture  $CO_2$ , and thus, can mitigate these challenges. In addition, some microalgae can produce lipids suitable for biodiesel. The potential application of microalgae biomass for biofuel production is a clean and sustainable approach to replace fossil fuels. An important consideration for reducing the cost of biofuel is the economical production of algal biomass with high lipid yields. In this study, the marine microalgae *Nannochloropsis gaditan* were investigated for biomass and lipid production using two wastewater streams, i.e., road salt run-off and vegetable greenhouse industry effluents as the growth media using batch or fed-batch cultivation. The recovery of biomass was also investigated using coagulants or pH change.

In the first phase of the research, the effect of salinity on biomass concentration and lipid production was examined in batch culture. Biomass yields of 1-2 g/L of culture were obtained with the maximum lipid of 22% (g/g) biomass in the road salt media. The crude road salt media provided all the essential micronutrients needed for algal cultivation. The fatty acid composition analysis of the obtained lipid indicated C16 and C18 as the major components (~ 45% of FAME) further confirmed the suitability of biomass generated for biofuel feed-stock. This study has established that the use of road salt containing nitrate and phosphate nutrients will support the growth marine micro algae for remediation of waste water system that are the concern at winter prevalent regions.

The controlled feeding of nitrates in fed-batch cultures can be applied to promote growth and lipid production in microalgae. The fed-batch cultivation system was, therefore, next applied to provide nutrition sufficient, depletion and repletion conditions. In this study, *N. gaditana* was cultured in batch and fed-batch bioreactors. Three modes of nutrient feeding i.e., pulse; continuous and staged were applied under two different light regimes for autotrophic cultivation. Higher nitrates levels negatively affected the lipid production. Increasing the light intensity was found to increase the lipid content in the microalgae for all three fed-batch feeding modes. A maximum of 58.3% lipid fraction of algal dry weight was achieved when

using pulse-fed cultures at an illumination of 200µmol photons/m<sup>2</sup>/and 10mg/d nitrate feeding. This condition also resulted in the maximum lipid productivity of 44.6 mg /L/d. The fatty acid composition of the lipids consisted predominantly of long chain fatty acids (C: 16 and C: 18) and accounted for 70 % of the overall FAME. The other two feeding modes (continuous and staged) were not as effective for lipid and biomass production. This study demonstrates the applicability of pulsed feeding strategies in fed-batch cultures.

In the final phase of the study, the coagulation/flocculation methods were applied to harvest the biomass from the growth media. Cell harvesting was investigated using two approaches: coagulant chemical addition and pH change. The pH value of 11 was identified as the threshold pH for the alkaline flocculation of *Nannochloropsis gaditana* cells in the road salt growth media. The harvesting efficiency increased to 89% at the pH 11 and above. However, the coagulant/flocculant ferric chloride provided the best flocculation with 95% harvesting efficiency and the concentration factor of 23.5. The bio-coagulant chitosan was not effective to coagulate the cells within the 60 minutes of settling time.

This study primarily resulted in three main conclusions. First, the phytoremediation of road salt run-off generates the biomass suitable for biodiesel production. Secondly, the application pulsed fed-batch cultivation with nitrate feeding and increased light intensity simultaneously improves the biomass and lipid production. Finally, the alkaline induced flocculation is an effective harvesting method for the recovery of *Nannochloropsis gaditana* from road salt media. The study provides new directions for the integration of biofuel production with greenhouse vegetable effluent and road salt run-off and the application of marine microalgae for the treatment of hyper saline wastewater.

# Keywords

Biodiesel

Biofuel

Biomass

Coagulation

Concentration factor (CF)

Fatty acid methyl esters (FAME)

Fed-batch cultivation

Flocculation

Greenhouse effluents

Harvesting efficiency (HE)

Marine microalgae

Nannochloropsis gaditana

Phycoremediation

Pulsed feeding

Road salt run-off

#### **Co-Authorship Statement**

The development of these chapters and manuscripts for submission were written and coauthored, the extent of the collaboration of the co-authors is stated below.

The papers that are being prepared or submitted are given below.

Chapter-4

Paper 1	Growth kinetics of Nannochloropsis gaditana	
	on road salt media	
Current status	Submitted to the Journal of Environmental	
	technology	
Roopa Devasya: Exp	berimental design, laboratory work, analysis of	
results, and		
Paper writing.		
Amarjeet Bassi: Tech	nical and theoretical advisor and corrections of	
several drafts and		

#### Chapter- 5

Paper 3	Investigation of feeding strategies for Fed batch cultivation systems for the biomass and lipid production from <i>Nannochloropsis</i> gaditana	
Current status	Submitted to the Journal: Biotechnology	
Current status	Submitted to the Journal. Diotechnology	
	progress	
Roopa Devasya: Exp	erimental design, laboratory work, analysis of	
results, and paper writin	lg.	
Amarieet Bassi: Technical and theoretical advisor and corrections of		
several drafts and		

Chapter- 6

Paper 3	Investigation of alkaline flocculation for Nannochloropsis gaditana in the phycoremediated road salt run-off media
Current status	Prepared for the submission in progress
Roopa Devasya: Exp results, and paper writin Amarjeet Bassi: Techn several drafts and	erimental design, laboratory work, analysis of ag. nical and theoretical advisor and corrections of

#### Acknowledgments

Foremost, I would like to express my sincere gratitude to my supervisor Professor Amarjeet Bassi for his continuous support, advice, trust, and motivation. Thanks for giving me this enriching opportunity. I also want to thank my Ph.D. advisory committee for all the guidance and good advice received. Thanks to all the teachers and professors who have contributed to my professional development. They all have a special place in my life.

My sincere thanks to Parmalat Canada and to my director Maria Pepe Francis, for providing the opportunity for continue my learning process. My sincere respect and thanks to Peter Van Esch, who rekindled my desire and showed the path to take the adventurous journey in the academic world.

Thanks to my laboratory team, Shreyas, Mengyue, Claudio, Ahmed, Essam, Teresa, Yulin, Valerie and Leila for the good times, and all the invaluable things I have learned from you. Thanks to Erin Johnson, Brian Dennis and Souheil Afara for making research easier by having practical solutions to everyday problems.

I am also very thankful to my friends who have contributed immensely to my personal development. They were always there for me when I needed them the most.

Thanks to my hubby Vijay and daughters Tanvi and Raashi. You are my strength, my biggest support and motivation. Without your sacrifice and support I would not have reached to this stage.

Thanks with love to my mother, my father, my brother, and my sisters. Your love has always been with me despite the thousands of miles between us. This would not be possible without you.

## **Table of Contents**

Abstract	i
Co-Authorship Statement	iv
Acknowledgments	v
Table of Contents	vi
List of Tables	xi
List of Figures	xiii
Chapter 1	
1 Introduction	
1.1 Significance of the problem	
1.2 Research goals and structure	
1.3 Overall and Specific Objectives	
1.3.1 Specific Objectives	
1.3.2 Specific Objective 1	
1.3.3 Specific objective 2	
1.3.4 Specific objective 3	
1.3.5 Specific objective 4	
1.4 Thesis structure	5
1.5 Major contributions	
1.6 Novelty	6
Chapter 2	7
2 Literature review	7
2.1 Introduction	7
2.2 Microalgae classification, biochemistry and production systems	
2.3 Industrial application of microalgae	

	2.3.1	Application of microalgae for biofuel production	13
	2.3.2	Mechanisms of lipid accumulation in microalgae	14
	2.3.3	Conversion of algal lipids into biofuels	16
	2.3.4	Application of microalgae for wastewater treatment	18
	2.3.5	Road salt runoff	20
	2.3.6	Hydroponic greenhouse effluents	21
	2.3.7	Phycoremediation	22
	2.3.8	Microalgae based wastewater treatment technologies	23
2.4	Microa	lgae production systems	24
	2.4.1	Algal cultivation methods	25
	2.4.2	Nutrimental and physico-chemical stresses	25
	2.4.3	Nutrimental factors: Nitrogen	25
	2.4.4	Carbon	27
	2.4.5	Phosphorus	27
	2.4.6	Sulphur	28
	2.4.7	Physico-chemical factors: salinity	28
	2.4.8	Physico-chemical factors: light intensities	29
2.5	Culturi	ing systems	31
	2.5.1	Culture systems: batch	31
	2.5.2	Culture systems: continues and fed-batch	32
	2.5.3	Nutrient feeding methods	33
2.6	Harves	sting techniques	33
	2.6.1	Common harvesting techniques	35
	2.6.2	Centrifugation	35
	2.6.3	Filtration	36
	2.6.4	Sedimentation	37

		2.6.5	Flotation	. 37
		2.6.6	Coagulation-Flocculation	. 38
		2.6.7	Flocculation phenomenon in microalgae	. 39
		2.6.8	Flocculants for micro algal harvesting	. 39
		2.6.9	Flocculation by organic flocculants	. 41
		2.6.10	Flocculation by pH change	. 43
		2.6.11	Mechanism of pH induced flocculation	. 43
		2.6.12	Bio/Auto flocculation	. 43
		2.6.13	Microalgae/fungal co-palletization	. 44
		2.6.14	Magnetic flocculation	. 44
		2.6.15	Electrical based harvesting	. 45
		2.6.16	Flocculation by electromagnetic particles	. 45
	2.7	Conclu	issions	. 45
Cł	napte	er 3		. 46
3	Mat	erials a	nd Methods	. 46
	3.1	Growtl	h media	. 46
	3.2	Algal c	culture	. 46
	3.3	Algal g	growth media and cultivation	. 46
		3.3.1	Microalgae growth media for the fed batch studies	. 47
		3.3.2	Experimental set up for the batch cultures	. 48
		3.3.3	Experimental set up and cultivation conditions for fed-batch cultures	. 49
		3.3.4	Flocculation experiments	. 49
	3.4	Analyt	ical Methods	. 50
		3.4.1	Cell count and biomass concentration quantification	. 50
		3.4.2	Nitrate measurements	. 51
		343	Chlorophyll quantification	. 51

		3.4.4	Lipid quantification
		3.4.5	GC-FAME analysis
	3.5	Detern	nination of flocculation efficiency
	3.6	Statist	ical methods
C	hapte	er 4	
4	Inv	estigatio	on of phyco-remediation of road salt run-off with marine <i>N. gaditana</i> 55
	4.1	Introdu	uction
	4.2	Result	s and Discussion
		4.2.1	Effect of media and salt concentration on growth kinetics of N.gaditana 58
		4.2.2	Effect of type of media and its salt concentration on nitrate consumption of <i>N. gaditana</i>
		4.2.3	Specific growth rates of <i>N.gaditana</i> at various media salt concentration. 67
		4.2.4	Effect of media, salt and initial media nitrate concentration on biomass production
		4.2.5	Effect of media, salt and nitrate concentration on lipid production and fatty acid profile of <i>N. gaditana</i>
		4.2.6	Lipid productivity and salinity70
		4.2.7	Effect media, salt and nitrate concentration on fatty Acid profile72
	4.3	Conclu	usions
C	hapte	er 5	
5	Inv N.g	estigatio <i>aditana</i>	on of three fed batch feeding strategies for biomass and lipid production of in Fed-batch cultivation
	5.1	Introdu	uction
	5.2	Result	s and Discussion
		5.2.1	Effect of cultivation methods and feeding rates nitrate conditions for media on nitrate uptake and biomass
		5.2.2	Effect of culture conditions on biomass
		5.2.3	Effect of feeding on nitrogen to cell conversion factor 100

	5.2.4	Effect of cultivation methods on lipid production	104
	5.3 Concl	usions	110
C	hapter 6		111
6	Investigati media by f	ion of harvesting efficiency of <i>N. gaditan</i> a from phyco-remediated reflocculation and settling	coad salt
	6.1 Introd	luction	112
	6.2 Resul	ts and Discussion	115
	6.2.1	Effect of pH on sedimentation of <i>N.gaditana</i> cells with alkaline flocculation	115
	6.2.2	Harvesting efficiency of <i>N. gaditana</i> cells with ferric chloride	123
	6.2.3	Harvesting efficiency of <i>N. gaditana</i> cells with chitosan	126
	6.2.4	Concentration factors of algal suspension	129
	6.2.5	Comparison of concentration factors	132
	6.3 Concl	usions	133
C	hapter 7		135
7	Conclusio	ns and recommendations	135
	7.1 Concl	usions	135
	7.2 Recor	nmendations	136
R	eferences		138
Appendices			
C	urriculum V	<sup>7</sup> itae	170

# List of Tables

Table 2-1 Biochemical composition of microalgae    11
Table 2-2 Lipid class content as a % of total lipid    12
Table 2-3 Comparisons of sources of biodiesel    13
Table 2-4 Oil content of some microalgae    14
Table 2-5    ASTM Biodiesel Standard D 6751    17
Table 2-6 Microalgae growth and lipid productivity in wastewater
Table 2-7    Essentional algal nutrients    26
Table 2-8 Characteristics of the major pigments in microalgae
Table 2-9 Optimal light intensity and light/dark cycles for different species
Table 2-10 Comparison of micro algal harvesting methods    34
Table 2-11 Energy consumption of different flotation systems for microalgae harvesting         [ <sup>131–133]</sup>
Table 2-12Comparison of inorganic and organic flocculants
Table 2-13 Overview of highlights and limitations of flocculation
Table 2-14 Comparison of different chemical flocculants for microalgae harvesting
Table 4-1 Elemental concentration of reef and road salt media analyzed by the ICP- MS 58
Table 4-2 Biomass and lipid to sodium conversation
Table 5-1 Biomass to substrate nitrogen conversion rates in the batch and fed batch cultivations at light 200 $\mu$ mol photons/ m2 /s

Table 5-2 Specific growth rate and Biomass concentration in various cultivation methods 102
Table 5-3 Lipid and biomass production in various cultivation stages in the batch culture with
starting nitrate concentration of 100 mg/L 104
Table 5-4 Comparison of difference cultivation regimes for biomass (X), Lipid content (P),
volumetric lipid concentration (Pvol.) and lipid productivity (QP) and yield per g of nitrate
Table 6-1 Comparison of Concentration factor dry weight method and flocculation volume
reduction methods

# List of Figures

Figure 2:1 Simplified diagram of lipid synthesis process in the microalgae cells <sup>47</sup> 15
Figure 2:2 Trans esterification reaction of triacylglycerides <sup>[49]</sup>
Figure 4:1 Growth curves by the cell count of the <i>N.gaditana</i> in road salt media 60
Figure 4:2 Growth curves by the cell count of the <i>N.gaditana</i> in reef salt media
Figure 4:3 Cellular concentration of chlorophyll "a" in biomass of road salt media 62
Figure 4:4 Cellular concentration of chlorophyll "a" in biomass of reef salt media 62
Figure 4:5 Concentration of Nitrate -Nitrogen versus cultivation in road salt media
Figure 4:6 Concentration of Nitrate -Nitrogen versus cultivation in reef salt media
Figure 4:7 Biomass dry weight of <i>N. gaditan</i> a in road and reef salt growth media
Figure 4:8 Lipid concentrations (%w/w) of <i>N. gaditana</i> in the biomass
Figure 4:9 Fatty acid profile of FAME in lipids of <i>N. gaditana</i>
Figure 5:1 Media nitrate concentration in batch culture media at 100 $\mu$ mol photons/ m <sup>2</sup> /s light intensity
Figure 5:2 Media nitrate concentration batch culture media at 200 µmol photons/ m <sup>2</sup> /s light intensity
Fire 5:3 Biomass concentrations of 100- $\mu$ mol photons/m <sup>2</sup> /s light intensity in batch culture. 82
Figure 5:4 Biomass concentrations of 200 µmol photons/m <sup>2</sup> /s light intensity in batch culture
Figure 5:5 Nitrate concentrations in pulse feeding fed batch culture at 100 $\mu$ mol photons/ m <sup>2</sup> /s light intensity

Figure 5:6 Nitrate concentrations in pulse feeding fed batch culture at 200 $\mu$ mol photons/ m <sup>2</sup>
/s light intensity
Figure 5:7 Biomass concentrations in pulse feeding at 100µmol photons/m <sup>2</sup> /s light intensity
Figure 5:8 Biomass concentration in pulse feeding fed batch culture at 200µmol photons/m <sup>2</sup> /s light intensity
Figure 5:9 Growth media nitrate concentration in continuous fed FB culture at 100 $\mu$ mol photons/ m <sup>2</sup> /s light intensity
Figure 5:10 Nitrate concentration during the continuous fed FB batch culture at 200 $\mu$ mol photons/ m <sup>2</sup> /s light intensity
Figure 5:11 Biomass concentration in continuous fed FB batch culture at 100 µmol photons/m <sup>2</sup> /s light intensity
Figure 5:12 Biomass concentration in continuous fed FB batch culture at 200 $\mu$ mol photons/m <sup>2</sup> /s light intensity
Figure 5:13 Media nitrate concentration in stage fed FB batch culture at 100 µmol photons/m <sup>2</sup> /s light intensity
Figure 5:14 Media nitrate concentration in stage fed FB batch culture at 200 µmol photons/m <sup>2</sup> /s light intensity
Figure 5:15 Biomass concentration in stage fed batch FB culture at 100-µmol photons/m <sup>2</sup> /s light intensity
Figure 5:16 Biomass concentration in stage fed FB culture at 200-µmol photons/m <sup>2</sup> /s light intensity
5:17 Percentage of fatty acids in FAME of the lipids 109
Figure 6:1 Harvesting efficiency versus the pH of media at various settling time

Figure 6:2 Harvesting efficiency as a function of settling time at various pH of media 117
Figure 6:3 Harvesting efficiency of <i>N.gaditana</i> as a function of time in the pH ranges (9 to
10.5)
Figure 6:4 Harvesting efficiency of <i>N. gaditana</i> as a function of time in the pH ranges (11 to
12.5)
Figure 6:5 Harvesting efficiencies at various settling time versus ferric chloride dosage 124
Figure 6:6 Harvesting efficiencies with ferric chloride and change in the media pH 125
Figure 6:7 Harvesting efficiency of chitosan in acetic and citric acid solution at various
dosage rates
Figure 6:8 Concentration factor by pH induced flocculation at 4 sedimentation times 130
Figure 6:9 Concentration factor with ferric chloride at 3 sedimentation times 131
Figure 6:10 Concentration factor with chitosan at 60 minutes of settling

## List of Appendices:

ppendix: A. Growth kinetics of Nannochloropsis gaditana in real vegetable greenhouse
fluents
ppendix: B Elemental concentration of reef and roads salt media
ppendix: C Experimental set up
ppendix: D Photo of the jar experiments

# Chapter 1

## **1** Introduction

The biomass of microalgae contains large amounts of lipids, which have strong potential for the production of renewable biodiesel. While microalgae have the characteristic ability to accumulate lipids photoautotrophically using simple nutrients such as nitrates, phosphates and carbon dioxide, currently, their large-scale cultivation for production of lipids is not economically competitive in particular due to the cost of nutrients. Using wastewaters as a nutrient source can potentially provide for more feasible alternatives for lipid production from microalgae. The removal of nitrogen and phosphorus from wastewater systems using microalgae is referred by the term "phycoremediation" and the process bio transforms the pollutants to biomass with lipids. Thus, phycoremediation offers the opportunity for integrating wastewater treatment with the biofuel production.

This study combined two important wastewater streams, found in Ontario, Canada, namely road salt run-off and vegetable greenhouse industry effluent waters, as the nutrient (nitrate and phosphate) source and cultivation media for the culturing of marine microalgae *Nannochloropsis gaditana* (*N. gaditana*). The manipulation of cultivation conditions and process operation in batch and fed-batch mode was investigated for the improvement of both biomass and lipid production to enhance the lipid productivity. The alkaline and chemical based coagulation/flocculation methods were further investigated to evaluate the harvesting efficiencies for the recovery to the biomass from the growth media.

## **1.1** Significance of the problem

Road salt run-off and effluent waters from the vegetable greenhouse sector are two examples of wastewaters in Southwestern Ontario region in Canada. A number of commercial vegetable greenhouses populate this region. These hydroponic based industries use liquid fertilizers and generate effluent with residual nitrogen and phosphorus that may exceed provincial water quality guidelines, e.g., an average nitrate to 90.3 mg/L against the standard 3 mg/L, phosphorus to 180 mg/L against standard 0.3mg/l, potassium to 180 mg/L.

(https://www.ontario.ca/page/greenhouse-wastewater-monitoring-project-2010-and-2011 accessed on July 2017).

Road salt is applied as a deicing agent on Canadian roads during the winter. The saline run off increases the alkalinity and chloride levels of the soil and also further deteriorates the water quality of receiving bodies of water such as lakes, rivers and ponds. The low levels of organic carbon and elevated levels of nitrate and phosphates in the effluents pose challenges to the traditional wastewater treatment approach. The disposal of effluents to the water systems causes eutrophication and the production of harmful algal blooms (HABs). Currently there is no treatment method carried out for road salt run-off. Thus this is becoming an effluent of increasing concern for Canadian waters.

Microalgae cultivation in road salt effluent is a potential option for environmental sustainability in Southwestern Ontario. Although the nitrate concentration of the road salt effluent is low, mixing the stream with vegetable greenhouse industry effluents is proposed as a feasible alternative in this study for micro-algal cultivation. The proposed approach can also reduce the water footprint required to produce algae for biofuel production. In particular, *N. gaditana*, a halophilic micro algal species which grows on salt media and produces high levels of lipids was investigated in this study.

### **1.2** Research goals and structure

The research was divided into three main phases as depicted in Figure 1.1. In the first phase, studies were conducted on the investigation of micro algal biomass production in varying road salt and nitrate concentrations and the effect on biomass; lipid production and fatty acid profiles were evaluated.

In the second phase, studies were conducted in both batch and fed batch mode. The fed batch cultivation strategy was investigated to increase biomass and lipid production and to incorporate the nutrients from greenhouse effluents by investigating the effect of feeding rate and light intensity on the efficiency of production of biomass with the target to optimize the high value metabolites lipid.

Finally, the third phase of the study focused on harvesting the micro algal biomass and downstream wastewater treatment. The chemical coagulants such as ferric chloride and chitosan at various concentrations were investigated. In addition, pH induced auto-flocculation using the NaOH was also investigated.



Fig 1.1 General scheme of the thesis structure

## **1.3 Overall and Specific Objectives**

The overall objective and specific objectives are given below.

The overall objective of this study was to demonstrate the applicability of phycoremediation using the marine microalgae *N. gaditana* for the treatment of the effluent streams: road salt run off and greenhouse effluent and to evaluate lipid production.

#### 1.3.1 Specific Objectives

The main objective is achieved by the following four sub objectives.

#### 1.3.2 **Specific Objective 1**

Phycoremediation of road salt run off using halophilic marine microalgae *N. gaditana*. The effects of salt and nitrate concentration on biomass, lipid productivity and chlorophyll content were studied. The suitability of the growth media for the marine microalgae cultivation was evaluated by compositional analysis and micro algal growth kinetics. The effect of varied salt concentration as a stress factor for the lipid and nitrate for biomass production were studied. The fatty acid profiles of the lipid were investigated for its suitability to biodiesel application.

#### **1.3.3** Specific objective 2

The fed-batch cultivation was investigated to control the supply of nitrate by the nutrient feeding. The feeding of nutrients nitrate using three different feeding regimes of pulsed, continuous and stage was investigated to evaluate its effect on biomass production and to enhance the lipid productivity.

#### 1.3.4 Specific objective 3

The effect of two light intensities (100 and 200  $\mu$ mol /m<sup>2</sup>/s) on the biomass and lipid production in the fed batch cultivation system was investigated on the above feeding regimes.

Two levels of light intensity and light/ dark cycle were studied to improve the photosynthetic ability in order to increase the biomass and lipid production system in the phycoremediation treatment method systems of road salt runoff.

#### 1.3.5 **Specific objective 4**

The coagulation and flocculation as a unit operation for harvesting of microalgae from the phycoremediation of effluent streams was investigated.

The effect of the dosages of ferric chloride and chitosan coagulation flocculation agents and media pH was explored for to harvest the biomass of *N. gaditana* generated in the phycoremediation studies.

#### **1.4** Thesis structure

The thesis is divided into 6 chapters: chapter 1 is an introduction to the research and thesis structure, chapter 2 covers the literature review and gives the general background on road salt run off, greenhouse effluent, concept of phycoremediation and its relation to biofuels production from microalgae. The cultivation strategies and harvesting of microalgae are reviewed. Chapter 3 details the material and methods and the experimental set up. Chapter 4 present results on the first phase of experiments regarding the suitability of the media and its effect of salinity on chlorophyll content, biomass and lipid production for the phycoremediation. Chapter 5 describes the effect of feeding rate and light irradiation on biomass and lipid production of *N.gaditan* in fed-batch cultivation system. Chapter 6 shows the results of the flocculation experiments for the development of efficient harvesting of microalgae using flocculation. Finally, chapter7 summarizes the conclusions of the study and provides some recommendations for future work.

## **1.5** Major contributions

The major contributions of this study are:

Microalgae were cultivated in batch on real effluent streams from road salt run-off and vegetable greenhouse effluents. These studies were used to determine a simulated media composition for further studies.

For the first time, fed-batch cultivation was carried out using three different feeding strategies namely, pulsed, continuous and stage feeding. The applications of operational conditions like the light are efficient to improve the lipid production in combination with the repeated feeding of nitrates to culture systems.

The application of coagulation/flocculation method increases the viability of the rapid recovery of the biomass from the large-scale cultivation systems otherwise it was considered as the bottleneck in the downstream processing. The flocculation using the alkaline method identifies the possibility

of the simple operating methods to pre-concentrate the cells without the application of the toxic chemical flocculants.

## 1.6 Novelty

The thesis made the following novel contributions:

- For the first time, growth media made up of road salt run-off and vegetable greenhouse effluents was applied to cultivate microalgae and evaluate lipid productivity and fatty acid profiles for the production of biodiesel.
- Batch studies were carried out to evaluate the effect of salt concentrations of road and reef salt in growth media on microalgal growth, lipid production and fatty acid compositions were measured.
- Different feeding regimes were used in a fed batch mode and pulse feeding was identified as an appropriate feeding strategy.
- Simultaneous improvement of biomass and lipid production was obtained using the pulse feeding regime with the controlled nitrate supply techniques for the utilization of the nutrients from effluents streams removed the restriction of lipid production at the cost of biomass.
- Finally, effective harvesting of micro algal cells by flocculants like ferric chloride or alkaline pH change was carried out for the harvesting and concentration of marine microalgae in spent media.

## Chapter 2

## 2 Literature review

Microalgae based biofuel is a potential renewable, carbon neutral green fuel which could address some of the growing needs of energy as the world moves to reduce over-dependence on finite sources of fossil fuels. However, the current available technology is considered unfeasible for the commercial production of microalgae to biofuels due to a competitive, economic disadvantage with lower priced fossil fuels. This chapter describes the background related to the production of microalgae on wastewaters for biodiesel and remediation of nutrients. These approaches are being investigated in many research laboratories to find effective strategies, which could make micro algal processing more competitive by improving both the lipid productivity and efficiency of harvesting of biomass.

#### 2.1 Introduction

A growing demand of energy and over dependence on finite source of fossil fuels and their negative environmental influence including greenhouse gas (GHG) emissions has raised the concerns to seek alternative renewable sources of energy. Currently about 90% of the global energy demand is supplied by fossil fuels. Many countries are developing strategies to reduce dependence on fossil fuels. Previously, the US department of energy (DOE) established the goal for 2030 to use 30% transport fuel or 60 billion gallons per year from renewable sources<sup>1</sup>. Biofuels produced from lipid-based biomass of several food crops and oil seeds are considered as the first generation biofuels. Increasing demands of arable land and water led to the controversy of food versus fuel and perceived potential problems to the global food supply and imbalance of ecosystem. Microalgae are considered the third generation feed stock following the second-generation non-food feed stocks (waste oil, and non-food -lignocelluloses)<sup>2</sup>.

Some species of microalgae accumulate up to 70% lipids in their cell body<sup>3</sup> and have the photosynthesis rates of 50 times higher than the terrestrial plants. Thus, microalgae offer greater advantages over the other feedstocks. Their high growth rate and environmental tolerance to a wide range of cultivation conditions is a potential advantage. Their ability to grow in non-arable land including on wastewater and carbon dioxide (CO<sub>2</sub>) from flue gases and utilize sunlight energy makes their cultivation economical attractive<sup>4</sup>. From an environmental stand-point,

microalgae also can remediate the nutrients from municipal, industrial and agricultural wastewater and could effectively mitigate the problems associated with water pollution and  $CO_2$  emission<sup>5</sup>. The use of biodiesel produced from micro algae formed by glycerol-esterification of fatty acids extracted from the lipid minimizes sulfur emissions contaminants compared to the petroleum based diesel<sup>6</sup>.

The micro-algal biofuel production consists of the upstream process of biomass generation and down process of lipid extraction and conversion. The main upstream production system contains algal species, cultivation and biomass separation systems. The commercial production of biofuel faces challenges due to low productivity and high cost associated with cultivation, harvesting and extraction techniques. The cost of cultivation accounts at least 25% and harvesting around 30% of the total cost of biomass production<sup>7</sup>. In one study, the production cost of biofuel feedstock was estimated to be 10.87-13.32 US\$ per gallon which was considered economically not feasible to the price of fossil fuel<sup>8</sup>.

The application of cost effective and efficient methods is therefore essential for achieving a viable process to overcome the challenges for the sustainable commercial production of microalgae to replace the dependence of fossil fuel. The technological advancements are critical to increase the efficiency of cultivation, harvesting, pretreatment, lipid extraction, and trans-esterification subsystems to reduce the overall production cost<sup>9</sup>.

Growth rate, lipid content and productivity are the main parameters, which determine the potential of micro-algal strains as biodiesel feed stock. Other performance factors accounting for selection of microalgae strain for the suitability to produce biodiesel depends on the lipid content of minimum 20% (w/w) and productivity of 40 mg/L/day <sup>10</sup>.

The lipid fraction of biomass represents the triacylglycerides (TAG). The success of biodiesel production depends on the content of TAG and the biomass productivity of microalgal species. The quality and quantity of biodiesel are influenced by the free fatty acid contents and composition in the lipids. These parameters vary between the species of the microalgae. The excessive saturation of fatty acids forms tar by cross linking of fatty acids and increase in polysaturation decreases the oxidative stability of biodiesel<sup>11</sup>. Genetics of the strains, various

culture conditions, the nutrients, light and environmental stresses are used to improve the quality and quantity of lipids in microalgae <sup>12</sup>.

Previous studies on life cycle assessment of biofuel production from microalgae have evaluated the barriers and noted some of the key issues in the cultivation systems. These include:

- Photosynthesis rate of the species,
- Nutrient dependence on biomass productivity
- Nutrient independent biosynthesis and regulation of lipid production systems

Lipid synthesis is improved by selection of algal strains by screening and genetic technology. The biomass production is improved by the development and design and engineered system for large-scale cultivation by suitable nutrient supply and light for culture systems to enhance the photosynthesis rates. The micro algal species or strains with environmental tolerance to high salinity, extreme pH, temperatures and high light intensities, production of by-products with commercial values, and rapid growth in the photo bioreactors are considered favourable for the biodiesel production<sup>13</sup>.

Fresh water supply and nutrients like nitrogen and phosphate are essential to algal production systems but are the primary contributors to the cost of the generation of biomass. Nutrient nitrogen sources alone cost around 80% of the overall nutrient supply<sup>14</sup>. Microalgae are aquatic species and require substantial amount of water to grow and proliferate. Nearly one metric ton of water is needed per kg of micro algal biomass production with a water loss by photosynthesis of 5–10 kg per 1.0 kg of dry micro algal biomass <sup>15</sup>.

Previously, various municipal, industrial and agricultural wastewaters containing nitrogen and phosphate have been investigated as an alternative and inexpensive sources of nutrients and water  $^{(16-18)}$ . However, the dilute concentration of biomass (<2g/L) produced requires the processing of large volumes of culture media could lead to inefficient downstream harvesting operations and processing using conventional approaches such as centrifugation can be prohibitively expensive at large scale. The main possible strategies proposed to reduce the overall cost of biomass for the commercial large-scale cultivation can be attained by, 1) cultivation in suitable nutrient rich wastewater and/or utilization of CO<sub>2</sub> from flue gases, 2) improvement of biomass and bio-refinery

products and 3) development and study of low cost and high efficiency recovery processes for micro algal biomass.

Cultivation in wastewater reduces the cost of nutrients and footprints of fresh water demands. Microalgae have the ability to grow in varied water sources like seawater, brackish and other industrial and domestic wastewater containing nitrogen and phosphorous which would potentially provide the necessary nutrients for large scale cultivation. This strategy integrates the feasibility of wastewater with microalgae cultivation as a source of nutrients for microalgae growth as well as effective remediation technique for the wastewater reduces the cost of the treatment and the generation of biomass.

Improvement of biomass and bio-refinery products approach include the use of the biomass as feedstock for diverse products such as fine chemicals for supplements of nutraceuticals and cosmetic industry <sup>(19–22)</sup>. The production of high value micro algal biochemical compounds such as polyunsaturated fatty acids, docosahexaenoic acid (DHA) and pigments (astaxanthin) have uses in nutraceuticals, medical and cosmetic products that in addition to the use of lipid for the biofuel balances the production cost of the biomass<sup>23</sup>. The recovery of micro algal biomass from the culture media is difficult due to the variety of microalgae species with different properties such a size, charge and low density. Highly efficient, low energy processes for the separation of biomass with the generation of recyclable media reduces the cost of harvesting and growth media. Centrifugation, filtration and sedimentation are energy intensive process accounts 20-30% of the total cost of production <sup>24</sup>. Approaches like auto-flocculation, pH induced flocculation and gravity sedimentation are considered more feasible for the separation of biomass from wastewater based algal cultivation system and require further investigations <sup>25</sup>.

## 2.2 Microalgae classification, biochemistry and production systems

Microalgae are unicellular microscopic, polyphyletic, photosynthetic, oxygen evolving microorganisms. One of the oldest living organisms on the planet, they can exist individually or in chains or groups that form the basis of the food chain in oceans and rivers as plankton <sup>23</sup>. These microorganisms are responsible for achieving the life promoting current composition of atmospheric oxygen. Microalgae are characterized by being adapted to live in an extremely broad spectrum of environment conditions due to their unicellular or simple multicellular structure and minimum growth requirements <sup>26</sup>.

There are about 200,000-800,000 species in existence with more than 1000 types of micro algal strains investigated for biodiesel only<sup>5,27</sup>. Microalgae are taxonomically classified based on the pigmentation, life cycle, and basic cellular structure. They are mostly eukaryotic although the prokaryotic cyanobacteria (blue-green algae) are frequently classified as algae. Algal biomass composition contains three main components carbohydrate, protein and lipids at various concentrations with around 40% of its mass is fatty acids and are extracted and converted into biodiesel. The biochemical compositions of various microalgae are used as basis for its application into many industrial products. The composition of carbohydrate, protein and lipids of various microalgae are shown in the below table 2.1.

Microalgae	Protein (%DW)	Carbohydrates (%DW)	Lipid (%DW)
Scenedesmus obliquus	50–56	10–17	12–14
Scenedesmus quadricauda	47	-	1.9
Scenedesmus dimorphus	8 - 18	21–52	16–40
Chlamydomonas rheinhardii	48	17	21
Chlorella vulgaris	51–58	12–17	14–22
Chlorella pyrenoidosa	57	26	2
Spirogyra sp.	6–20	33–64	11–21
Dunaliella bioculata	49	4	8
Dunaliella salina	57	32	6
Euglena gracilis	39–61	14–18	14–20
Prymnesium parvum	28–45	25–33	22–39
Tetraselmis maculata	52	15	3
Porphyridium cruentum	28–39	40–57	9–14
Spirulina platensis	46–63	8–14	4–9
Spirulina maxima	60–71	13–16	6–7
Synechoccus sp.	63	15	11
Anabaena cylindrical Nannochloropsis sp.	43–56	25–30	4–7 20-70

 Table 2-1 Biochemical composition of microalgae
 [26]

The high carbohydrate contents favour the production of biogas and alcohol based fuels. The higher protein and carbohydrates containing microalgae are used in aquaculture and can be applied to extract essential components for the preparation of human and animal feed supplements <sup>28–30</sup>. Most of the microalgae contain many carbohydrates and proteins formed at early stage of photosynthesis to serve the structural and metabolic functions and are the starting point for the synthesis of the other biochemical products. Proteins as enzymes are the prime catalysts for cell

metabolism and facilitate growth<sup>31</sup>. While, the lipids are biosynthesised during the adverse growth stage as a secondary metabolite and its content vary between the species and for their suitability to biodiesel production.<sup>32</sup>

There are three types of lipid found in the algal cells. Triacylglycerol (TAG) is the storage lipid serves as energy reserve are suitable for the biofuel production. Phospholipids and glycolipids are the structural components of membranes of the cell are not suitable for the biofuel. The TAG proportion of lipid fraction increases as the metabolic rate slows down during the late growth stage and this shift in TAG lipid composition occur through the various phases of growth used as the criteria for enhancing the efficiency lipid productivity. The compositions of three lipids in various microalgae species are listed in the Table 2.2. Almost all listed microalgae species accumulated around 50% of storage lipid with the maximum of up to 68% total lipid noted in blue green algae with an average of 35% lipid content accounting to 1/3<sup>rd</sup> of the biomass dry weight. However, the lipid content varies with the changes in the nutritional and culture conditions<sup>33</sup>.

Microalgae	Storage lipids (TAG)	Glyco-lipids	Phospholipids
Diatoms - Chaetoceros species	37 ± 16	$36 \pm 8$	$25\pm8$
Phaeodactylum tricornutum	54± 6	$34\pm5$	$11 \pm 1$
Green algae Chlamydomonas species Dunaliella tertiolecta	$48 \pm 10$ 7 ± 1	44 ± 13 67± 1	$\begin{array}{c} 6\pm 3\\ 25\pm 0\end{array}$
Dunalliella viridis	$13 \pm 1$	44±3	$42 \pm 2$
Blue green algae Nannochloropsis oculata	11-68 $22 \pm 1$	12–41 39± 0	16–50 38 ± 1
Isochrysis species	$36 \pm 3$		$27 \pm 3$
Average	$35 \pm 3$	$40 \pm 2$	$25 \pm 2$

 Table 2-2 Lipid class content as a % of total lipid [<sup>31]</sup>

## 2.3 Industrial application of microalgae

The composition of the algal biomass with regards to lipids, carbohydrates and proteins greatly determine its overall value. Algal lipids are potential substrates used for the biofuel production. High concentration of protein, carbohydrate and long chain fatty acids (Omega-6) present in many species are used for the preparation of functional foods, food additives and nutraceuticals, cosmetics, food, drug and aquaculture application<sup>19,34</sup>. Long chain polyunsaturated fatty acids (LC-PUFAs) derived from the microalgae are the key components of many of the nutraceutical

and pharmaceutical products and are used for the treatments related to maintenance of mental and cardiovascular health<sup>35</sup>. In general microalgae can be applied for three types of biotechnological processes.

1) Biomass generation for biofuel and/or nutraceutical production<sup>36</sup>

2) Wastewater remediation process<sup>37,38</sup>

3) Carbon dioxide sequestering <sup>39</sup>

#### 2.3.1 Application of microalgae for biofuel production

Micro algal biomass have been explored as a renewable source of fuel as early as 1950-1970<sup>36</sup>. But the attention has increased recently due to the increased environmental concerns like GHG emission caused by the fossil fuels<sup>40</sup>. The success of biodiesel production from microalgae as mentioned previously is due to the high oil content (20-50%DW) with high TAG (20-70% total lipid) in the biomass<sup>41</sup>. The specific advantages of using microalgae for biofuel include high oil yield (44) demand less land compared to the first and second generation feed stock. The comparison of some sources of feed stocks of biodiesel are given in Table 2.3.

26

Crop	Oil yield (L/ha)	Land area needed (M ha)	Percent of existing US cropping area
Corn	172	1540	846
Soybean	446	594	326
Canola	1190	223	122
Jatropha	1892	140	77
Coconut Oil	2689	99	54
Palm	5950	45	24
Microalgae (79% oil)	136,900	2	1.1
Microalgae (30% oil)	58,700	4.5	2.5

Tab	le 2	2-3	Comj	pariso	ons of	sources	s of	biod	iesel	. 50
-----	------	-----	------	--------	--------	---------	------	------	-------	------

According to this data, microalgae with 30% lipid content results in 58,700 L/ha of oil yield would occupy only 2.5% of existing US cropping area, while, the production of corn for the biodiesel requires 846% of existing US cropping area. The increase in lipid content drastically reduce the land usage<sup>11,23</sup>. Many species of microalgae have the lipid content higher than 50% provides the

advantages over the oil seeds of vascular plants for biodiesel production. The higher aerial production yield of lipids from microalgae and its ability to grow in wastewater and seawater, demanding less arable land relaxes the stress of fuel versus feed controversy and the negative impacts on the environments.

#### 2.3.2 Mechanisms of lipid accumulation in microalgae

Microalgae with high lipid productivities are desired for producing biodiesel. Lipid productivity is defined as the mass of lipid produced per unit volume of the micro algal culture per day, depends on the algal growth rate and the lipid content of the biomass. Lipid content in microalgae at an average 20-50%(w/w) is commonly found in many species (Table 2.4) can be used for the sustainable and renewable production of energy.

Microalgae	Lipid content (%dry wt.)
Botryococcus braunii	25-75
Chlorella sp.	28–32
Crypthecodinium cohnii	20
Cylindrotheca sp.	16–37
Dunaliella primolecta	23
Isochrysis sp.	25–33
Monallanthus salina	>20
Nannochloris sp.	20–35
Nannochloropsis sp.	31–68
Neochloris oleoabundans	35–54
Nitzschia sp.	45–47
Phaeodactylum tricornutum	20–30
Schizochytrium sp.	50–77
Tetraselmis sueica	15–23

Table 2-4 Oil content of some microalgae <sup>1</sup>	36]
---	-----

As shown in Table 2.4, *Schizochytrium sp., Nannochloropsis sp.* and *Botryococcus braunii* the highest oil content accounting more than 50% of dry weight (32). The lipid content and productivity differ tremendously because of the vast diversity of microalgae and the levels of lipid accumulation affected by growth conditions like nutrients especially nitrogen deprivation and other environmental stresses <sup>28,33,42,43</sup>.

In microalgae, triacylglycerol (TAGs) are the most suitable class of lipid for biodiesel production and are formed under the adverse growth conditions<sup>44,45</sup>. TAGs are non growth associated products, stored in the cells as the carbon and energy reserve and are not formed during cell cycles and hence are not biomass dependent products<sup>46</sup>. Lipid biosynthesis involves two different metabolic pathways for fatty acid synthesis i.e., photosynthesis and the biochemical pathway converting polysaccharides into lipids (Figure 2.1).



Figure 2:1 Simplified diagram of lipid synthesis process in the microalgae cells<sup>47</sup>

The first pathway is the synthesis of lipids by the photosynthesis process occurs in the plastids (chloroplast). The CO<sub>2</sub> produced in photosynthesis comes from outside the plastid into the Calvin cycle, where glycerol-3-phosphate (G3P) is converted to pyruvate and to acetyl-CoA through a reaction catalysed by the pyruvate dehydrogenase complex. Carboxylation of acetyl-CoA is ATP dependent converts it to malonyl-CoA. Conversion of malonyl-CoA is followed by cycles of decarboxylation, addition of malonyl-CoA to acyl units, and β-reduction, reactions that are catalysed by the fatty acid synthase system (FAS) until production of saturated molecules with 16 (C16) and 18(C18) carbon atoms. Palmitic (C16) and oleic acids (18) are the precursors of polyunsaturated molecules produced by aerobic desaturation and elongation mechanisms <sup>31,48</sup>.

The heterotrophic cultivation with the supply of sugar as carbon sources follows the second pathway. This pathway uses the glycolysis and citric acid pathway where the citrate is converted to acetyl-CoA and to malonyl-CoA. Malonyl-CoA is transferred to one of the subunits of FAS, enters the fatty acid synthesis cycle<sup>31,48</sup>.

#### 2.3.3 Conversion of algal lipids into biofuels

Biodiesel from algal lipids is obtained by Trans esterification process of free fatty acid of the TAG in the presence of catalyst (homogeneous or heterogeneous). Figure 2.2 below represents the trans esterification process using NaOH catalyst and methanol for the production of fatty acid methyl ester (biodiesel).



Figure 2:2 Trans esterification reaction of triacylglycerides [49]

The unit operation of processing for the production of biodiesel from microbial oils involves the three steps such as biomass drying, lipid extraction and trans-esterification. Usually the biomass is dried at  $60-70^{\circ}$ C, to maintain the lipid composition as unchanged. In the trans-esterification process three fatty acid molecules are esterified with a molecule of glycerol in presence of methanol/alkalis as trans- esterification or alcolysis. Acids, alkalis and lipase enzymes are used as catalytic agents for trans-esterification<sup>50</sup>. Alkali-catalyzed trans esterification is about 4000 times faster than the acid catalyzed reaction<sup>49</sup> are commonly used in the biodiesel production. The biodiesel obtained from the microalgae should meet the existing standards (ASTM Biodiesel Standard D 6751 in Table 2.5) or European (Standard EN 14214 and if being used for heating oil - Standard EN 14213<sup>36</sup>. The American Society for Testing and Materials (ASTM) defines biodiesel, as a fuel comprised of mono alkyl esters of long chain fatty acids from vegetable oils or animal fats.

Property	Test method	Limits	Units
Flash point closed cup	D 93	130.0 min	°C
Water and sediment	D 2709	0.050 max	%Volume
Kinematic viscosity, 40°C	D 445	1.9–6.0	mm <sup>2</sup> /s
Sulfated ash	D 874	0.020 max	% mass
Sulfur (ppm)	D 5453	0.0015 max (S15)* 0.05 max (S500) *	% mass
Copper strip corrosion	D 130.	No 3 max number	
Cetane no.	D 613	47 min	
Cloud point	D 2500	Report	°C
Carbon residue	D 4530	0.050 max	% mass
Acid number	D 664	0.50 max	mg KOH/g
Free glycerin	D 6584	0.020	% mass
Phosphorus content	D 4951	0.001 max	% mass
Total glycerin	D 6584	0.240	% mass
Sodium/potassium	UOP 391	5 max. Combined	ppm
Distillation temperature, atmospheric equivalent temperature, 90% recovered	D 1160	360 max	°C

 Table 2-5
 ASTM Biodiesel Standard D 6751
 [51]

\*The limits are for Grade S15 and Grade S500 biodiesel, with S15 and S500 refer to maximum sulfur specifications (in ppm)

Biodiesel is the main alternative of diesel fuel in European countries contributes more than three fourths part of overall bio-fuels production <sup>3</sup>. Fatty acids and fatty acid methyl esters (FAME) with four or more double bonds found in some vegetable oil are susceptible to oxidation during storage and this reduces their acceptability for use in biodiesel. The microalgae lipid containing higher amount polyunsaturated fatty acids are not suitable for the biodiesel production<sup>36</sup>.

Even though algal fuel is the best match for the biodiesel production and its commercial cultivation for the effective production of biodiesel depends on the biomass productivity and lipid content. The improvements of oil content and biomass productivity were previously investigated by employing various strategies to reduce the cost of production of algal oil to completely to replace fossil fuels as discussed below.

#### 2.3.3.1 Strategies for improvement of biomass and lipid production

Improvement in lipid productivity is the key factor determines the economic feasibility of micro algal biomass for biodiesel<sup>52,53</sup>. Optimization of lipid productivity can be approached by increasing the growth rate or lipid content of the algal cells<sup>54</sup>. Lipid productivity is mainly affected by the biological potential (genetics) of the species and the environmental conditions<sup>48,55</sup>. Growth rate and lipid production are inversely related and are affected by the culture condition and cultivation systems<sup>44</sup>. The highest lipid yields always resulted in the least biomass production and vice versa<sup>7</sup> and the manipulation of culture conditions for one adversely affects the other. The selection of algae species and the manipulation of physiochemical cultivation parameters are used to determine lipid content. Environmental factors (temperature, light and pH) and availability of nutrients are considered to induce the stress conditions like nitrogen limitation, salinity stress, pH and temperature variation <sup>52,53,56,57</sup> to improve the overall lipid productivity in microalgae. Among these, nutrient-induced shifts for biomass composition are the most cost-efficient and environmental friendly practices <sup>58</sup> applied in biotechnology industries by either nutrient limitation or deprivation to optimize the productivity of the desired target compound(s) used in microalgae

#### 2.3.4 Application of microalgae for wastewater treatment

The other emerging application of micro algal applications is for wastewater treatment. Cultivation of microalgae in wastewater reduces the demand of water and supply of nutrients. The large quantity of nitrogen and phosphorus in the wastewater streams can be effectively utilized without the need for additional carbon sources or nutrients by the algae-based technologies. This technique provides the dual benefits of treatment of waste water and generation of biomass for the biofuel production<sup>60</sup>.

The quantity of nutrients at the specific ration is required for the growth of microalgae. Municipal, industrial and agricultural wastewaters contain different amount of total nitrogen (TN) and Phosphorus (TP) and COD with point source treated with the conventional methods. TN and TP of municipal wastewater range between 19-90 and 2- 20 mg/L with the TN/TP ratio of 3.3 and COD but may contain high quantity of heavy metals. Industrial wastewater from textile, winery, tannery, pulp, and paper and oil mill has varied amount of TN (1.1-532 mg/L), TP (0.6-52 mg /L) and its ratios (2.1-13). Agricultural wastewater from livestock production (Beef, poultry, swine

and dairy) are very rich in nitrogen (63- 4165 mg/L), phosphate (14-1195 mg/L) and this ratio ranges from 2 to 7.8  $^{61}$ .

The wastewater treated by the conventional methods applies primary, secondary and tertiary (advanced) treatment methods. Primary treatment uses mechanical and chemical aids to remove buoyant and non-buoyant suspended materials. Secondary treatment removes dissolved organic and colloidal components by chemical and biological methods. The dissolved inorganic such as N and P are removed by the advanced or the tertiary methods<sup>5,62,63</sup>.

Conventional treatment demands point source and a supply of organic carbon in the wastewater. Treatment by the chemical processes has inherent disadvantages of increasing the dissolved constituents hinders the reuse of water <sup>37</sup>. Biological methods by the microbes are suitable to reduce the organic and inorganic load in the carbon rich only wastewater systems<sup>61,64</sup>.

Not all the wastewater can be treated with conventional methods. Various industrial processes like sea-food processing, abattoirs, meat processing produce the wastewater with high salt contents <sup>63,65,66</sup> and are limited by the biological methods since the presence of salt negatively affects the biological growth<sup>65</sup>. Wastewater effluents with limited carbon sources are also, unable to be treated by biological treatment process. Wastewater generated from some streams may have very low or no organic carbon loading and also may originate from non-point sources, which further limit the conventional treatment strategies. This untreated wastewater causes severe water pollution that disrupts the natural recycling processes such as photosynthesis, respiration, nitrogen fixation, evaporation and precipitation of nutrients in the water systems cause detrimental effect on aquaculture and human beings <sup>67</sup>. The presence of high levels of pathogenic microorganisms, nutrients and toxic compounds accumulates in rivers and lakes<sup>38</sup>. Excessive loading of nutrients into natural water systems causes serious risk such as eutrophication leads to an undesirable disturbance in the aquatic ecosystem and accelerates the harmful micro algal growth<sup>61,68</sup>.

There are two wastewater streams present in Southwestern Ontario that are currently not subjected to the traditional wastewater treatment process, i.e., road salt runoff and vegetable greenhouse industry effluents. These later wastewater streams rich in nitrates, phosphates and other residual nutrients but low in organic carbon loading and the former, the presence of salinity its treatment are limited for the traditional biological process. These wastewater are raising concern on
pollution in Southwestern Ontario<sup>69,70</sup> and are implicated the concern for the algal blooms in Lake Erie of southwestern Ontario in the recent years <sup>71,72</sup>.

#### 2.3.5 Road salt runoff

The deicing agent of most highway maintenance departments use common salt, sodium chloride, sometime supplemented by calcium chloride in the colder regions at alarmingly large quantities in every winter<sup>69,73–75</sup>. The salt is spread on surfaces to melt ice by reducing the freezing point and to act as an abrasive for the road safety. Canada uses on average, 5 million tonnes of road salts per year and approximately 15 million tons of road salts applied to US roads every winter as deicers on roadways <sup>76</sup>. Much of the salt, however, does not remain on the streets and highways, but gets dissolved in the melt water and high runoff flows into surrounding surface waters<sup>77</sup>. Several studies have also suggested that addition of road salt deicers increase environmental problems such as increased salinity in surface and groundwater and increases the transport of heavy metal into roadside soil in the northern hemisphere. Sodium chloride has variable impact on aquatic an environment and aquatic biota<sup>76</sup>. The concentrations ranging up to over 9,000 mg/L of sodium and 13,000 mg/L of chloride in the brine flows on the surface or in the shallow subsurface results in dead vegetation, degradation of fish population in nearby surface streams and significantly affected the contamination of ground water and surface water supply<sup>78</sup> were also noted. Chloride concentrations in wells and springs exceeded the limit of secondary drinking water standard of 250mg/L (U.S.Environmental Protection Agency-EPA-2002). Sodium is an unregulated contaminant recommended not to exceed 60 mg/L because of adverse effects on taste, and not to exceed 20 mg/L for protection of individuals who are on low-sodium diets<sup>78</sup>.

There are various studies conducted reports the impact of road salt runoff on environment. The urban stream of the large transport route in South-central Ontario, Canada is predicted to cause long-term impact of salt loading on surface and groundwater resources. The chloride of approximately 50% of the total road salt applied to Pine Creek (1700tones per winter) removed annually via overland flow with the remainder accumulating in the shallow subsurface results severe degradation of groundwater quality<sup>69</sup>.

Road salt migration is also noted to enhance mineral weathering in shallow aquifers<sup>76</sup>. The road salt runoff of 2004-05 increased the chloride concentrations in the stream to 2000 mg/L, and delivered approximately 850 tons of chloride to a shallow (< 3.5m) semi enclosed lagoon on the

shore of Lake Ontario at Frenchman's Bay of 0.85km<sup>69</sup>. Enormous seasonal influx of NaCl, CaCl<sub>2</sub> and other deicing agents in the wetland soils with increased concentrations of calcium, potassium and magnesium in wetland pore waters were reported by the sediment biogeochemistry. The addition of road salt deicers increase trace metal mobility and its bioavailability in soils and sediments lead to additional risk for the contamination of ground-water through mobilization of heavy metals by road salt deicer with increased amount of dissolved Zn, Cu, Cr, Ni and Pb impacted the soils and sediments <sup>70</sup>. Pond bottom with large concentrations of Na (~2,000 ppm), Cl (~3,000 ppm), Fe (up to 100 ppm), and Mn (up to 4 ppm) from road salt were suggested to influence the redox processes in wetland sediments and soils<sup>79</sup>.

#### 2.3.6 Hydroponic greenhouse effluents

Hydroponic vegetable industry greenhouses are clustered in the winter prevalent regions. A majority of such greenhouse systems located in southwestern Ontario with many hydroponic gardens near Niagara and Learnington area. These industries around the regions produce considerable quantity of wastewater with high concentration of inorganic compounds. The residual nutrients in the hydroponic greenhouse effluent has concentrations of nitrates in the range 200-300 mg NO<sub>3</sub><sup>-</sup>/L and phosphorus (30-100 mg PO<sub>4</sub><sup>-</sup>/L) <sup>80,81</sup>. There are limited treatment methods available for the removal of phosphate and nitrate from these effluents. The predominant modes of treatment for phosphate are crystallization, adsorption and chemical precipitation<sup>82,83</sup>. Precipitation/flocculation process incorporating addition of lime and a biodegradable flocculants (guar gum, cationic starch, or chitosan) for the removal of phosphorus were noted effective <sup>84</sup>.

The common nitrate removal processes include biological denitrification, chemical methods and anaerobic ammonium oxidation process. Anaerobic ammonium oxidation and heterotrophic denitrification with heterotrophic bacteria converts nitrate-N and nitrite N to nitrogen gas with sufficient organic carbon (as an electron donor) for the complete nitrate removal<sup>81</sup>. However, the low carbon availability in these effluents lead to incomplete de nitrification and accumulation of de -nitrification intermediates such as nitrite, nitric oxide and nitrous oxide<sup>80</sup>. Due to the lack of carbon source, there is currently no centralized treatment system for road runoff and hydroponic effluents are available. Microalgae may offer a useful solution to the treatment of such wastewaters <sup>38</sup>. Algae can achieve remediation of wastewater in a more environmentally friendly way by assimilating the nutrients using CO<sub>2</sub> and light (photosynthesis) in to the formation of cellular constituents i.e. lipids and carbohydrates <sup>85</sup>. The biochemical pathways of algal cells for the growth to generate biomass with biogenic products have proven ability to assimilate nitrogen, phosphorus and chemical oxygen demand<sup>5</sup> are referred with the term "Phycoremediation". The generated biomass can be used for biofuel production and the use of wastewater for cultivation of algae would be economical and environmental sustainable way of employing algae for the production of biodiesel oil and high-value end products.

#### 2.3.7 **Phycoremediation**

The term "phycoremediation" refers to the combination of words "phyco" and "remediation" introduced for the treatment process carried out by algae. Phycoremediation defined in a broad sense as the use of macro algae or microalgae for the removal or biotransformation of pollutants, including nutrients and xenobiotic from wastewater<sup>37</sup>. The use of algae for the bioremediation of wastewater has been in vogue for over 40 years to treat the wastewaters with nutrients (nitrogen and phosphorous) and successfully extended to treat effluents from agricultural, agro industrial and industrial wastes <sup>3,68,86,87</sup>.

The use of beneficial microalgae in the bioremediation of wastewater generates biomass suitable for biofuel production with efficient nutrient recycling. This approach of cultivation of microalgae is considered as one of sustainable and feasible option <sup>37</sup> to reduce the cost of the biomass and summarized with the following three advantages,

- Remediation of wastewater nutrients,
- Generation of biomass for biofuel,
- Heavy metal removal.

The use of microalgae were successfully applied for the treatment of various wastewater for the removal of nitrogen and phosphorus from various effluents <sup>40</sup>. Microalgae have the ability to bio remediate the heavy metals and remove the toxic and hazardous compounds like metals, pesticides and recalcitrant from industrial and agricultural wastewater. The functional sites on the cell surface and intracellular ligands like carboxyl, hydroxyl, amino, carbonyl, amide, etc. serves as

bio sorbent assists in binding of metals referred by the term phyco-chelation process to produces reusable effluents <sup>68</sup>.

# 2.3.8 Microalgae based wastewater treatment technologies

The efficient growth of microalgae in wastewater depends on composition, the ratio of N, P and C and conditions such as pH, temperature and the availability of light. Microalgae biomass and lipid production results mainly from photosynthesis using the nutrients ammonium and nitrate are the nitrogen sources from the growing media<sup>88</sup>.

Substrate	Microalgae genera/species	Total nitrogen (mg/L)	Phosphorous (mg/L)	Biomass productivity (mg/L/day)	Lipid productivity (mg/L/day)
Dairy &municipal wastewater	pal wastewater Chlamydomonas reinhardtii		120.6	2000	505
Dairy & municipal wastewater	Green algae	51	2.1	n/a	17/24
Municipal wastewater before primary settling	Chlorella sp.	40.65	5.66	n/a	n/a
Municipal wastewater after primary settling	nicipal wastewater after Chlorella sp. nary settling		6.86 n/a		n/a
Municipal wastewater after Chlorella sp activated sludge tank		19.1	0.32		
Municipal wastewater after sludge Chlorella sp centrifuge: centrate Sea water (50%) & municipal wastewater (50%)		131.5	201.5		
Concentrated municipal wastewater <i>Hindakia sp.</i> 134 (centrate)		110.2	5.3		109
Concentrated municipal wastewater <i>Chlorella</i> (centrate) <i>sorokiniana 134</i>		134	212	275	78.8
Concentrated municipal wastewater <i>Auxenochlorella</i> (centrate)		134	212	183.3	94.8
Concentrated municipal wastewater (centrate) protothecoides Scenedesmus sp. 134 134		134	212	268.8	77.2

 Table 2-6 Microalgae growth and lipid productivity in wastewater
 [87]

The stoichiometry analysis of the chemical formula of algal biomass representing the  $C_{106}H_{263}O_{110}N_{16}$ , indicates 1g of ammonia-nitrogen (NH<sub>3</sub><sup>-</sup>N) or nitrate-nitrogen (NO<sub>3</sub><sup>-</sup>N) produces about 15.8 g of biomass and consumes 18.1 and 24.34 g of CO<sub>2</sub> in the processes, respectively. Various studies have investigated the treatment of the various industrial effluents using micro algae summarized (Table 2.6) and evaluated the biomass and lipid productivity for the integration to biofuel industry. As seen in the Table 2.6, The total nitrogen ranging from 19 to 128 mg/L and phosphorous from 2.1 to 212 mg/L of sources of wastewater treated using microalgae generated the biomass with the productivity maximum 20f g/l/day and lipid of 0.5g/l/day. Thus wastewater has the potential to generate the biomass to the same level as defined growth media <sup>(88)</sup>.

The main challenges in integrating the phycoremediation with biofuel depend on finding the suitable microalgae, which have the ability to survive in different type of wastewater with high nutrient removal efficiency and generation of biomass with high lipid content. Hence the biochemistry and production systems of system of microalgae with the focus on biomass for the biofuel feedstock are discussed in the following sections. The production of the high-density culture is the key factor to reduce the process cost. The cultivation batch, continuous and fed batch have applied for the microalgae. Algal production systems are discussed to improve the biomass and lipid content in the following sessions.

# 2.4 Microalgae production systems

The whole process of microalgae for the biofuel production consists of strain isolation and selection, cultivation, harvesting, drying, lipid extraction and finally the biodiesel production. The optimization of biomass and lipid contents of micro algal cells during the process depends on various factors. Among them genetics is one of the factors that affect the microalga lipid and biomass productivity. Hence, isolation of robust species of microalgae with highest growth and lipid content is vital for efficient biodiesel production. Diverse environments ranging from fresh water, brackish water, saline and hyper saline environment were used as baseline to select the strains. Species with high grow rate significantly reduces the cultivation time and the cultivation cost, while, high lipid contents attributes to high biodiesel yield. Manipulation of culture system using engineering design and changing the physiological metabolism of micro algal cells have been widely investigated to improve lipid production <sup>53,89,90</sup>.

#### 2.4.1 Algal cultivation methods

The nutrient requirements during the growth stage are supplied at the cultivation step. Depending on the energy and carbon source the microalgae can be cultivated by autotrophic, heterotrophic, mixotrophic, or in photo-heterotrophic modes. The impact of stress conditions by the manipulation of the nutritional composition of culture media may be applied to generate specific end products. Among the stress factors nutrimental, physical and chemical stresses were known affect the lipid production.

#### 2.4.2 Nutrimental and physico-chemical stresses

The nutrimental factors are carbon, nitrogen, phosphorus and iron deficiency and physicochemical stress are operational conditions and external factors like light intensities, temperature, pH, salinity and electromagnetic fields that can affect the microalgae growth and influences the lipid production by the addition, deletion or changing some components of cultivation media. Microalgae in autotrophic growth requires about 30 essential elements in various amounts, these essential nutrients are grouped into two categories: (i) macronutrients and (ii) micronutrients (trace elements) as per the quantity of requirements in the media for the optimal growth are presented in the in Table 2.7.

The nutrient requirement can vary with species of microalgae<sup>58</sup>. Nutrient requirement of freshwater microalgae varies from marine and halotolerant/halophilic species. Sea water is common media for the cultivation of marine microalgae species has a relatively constant composition of major ions  $(Na^+,K^+,Mg^{2+},Ca^{2+},Cl^-,SO_4^{2-},HCO_3^-,CO_3^{2-})$  and pH, whereas freshwaters have highly variable compositions <sup>91,64</sup>.

#### 2.4.3 Nutrimental factors: Nitrogen

Nitrogen (N) is an essential element and the key structural component of the most abundant cellular macromolecules such as proteins, nucleic acids and enzymes. The nitrogen content ranges from 1 to 14% of dry weight <sup>64</sup>. Microalgae do not have the ability to convert the atmospheric nitrogen and hence it is supplied in inorganic or organic forms in the media. The inorganic sources like  $NO_3^{-1}$ ,  $NO_2^{-1}$ ,  $NH_4^{+}$  and organic sources like amino acids, urea, uric acid, and dipeptides like xanthine are used by the microalgae. The most widely used nitrogen source for micro algal cultures are nitrate ( $NO_3^{-1}$ ) and Nitrite ( $NO_2^{-1}$ ). Nitrogen uptake by the cells depends on the

availability of nitrate and the other nutrients in media. The transport of nitrogen through the cytosol and the chloroplast finally incorporated into carbon skeletons by the glutamine synthetase/glutamate synthase pathway for the amino acid synthesis <sup>92</sup>.

Nitrogen starvation conditions increase the lipid content but significantly lower the biomass productivity results in lower lipid productivity.

Element	Component added to culture medium	Concentration in	Cell composition
		culture medium (L- <sup>1</sup> )	(mg/g dry weight)
С	$CO_2$ , $HCO_3^-$ , $CO_3$	g	175–650
0	O <sub>2</sub> , H <sub>2</sub> O	g	205–330
Н	H <sub>2</sub> O	g	29–100
Ν	$\mathrm{NH}_4^+$ , $\mathrm{NO}_3^-$ , $\mathrm{NO}_2^-$ , urea etc.	g	10–140
Na	Inorganic salts, i.e. NaCl, Na <sub>2</sub> SO <sub>4</sub> , Na <sub>3</sub> PO <sub>4</sub>	g	0.4–47
K	Inorganic salts, i.e. KCl, K <sub>2</sub> SO <sub>4</sub> , K <sub>3</sub> PO <sub>4</sub>	g	1–75
Ca	Inorganic salts, i.e. $CaCl_2$ , $CaCO_3$	g	0.0-80
Р	Inorganic salts, i.e. Na or K phosphates	g	0.5–33
S	Inorganic salts, i.e. MgSO <sub>4</sub> ·7H2O, or amino acids	g	1.5–16
Mg	Inorganic salts, i.e. Mg sulphates or chlorides	g	0.5–75
Cl	As Na <sup>+</sup> , K <sup>+</sup> , Ca <sup>2+</sup> or NH <sub>4</sub> <sup>+</sup> salts	g	
Fe	In complex with metal ion buffer (e.g. EDTA)	mg	0.2–34
Zn	Inorganic salts, i.e. ZnSO <sub>4</sub> , ZnCl <sub>2</sub>	mg	0.005-1.0
Mn	Inorganic salts, i.e. MnSO <sub>4</sub> , MnCl <sub>2</sub>	mg	0.02–0.24
Br	As Na <sup>+</sup> , $K^+$ , Ca <sup>2+</sup> or NH <sub>4</sub> <sup>+</sup> salts	mg	-
Si	Na <sub>3</sub> SiO <sub>3</sub> ·9H2O	mg	0–230
В	H <sub>3</sub> BO <sub>3</sub>	mg	0.001-0.25
Мо	$Na^+$ or $NH_4^+$ molybdate salts	μg	0.0002-0.001
V	$Na_3VO_4$ ·16H <sub>2</sub> O	μg	-
Sr	As sulphates or chlorides	μg	-
Al	As sulphates or chlorides	μg	-
Rb	As sulphates or chlorides	μg	-
Li	As sulphates or chlorides	μg	-
Cu	As sulphates or chlorides	μg	0.006–0.3
Co	Vitamin B12, sulphates or chlorides	μg	0.0001-0.2
Ι	As Na <sup>+</sup> , K <sup>+</sup> , Ca <sub>2</sub> <sup>+</sup> or NH <sub>4</sub> <sup>+</sup> salts	μg	-
Se	SeO <sub>3</sub> <sup>2-,</sup> SeO <sub>4</sub> <sup>2-</sup>	ng	-

 Table 2-7
 Essentional algal nutrients
 [58]

The limitation in the cells promotes the production of storage compounds like TAGs and starch that do not contain nitrogen<sup>6</sup>. The overproduction of lipid in the cells serves as an alternate carbon and energy reserve. This phenomenon of nitrogen limitation has been widely applied for the improvement of lipid content of biomass for the biofuel production by applying a two stage

cultivation strategy to ensure high growth rate initially and switching the medium components to trigger the lipid accumulation by the nutrient stress <sup>93</sup>

#### 2.4.4 Carbon

All the organic compounds of the microalgae cells consist carbon element ranging from 17.5 to 65% of dry weight and varies among the species and culture conditions with majority of species on an average contain about 50% carbon<sup>64</sup>. Inorganic carbon generally from the CO<sub>2</sub> is converted to organic matter through a complex process of photosynthesis through the Calvin cycle forms the three-carbon compound catalyzed by Ribulose-1,5-bisphosphate carboxylase oxygenase (Rubisco). One gram of dried microalgae biomass requires 1.65-1.83 g of CO<sub>2</sub> are from the carbon fixation process. Microalgae utilize the dissolved CO<sub>2</sub> in water in the form of bicarbonate (HCO<sup>-</sup><sub>3</sub>). Aeration and flue gases are major sources of inorganic CO<sub>2</sub> widely used for the cultivation of microalgae but the CO<sub>2</sub> tolerance varies between the species<sup>60,64</sup>. Microalgal growth via heterotrophic or mixotrophic modes uses organic carbon like monosaccharaides (such as glucose, fructose etc.), starch, volatile fatty acids (such as acetic acid), glycerol and urea<sup>6,33,94</sup>. Sufficient carbon supply and its bioavailability required for increasing growth and biomass generation<sup>95</sup>.

## 2.4.5 **Phosphorus**

Phosphorus (P) is an essential macronutrient necessary for normal growth of all algae and its content varies between 0.5 to  $3.3\%^{64}$ . Phosphorus is required for nucleic acid and phospholipid biosynthesis, modification of protein function and energy transfer (ATP). Algal biomass usually contains less than 1% of P, but under conditions of "luxury uptake," its content can exceed 3 % by DW (Powell et al. 2009) and are obtained from non-renewable sources like phosphate rock<sup>64</sup>. P concentration is growth limiting in natural aqueous environments, because it is easily bound to other ions (CO<sub>3</sub><sup>2-</sup>, Fe<sub>3</sub><sup>+</sup>) resulting in its precipitation, and consequently, the nutrient is unavailable for algal uptake.

The major form in which algae acquire P is inorganic phosphate. The transport of P into the micro algal cell is an energy dependent process and its uptake is affected by the culture condition like pH, temperature and salinity. Under P-replete conditions, microalgae form large polyphosphate granules that serve as internal P storage and are metabolised under P deprivation<sup>43</sup>. This ability exploited in biological P removal from wastewater. Polyphosphate bodies or granules are also

widely distributed amongst many microbial species, where they serve as modulators to overcome stress involved in acclimation to extreme environments, an essential ability for the cell's survival. P limitation lowers the growth rate by reduced concentrations of protein and chlorophylls and decreased level of total adenylates to only 23 % and reduced ADP and ATP levels to 9% and 30%, respectively, as compared to the nutrient sufficient controls were noted <sup>96</sup>.

#### 2.4.6 Sulphur

Sulphur (S) is an essential macronutrient required for many key cellular processes. The content of sulphur ranges from 0.15 to 1.6% <sup>64</sup>. It is required for the synthesis of S-amino acids, membrane sulpho-lipids, cell walls, thiol compounds such as glutathione (GSH) that participate in the stress response, vitamins such as thiamine and biotin, thio-ether and thio-ester compounds such as coenzyme A and S-adenosyl-L-methionine for essential cellular phenomenon is the formation of di-sulphide bonds<sup>9</sup>.

The majority of microalgae use free  $SO_4^{2-}$  as the S source for their nutrition, but inorganic  $SO_4^{2-}$  is often not the most common form of S in the environment. Deprivation and/or low concentrations of N, P or S increase the carotenoid/ chlorophyll ratio. S deprivation was found to trigger lipid (TAGs) accumulation in cytoplasm of *Chlorella sp.* and *C. reinhardtii* <sup>97,98</sup>. Limitation of N, P or S also enables production of algal biomass rich in starch <sup>9</sup>.

Besides macronutrients, a healthy micro algal population also requires micronutrients (e.g. Fe, Mn, Zn, Ni, B, V, Co, Cu, Mo, Se) for growth-promoting effect and cell toxicity, as these elements are required in very small amounts of microgram, nanogram or even picogram per liter <sup>23,58</sup>. Micronutrients usually serve as cofactors in a variety of metabolic pathways essential for micro algal growth. Cellular requirements for micronutrients influenced by the availability of other essential resources like light, N, P, CO<sub>2</sub>, etc., for regulating the cell's metabolism to induce the production of certain metabolites <sup>99</sup>.

# 2.4.7 Physico-chemical factors: salinity

Salinity, abiotic physical environmental stress factors applied by altering the level of common salt in the growth media are used to maintain the axenic conditions and produce neutral lipids. Increases salt concentration causes biochemical changes in the photosynthesis system and induce pigment and lipid production but decreases the growth in both fresh and marine microalgae species <sup>43</sup>. The osmotic changers due the salt stress in combination with nitrogen starvation and low light decreases the photosynthesis ability <sup>100</sup>. Unlimited supply of nutrients increases polar lipid (phospho and glycolipid) production under salt stress but the neutral lipids are accumulated during the unfavorable growth condition.<sup>89</sup>

## 2.4.8 **Physico-chemical factors: light intensities**

Microalgae can grow under autotrophic, heterotrophic, mixotrophic or photoheterotrophic conditions. However, all microalgae are photoautotrophs and can use light as their only energy source for the synthesis of biomass and metabolites. Natural sun light or artificial light in the range of 400- 700 nm intensity provide the energy for photosynthesis but the intensities varies with the micro algal species<sup>53</sup>. Photosynthetically active radiation (PAR) is 45% of the light energy of the earth's surface. Light energy in the form of Photon Flux density (PFD- $\mu$ mol/m<sup>2</sup>/s) is absorbed by the micro algal pigments are used for the Photosynthesis process. Higher PFD causes photo damages and are regulated by the light dark cycles. Light transfer up to 15 cm depth and light intensity between 76-600 micro  $\mu$ mol/m<sup>2</sup> s can be applied for the cells to stimulate the biochemical reaction<sup>101</sup>. The photosynthetic process involves two main steps: light-dependent and dark reactions. During the light reactions, electrons donated by water are transferred to two types of photosynthetic units, PSI and PSII<sup>32</sup>. Along with the cytochrome photons is converted into adenosine triphosphate (ATP) an electron carrier nicotinamide adenine dinucleotide phosphate (NADPH). The products produced by light reactions are then utilized for CO<sup>2</sup> fixation in the Calvin cycle in the subsequent dark reactions<sup>102</sup>.

Pigment group	Color	Ranges of absorption bands (nm)	Pigments
Chlorophylls	Green	450–475	Chlorophyll a
		630-675	Chlorophyll b, c1, c2, d
Carotenoids	Yellow, orange	400–500	β-Carotene,α-Carotene,Lutein
			Violaxanthin, Fucoxanthin,
Phycobilins	Blue, red	500-650	Phycocyanin, Phycoerythirn
			Allophycocyanin

 Table 2-8 Characteristics of the major pigments in microalgae
 [105]

Light harvested by light-capturing pigments on light-harvesting antenna complexes. Different pigments absorb specific wavelengths of the solar spectrum and exhibit a distinctive color as shown in Table 2.8. The light intensities of the wide ranges with the objective on the biomass and lipid production were summarized in the Table 2.9.

i uipose	$(\mu mol/m^2/s)$	Light/dark cycle		
Biomass production	200	12/12		
Biomass and lipid production	105.41-175.68	12/12		
Wastewater treatment	2500	_		
Fatty acid accumulation	62.5	16/8		
	100			
	37.5			
Biogas ungrading	2000	16/8		
Diogas appracting	2000 10/0			
Biogas upgrade and biogas	350	8/16		
effluent nutrient reduction				
Diamaga maduation	100			
Biomass production	100	_		
Biomass production	250	14/10		
-				
Lipid accumulation	82.5	18/6		
Lipid accumulation	87.5 μEm-2 s-1	_		
D'anna 1 atian	100	17/0		
Biomass production	100	10/8		
Biomass production	1200 IUX	24/0 16/9		
Chargen production	100 700  wEm 2  a 1	10/8		
Dispesse and linid and	700 μEIII=2 S=1 420			
carbohydrate production	420	-		
Biomass production	400	_		
Biomass and lipid production	600			
Mixed culture of <i>Chlorella</i> Oil production		_		
on production	1000 104			
	Biomass production Biomass and lipid production Wastewater treatment Fatty acid accumulation Biogas upgrading Biogas upgrade and biogas effluent nutrient reduction Biomass production Lipid accumulation Lipid accumulation Biomass production Biomass production Biomass production Biomass production Biomass production Biomass and lipid and carbohydrate production Biomass and lipid production Oil production	AupoteOptimiz $(\mu mol/m²/s)$ Biomass production200Biomass and lipid production105.41–175.68Wastewater treatment2500Fatty acid accumulation62.5 100 37.5Biogas upgrading2000Biogas upgrade and biogas effluent nutrient reduction350Biomass production100Biomass production100Biomass production250Lipid accumulation82.5Lipid accumulation87.5 $\mu$ Em–2 s–1Biomass production100Biomass production100Biomass production100Biomass production100Biomass production400Biomass and lipid and Biomass and lipid production400Biomass and lipid production600Oil production1000 lux		

 Table 2-9 Optimal light intensity and light/dark cycles for different species
 [105]

The light intensity of < 700-  $\mu$ mol/m<sup>2</sup>/s is used in the control media and while the study of the waste treatment used higher light intensity of 2500  $\mu$ mol/m<sup>2</sup>/s<sup>102</sup> to promote the penetration in the high turbid media. Short wave length light (blue) has high-energy which causes photo inhibition by striking the peak electrical energy to the photo harvest system.

Higher wavelength (red) light with lower energy inhibits the penetration into deep and highdensity cultures<sup>103</sup>. In addition to the type and intensity, the photoperiods with light and dark cycles are essential for the photosynthesis process The light –dark cycle of 12:12, 12:14,14: 10,16:8 have reported to have an effect on cell lipid content <sup>104</sup>. The higher the duration of the light cycle influence the lipid production and fatty acid composition. The application sufficient light energy with nutrients supports the cell growth and hence increases the biomass production. Stepwise increments of light intensity were applied to improve the biomass <sup>105</sup>.

# 2.5 Culturing systems

The growth characteristics and composition of microalgae significantly depend on the cultivation conditions<sup>106</sup>. The major types of cultivation conditions based on the supply energy and carbon sources are photoautotrophic, heterotrophic, mixotrophic and photoheterotrophic used for microalgae. Microalgae use light, (sunlight, as the energy source), and inorganic carbon (e.g., carbon dioxide) as the carbon source to form chemical energy through photosynthesis in phototrophic cultivation and is the commonly used cultivation method<sup>104,107</sup>.

The heterotrophic cultivation widely used to improve the productivity of biomass and lipid. The microalgae grow under either phototrophic or heterotrophic conditions, or both. Microalgae use both organic compounds and inorganic carbon (CO<sub>2</sub>) as a carbon source for growth in the mixotrophic cultivation. The mixotrophic and photoheterotrophic cultivation are same with the latter requires light as the energy source. Some algal species accumulate large amounts of TAG under nutrient limitation decrease in cell division. The variation of these factors are considered as stress and large number of studies notes the effects of environmental stress on TAG biosynthesis and the change of fatty acid profile due to the decrease or cessation of cell division utilizes less storage lipids while synthesis of new lipids continues. In green microalgae, lipid synthesis shares common carbon precursors with starch synthesis. Blocking of the starch synthesis by inactivation of ADP-glucose pyro phosphorylase lead to remarkable increase of TAG biosynthesis is an efficient strategy for enhancing oil content.<sup>108</sup>

#### 2.5.1 Culture systems: batch

High cell density culture is desirable in order to reduce the cost for down- stream processing in commercial production. Culture methods like batch, fed batch and continuous used in the cell cultures are applied for the cultivation of microalgae. In the batch culture, the nutrients and microalgae inoculate added in the beginning of the batch and the cultivation is continued until the

end of the growth period. The growth stage contains lag, log a, stationary and decline stages. During the sufficient availability of nutrient in the beginning of growth time, the cell proliferate at the exponential rate and reaches to stationary phase when the nutrients are exhausted where the cells experience stress and induce the lipid synthesis. Biomass productivity is reduced as the nutrients were depleting during which the lipid production is enhanced. Nutrient limitation evokes change in the physiology and the cells re-program their metabolism to cope with change in nutrient supply or to activate a survival program to outlive sustained periods of starvations.

#### 2.5.2 Culture systems: continues and fed-batch

A batch culture of microbes fed continuously with culture medium is described as a "fed-batch" culture. The reduced biomass productivity compromises the increased lipid productivity enhanced by the feeding of the fresh media or nutrients to increase the cell growth. In fed-batch, cultures may reach a "quasi-steady state" in which the specific growth rate ( $\mu$ /d) virtually equals the dilution rate, that is, the ratio of medium flow rate to culture volume. In a quasi-steady state the specific growth rate gradually decreases. A unique feature of a fed-batch culture is that it allows continuous reproduction of the transient conditions between two specific growth rates, which can be chosen at will. Fed-batch culture may be used to determine the relation between specific growth rate and the growth-limiting substrate concentration and to determine the maintenance energy<sup>109</sup>.

Nutrient-rich medium is used to ensure high growth rate of the micro algal cells, by switching to a different composition to induce the nutrient stress. This can be achieved by the fed-batch techniques<sup>110</sup>. The fed-batch process is an important strategy because it makes possible to control the growth and the lipid accumulation phased by modifying the feed throughout the cultivation process<sup>111</sup>. During the growth period, the metabolite generated contributes the detrimental effect and to reduce these portions of a "fed-batch culture" is withdrawn at intervals leads to a "repeated fed-batch culture". Fed-batch cultivation systems have provides better opportunity to increase the biomass generation which has the potential to modify the light intensity <sup>112</sup>, carbon feeding source <sup>92</sup> for meeting the growth requirements. Various strategies used in fed-batch to improve the biomass and lipids but the contamination due to the additional feeding lines are the major drawback of the application of fed batch systems for the pure cultures<sup>113</sup>.

#### 2.5.3 Nutrient feeding methods

For microalgae, changes in culture conditions and nutrients deficiency affect the cell growth and composition. Light and nitrogen are most reported for their effect on biomass and lipid. Without the nitrogen source, the media is considered nitrate-depleted condition sand this lowers the photosynthesis process. While an insufficient supply of nitrate can support the photosynthesis process but the cells product nitrogen less products like fatty acid and carbohydrates <sup>43</sup>

In fed batch cultivation, staged feeding, repeated feeding or pulse feeding can be provided to apply the controlled supply of nutrients to maintain cells in stages of growth for either increase in biomass or lipid productivity <sup>114,115</sup>. Pulse feeding strategy noted as the easier and cheaper method because of the absence of pumping costs, could lead to lower cell growth, productivity and nitrogen-to-cell conversion factor <sup>115</sup>. The major challenge to the implementation of an integrated algae system in large-scale production of algae is unavailability of cost efficient harvesting process that allows the downstream processing to produce biofuels and other bio-products of value.

# 2.6 Harvesting techniques

Harvesting of biomass is the important step, which involves the recovery of algal cells from the growth media. This step is considered as one of the major bottlenecks in the production of micro algal-based bulk commodities. Harvesting of microalgae accounts for about 20-65% of the micro algal biomass cost<sup>14</sup> which, translates to 50% of the total biofuel production. Harvesting efficiencies can vary depending on the properties of the growth media and morphology of the strains. Microalgae have wide range of properties including size, shape, specific gravity, surface charge, motility, growth phase, presence of appendages and extracellular organic matter that affect their separation from the media. The challenges of harvesting of microalgae are mainly due to the dilute concentration of microalgae in the bioreactor and the characteristic small size of the algal cells<sup>116</sup>.

Microalgae harvesting can be a two-step process. The first is thickening in which micro algal biomass is separated from the bulk culture to the solid concentrates of biomass ranging 2–7% dry weight. The second step involves dewatering, the algal slurry to algal cake with solid concentrate of 15-25% dry weight. Thickening is more energy intensive since the large volume of the culture

is handled <sup>117</sup>. The selection of harvesting methods depend on many factors namely, 1) value and properties of the end product, 2) Nontoxic, non-contaminated biomass or further processing and 3) generation of recyclable culture medium and 4) reasonable cost and very high degree of recovery rate. Following table (Table 2.10) lists the main advantages and disadvantages of the most commonly used harvesting methods.

Harvesting method	Advantages	Disadvantages	Dry solids output concentrat ion (%)
Centrifugation	Fast method, High recovery efficiencies, Can handle most algal types, Recycle of culture media	Expensive method (High capital and operational costs), High energy requirement, Suitable only for the recovery of high value products, possibility of cell damage due to high shear force	10-22
Filtration	High recovery efficiencies, Recycle of culture media Allows the separation of shear sensitive species, Wide variety of filter and membrane types available	Highly dependent on algal species; Best suited to large algal cells, Clogging or fouling Increasing the operation cost (regular cleaning, membrane replacement and pumping)	2–27
Ultrafiltration	Can handle delicate cells , Recycle of culture media	High capital and operational costs	1.5–4
Gravity Sedimentation	Simple and inexpensive method, Low cost, potential for use as a first stage to reduce energy input and cost of subsequent stages Recycle of culture media	Algal species specific, Best suited to dense non-motile cells. Time consuming, Separation can be slow. Possibility of biomass deterioration, Low concentration of algal cake	0.5–3
Chemical flocculation	Simple and fast method, No energy requirement, Wide range of flocculants available, Price varies although can be low cost	Removal of flocculants, Chemical contamination (toxic) of biomass, Recycle of culture media is restricted	3–8
Flotation	Feasibility for large scale application, Low cost method, Low space requirement, Short operational times (can be more rapid than sedimentation) Possibility to combine with gaseous transfer, Recycle of culture media	Algal species specific. Unfeasible for marine microalgae harvesting, Generally requires the use of chemical flocculants, surfactants, High capital and op transfer	7
Electric based process	Applicable to wide variety of microalgae, Do not require the addition of chemical flocculants.	Poorly disseminated, High energy equipment cost	

 Table 2-10 Comparison of micro algal harvesting methods
 [24, 25]

Harvesting of micro algal biomass currently uses mechanical, chemical, biological and electrical based methods. These methods can be combined to improve the separation efficiency to reduce the costs. Conventional harvesting methods like filtration and centrifugation require high amounts of energy and are therefore cost-intensive<sup>118</sup>. Harvesting, such as centrifugation and tangential filtration, can represent 20–57% of the final biomass cost and harvesting cost that they estimated to be \$0.864/L oil.<sup>119</sup>. The selection of the optimal concentration pathway is essential for the reduction of energy requirements <sup>120</sup>.

# 2.6.1 Common harvesting techniques

Various techniques like centrifugation, filtration, flocculation/coagulation, electro flocculation, gravity sedimentation or flotation are compared in Table 2.10 are detailed below.

# 2.6.2 Centrifugation

Centrifugation, a common separation technique used in bioprocessing involves the use of centrifugal forces to separate solids and liquids. The separation is based on the particle size and density difference of the medium components. Centrifugal force accelerates the rate of sedimentation by high-speed rotation of centrifuge that increases the gravity field. The behavior of the smallest particles in the system has the largest effect on separation efficiency <sup>121</sup>.

The use of centrifugation for the recovery of microalgae biomass offers advantages of quick process, high recovery rate and non-contaminated recovered biomass <sup>29</sup>. The physiological properties of most micro algal species and their small size and low concentration in culture result in large energy requirements and operation costs of the centrifugation process <sup>122</sup>. Although centrifugation is a successful dewatering method, the exposure of micro algal cells to high gravitational and shear forces can damage cell structure. Also, the processing of large volumes of culture could be time consuming and costly. The high investment and operating costs required make it not a feasible option for the separation of the micro algal biomass from water <sup>123</sup>

Studies using laboratory centrifuge (500g-1,000g) showed clarification up to 80%-90% algal cells in 2–5 min. Two forms of high-speed centrifugation with acceleration factors of 6,000g and 13,000g resulted >95% cell recovery. The efficiency of recovery decreased with decreasing acceleration: 60% recovery was achieved for centrifugation at 6000g and 40% recovery for centrifugation at 13,000g <sup>124</sup>.

## 2.6.3 Filtration

The filtration of micro-algal cells is based on a solid-liquid separation mechanism involving a semi-permeable barrier with small pores that forces the micro algal cells to be retained and harvested. Harvesting biomass via filtration is one of the favorable methods due to its ability to harvest microalgae with low density <sup>125</sup>

Advantage of using membrane filtration for harvesting the microalgae

- Environmental friendly method since it uses few or no chemicals that are harmful to human and environment.
- Water and residual nutrients in the media can be reuse able
- Inclusion of membrane processes will assist in developing continuous production and process system for the algal biomass
- Low operational cost under low trans membrane pressure compared to centrifugation

There are several filtration methods used for microalgae processing including submerged membranes, cross-flow filtration and dynamic microfiltration. Ultrafiltration refers to filters with pores from 0.02 to 0.2  $\mu$ m and microfiltration refers to filters with pores from 0.1 to 10  $\mu$ m<sup>25</sup>. Microfiltration of pore size of 0.1–0.5  $\mu$ m is preferable for the most microalgae strains <sup>126</sup>. Deadend and the tangential (cross) flows are the two process filtration techniques. Dead-end systems consist of cartridge filtration, horizontal filter press, vacuum drum filter, and belt filter are usually conducted in batch modes around 5–37% solid content can be achieved<sup>127</sup>.

Fouling of the filter causes flux reduction and reduces the performance, which increases the processing costs and is the main disadvantage of membrane filtration. Fouling is caused by the formation of layer of cells (cake) and the cake thickness and effect varies by shape, size, and cell wall rigidity. Microalgae with non-spherical shape, large size and rigid cell wall filtered better than without the cell wall or flexible cells <sup>126</sup>

Fouling can be controlled by optimizing the process conditions including pressure, retention time, and cultivation conditions, improving system design like reactor structure and membrane surface modification, and incorporating pretreatment like mechanical washing, chemical cleaning, and chemical coagulation. Back-flushing, supplementation of coagulant, alteration of membrane surface and shear improvement are studied and reported to reduce the fouling <sup>128</sup>.

#### 2.6.4 Sedimentation

Gravity sedimentation is a simple method to separate liquids and/or solids from liquid with different densities. Density, radius of the particle and sedimentation velocity influences the separation efficiency. The process can be slow when the density of the components being separated is similar. Settling velocities can be theoretically calculated through Stokes' Law using the density, radius of the particles, density of liquid in which they are suspended or discharged<sup>129</sup>. The sedimentation rates of micro algal cells are very low (0.1–2.6 cm h<sup>-1</sup>. Colonial and filamentous algae can be harvested by this method but the singular and motile algae have to flocculate before gravity sedimentation can be applied<sup>130</sup>. The limitation of using gravity settling when the density of media (fresh or salt water around 0.998 and 1.025 g cm<sup>-3</sup> at 20°C, respectively) is close to the micro algal density and its use to separate cells of *Chlorella sp.*, whose density (1.070 g cm<sup>-3</sup>) is not efficient<sup>28</sup>

Sedimentation is however a low cost technology to harvest the microalgae, due to the low capital, operating cost and energy input (0.1 kWh m<sup>-3</sup>). This process does not use chemicals and produces contamination free recyclable media into the cultivation system<sup>129</sup>.

## 2.6.5 Flotation

Flotation is a separation technique used in mineral industries and has been found to be effective for removing algae from suspension. In this process air or gas transformed into bubbles passed into the solid/liquid suspension. The solid particles of the suspension (microalgae) adhere to the gaseous molecules (bubbles) and rise to the surface <sup>121</sup> where they can be collected. This process is faster than sedimentation and requires lower cost.

The flotation efficiency depends on the size and surface characteristics of the microalgae being separated and the size of the air/gas bubbles. The size, charge and hydrophobicity of the micro algal cells will also affect the interaction between the cell and the bubbles. <sup>25</sup>. A higher surface-to-volume ratio (as the size of the bubbles reduces -diameter of 10–30 micrometer) increases the stability of the bubble <sup>122</sup>. Dissolved air, dispersed air, electrolytic flotation, Jameson cell (high mixing) and dispersed ozone flotation (ozone replaces the air) are the techniques used to separate the microalga and are summarized in the following table (Table 2.11).

Table	2-11	Energy	consumption	of	different	flotation	systems	for	microalgae	harvesting
[ <sup>131–133</sup>	]									

Flotation process	Algal species	Flotation	Energy input
		performance (%)	(kWh/m3)
Dissolved air flotation (DAF)	Chlorella	83.7	7.6
	M. aeruginosa	87	Not determined
Dispersed air flotation (DiAF)	Chaetoceros sp.	90	Not determined
	Chlorella and Scenedesmus	76.6	0.003
	D. salina 19/30	>95	Not determined
Electro flotation (FF)	Mixed microalgae	99	20.7
Jameson cell flotation (JCF)	Tetraselmis sp. M8	97.4	Not determined
Dispersed ozone flotation	S. obliquus FSP-3 and C.	95	Not determined
(DOF)	vulgaris		

Flotation in combining the flocculants or surfactants was significantly affected by ionic strength, pH, temperature were reported<sup>131–133</sup>. However these techniques are yet to be explored for industrialization with extensive studies focusing on the sustainability, optimization of the operating parameters using different microalgae species with use of cheap natural surfactants and collectors<sup>134</sup>. Novel methods such as coagulation-copolymerization where distribution of microbubbles and an air release induced microalgae cell flocculate in air bubbles with 94% harvesting efficiency have also been reported<sup>135</sup>.

## 2.6.6 Coagulation-Flocculation

Flocculation based separation been applied successfully in various industries ranging from mining and brewing to water treatment. Flocculation is well known process for the removal of suspended particles from water to produce potable water <sup>136</sup>. Solute particle in a solution forms an aggregate (floc) when the solute particles collide and adhere to each other in the flocculation process. Floc formation is a physico-chemical process influenced by the mixing speed. Chemicals called flocculants are added to induce flocculation by charge neutralization, electrostatic patch mechanism, bridging polymers and sweeping flocculants <sup>137</sup>. <sup>138</sup>. Flocculation improves the rate of sedimentation by aggregating the dispersed particles into larger particles and thus, increasing the recovery of biomass and the dosage of flocculants increases linearly with cell number<sup>117,127</sup>.

## 2.6.7 Flocculation phenomenon in microalgae

Most micro algal cells have a size range between 5 and 50 microns. Micro algal cells can form stable suspensions with chemically reactive cellular surfaces that have a net negative surface charge due to the ionization of functional groups. The stability of these micro algal suspensions is dependent on the forces that interact between the particles themselves and the particles and water. The negative charges of micro algal cell walls keep the cells in dispersion in the solution<sup>119122,139</sup>. Harvesting marine microalgae requires suitable flocculants to induce the flocculation under marine conditions<sup>118</sup>.

Bridging phenomenon describes the linking of the cells by the polymers that neutralize the electronegative charges facilitate the binding of the cells<sup>140,141</sup> induced by applying external chemicals and extra cellular polymeric substances<sup>142</sup>. The zeta potential indicates the stabilization of the particles. Particles with >25 mV zeta potential (positive or negative) maintains stable suspensions due to strong electric repulsion. Particles aggregate when the zeta potential is close to zero <sup>127</sup>. The hydrophobicity and micelle properties of microalgae affect the zeta potential and flocculation efficiency. Longer alkyl chains lead to greater hydrophobicity and a lower critical micelle concentration. Surfactants have hydrophobic ends that can bind insert into the hydrophobic cell membranes<sup>134</sup> enables the separation process.

Charge neutralization occurs due to the interaction of metal and algal cells<sup>122</sup>. Metal hydroxides are formed at high pH, which precipitate the flocs and cause physical linkage between the algae and increases the density of biomass resulting in sedimentation<sup>122,139118140,141 143</sup>.

## 2.6.8 Flocculants for micro algal harvesting

Many different chemicals have been tested for micro algal flocculation can be classified as inorganic and organic polymer/ polyelectrolyte chemical flocculants <sup>9,117,119,127</sup>. Various chemical flocculants applied are listed in Table 2.12 <sup>117</sup> with their advantages and limitations. Inorganic chemical flocculants have binding capabilities, which vary with the dosage, molecular weight and pH. Organic flocculants have high charge densities, are less sensitive to pH and lower dosages are effective to flocculate a wide range of micro algal species. Flocculation followed by sedimentation or filtration consumes less power and can be a cost-effective method for harvesting <sup>144</sup>.

Parameters	Inorganic flocculants	Organic flocculants
Nature of flocculants	Multivalent salts	Polyelectrolytes/polymers
Key characteristics of an	Increasing molecular weight	Flocculants that have a high
effective flocculants	and charge on the polymers	charge density are therefore
	has been shown to increase	more effective
	their binding capabilities	
Sensitivity to pH	Coagulation using inorganic	Coagulation using organic
	coagulants is highly sensitive	coagulants is less sensitive to
	to pH level	рН
Dosage of flocculants required	A large concentration of	Lower dosages of organic
	inorganic flocculants needed	flocculants are required for the
	in order to maintain	flocculation process thus
	flocculation efficiency thereby	producing less quantity of
	producing a large quantity of	sludge and lesser
	sludge and may contaminate	contamination probabilities
	the end product (for example	
	addition of aluminum and iron	
	salts)	
Applicability	Although some coagulants	Wide range of applications i.e.
	may work for some microalgal	they can be used for larger
	species, they do not work for	number of microalgal species
	others	

 Table 2-12
 Comparison of inorganic and organic flocculants
 [119]

Various flocculants were used for the harvesting of microalgae. The application has its own limitation and advantages. The advantageous and the limitation of the specific group flocculation are given in the Table 2.13.

Flocculation	Advantages	Limitations
Metal coagulants	Reliable, well-established technology	Metal enrichment of biomass, toxicity
Biopolymers	Low coagulant dosages, bio-based chemicals	Expensive
Magnetic coagulants	Enhanced separation with magnetic force	Expensive, only established on lab scale
Electrocoagulation	Reliable, low energy requirement.	Metal enrichment of biomass
Ultrasound	No contamination	High power consumption, heating, scalability
Bio flocculation	No addition of chemicals, cheap, sustainable	To be confirmed at scale

 Table 2-13 Overview of highlights and limitations of flocculation
 [144]

## 2.6.9 Flocculation by organic flocculants

Various flocculants are compared in Table 2.14. Polymeric organic substances have the binding sites to link several colloidal particles that in turn is effective for algal flocculation. Among the organic polymers, natural cationic substances like starch from different sources such as wheat, potato, and corn are effectively applied for the flocculation. Chitosan and synthetic polyacrylamide polymers are also being investigated for their effectiveness to harvest the microalgae<sup>145</sup>. These studies have identified the factors such as salinity, pH, biomass concentration, and algal organic matter (AOM) which can affect the efficiency of cationic flocculants<sup>146</sup>.

A desirable flocculants should be non-toxic, recyclable, inexpensive, and efficient at low concentrations. Various chemical flocculants like alum (hydrated aluminum potassium sulfate) or alkali cannot be considered for application in harvesting microalgae for biofuel production due to the toxicity and the cost <sup>117</sup>. The natural polymers are bio degraded to non-toxic compounds and can be used for the production of algae for animal feed or in a bio refinery context <sup>147</sup>. Mung bean (*Vigna radiata*) protein extract (MBPE) used for flocculation of *Nannochloropsis* and the dosage of 20 ml/L, With a mixing rate of 300 rpm for 6 min achieved a efficiency of >92% after 2 h of settling time <sup>148</sup>.

Chitosan, natural cationic polymer, a derivative of chitin obtained from shrimp shells is quite effective at flocculating microalgae but too expensive for large-scale application <sup>147,149</sup>. Chitosan has low solubility and increased viscosity upon hydration is major issue. It is not soluble in pure water or organic solvents but is soluble in aqueous solutions of organic or mineral acids under

specific conditions. The properties of the chitosan change with the pH or the solvents used for the solublisation, impact the flocculation rate and harvesting efficiencies<sup>150,151</sup>

Chemical	Dosage (L <sup>-1</sup> )	Classification	Microalgae	FE settle	Features
flocculants			(Cell density)	time	
$Al_2(SO4)_3$	0.1 g	Inorganic	C. zofingiensis (0.5	>90% (60	pH depended, and risk of
			g L <sup>-1</sup> )	min	secondary pollution
FeCl <sub>3</sub>	0.2 g		Scenedesmus sp.	99% (60	>High dosage, cell
			0.23 g L <sup>-1</sup> )	min)	damages, and risk of
					secondary pollution
$\operatorname{Fe}_2(\operatorname{SO}_4)^3$	1 g	Inorganic	C. minutissima 2.2-		
			108 m L <sup>-1</sup> )		
Aluminium	5.4 mg	Inorganic	N. salina (10-20 g	>95% (30	Residual aluminium in
nitrate sulphate			$L^{-1}$ )	min)	mciroalgal biomass
Ammonia	38-120 mM	Inorganic	C. sorokiniana	>85% (180	Reuse of medium, long
				min)	flocculating time, and
					species dependent Chen
					et al. (2012)
Polyacrylamide	0.05 g	Inorganic	Scenedesmus sp.	60% (10	High pH depended, and
Poly aluminium	20_40 mg	polymer Inorganic	(0.54 g /L) N gaditana	min) 70_80%	risk of toxic acrylamide
chloride	20 40 mg	polymer	$(132 \text{ mg L}^{-1})$	(30 min)	secondary pollution
Chitosan	20–30 mg	Organic	C. sorokiniana (2 g	60–99%	Favour pH depended, and
		polymer	$L^{-1}$ )	(30 min	high cost of chitosan
Cationic starch	20–40 mg	Organic	C. protothecoides $(0.56 + 0.77 \times 1^{-1})$	79–90%	pH depended
Moringa	1 g	Organic	(0.36–0.77 g L ) C.vulgaris (OD730	(80 mm) 88% (120	High dosage, and alkaline
oleifera seed	-	polymer	0.9–1.06)	min)	pH favoured
flour Polyc-glutamic	20 mg	Organic	C. protothecoides	>90% (120	Salinity depended, long
acid (c-PGA	C	polymer	$(0.6 \text{ g L}^{-1})$	min)	flocculating time
			N. oculata (0.6 g $/L^{-}$ ) P. tricornutum		
			(0.6 g /L )		

 Table 2-14 Comparison of different chemical flocculants for microalgae harvesting
 (154)

Synthetic polyacrylamide polymers form the majority of coagulants-flocculants in commercial use are considered nontoxic, they contain monomer residues that are toxic and are effective to flocculate the microalgae<sup>152,153</sup>.

## 2.6.10 Flocculation by pH change

Flocculation induced by pH change has emerged as a viable alternative for large-scale harvesting of algal biomass <sup>154,155</sup>. Increase in pH obtained by adding lime to sewage, to induce the precipitation of calcium carbonate, calcium ortho-phosphate and magnesium hydroxides were commonly used in the wastewater <sup>119</sup> The pH of the culture can increase naturally due to inorganic carbon consumption by the photosynthetic activity of the microalgae<sup>156</sup> and pH can be increased by adding the base like sodium hydroxide which is available at lower cost. After flocculation, the flocculated cells could be re-cultured and the flocculated medium could be reused after neutralizing pH with CO<sub>2</sub> and supplementing nutrients<sup>155</sup>.

# 2.6.11 Mechanism of pH induced flocculation

In natural water, microalgae cells are negatively charged and the repulsive forces impede the cells joining together<sup>157</sup>. An increase in pH causes precipitation of inorganic compounds in the culture medium to cause auto- flocculation or charge neutralisation. When sodium hydroxide is added to raise the pH of culture medium a large amount of positive charges are available and charge neutralization occurs leads to disappearance of repulsion forces. During this stage the algal cells tend to coalesce into large aggregates called flocs, which due to its weight, settle down at the bottom by gravity <sup>158</sup>.

Both media composition and the salinity play major role in the pH-induced flocculation. The efficiency of the flocculation is increased with the amount of Na, Ca and Mg ions <sup>154</sup>. Previously, flocculation induced by pH change for *Scenedesmus obliquus* or *Dunaliella* microalgae harvesting have been reported<sup>159</sup> <sup>160,161</sup>.

## 2.6.12 Bio/Auto flocculation

Bio flocculation methods are successfully applied in micro algal harvesting and wastewater treatment. Changing the physiology of algae or species in co-culture which is often referred to as bio flocculation <sup>162</sup>. Bio flocculation is non-toxic and low-cost harvesting method for microalgae has recently attracted a lot of attention in the field of biofuel research <sup>146</sup>. There are three types of methods which are used: 1) microbial bio flocculants-associated bio flocculation, 2) microorganism-associated bio flocculation, and 3) micro algal cell self-flocculation. The mechanism of microbial bio flocculants-associated bio flocculation is due the ability of

production of the Extracellular Polymeric Substances (EPS) by the microalgae. EPS a glycoprotein consists of polysaccharides (57%) and protein (41%) and presence of negatively charged functional groups such as hydroxyl and carboxyl groups stimulates the formation of floc in the presence of metal ions in the media was reported <sup>146,163</sup>. Bio flocculation depends on multiple factors including the nutrient conditions, pH, and species. Algal–bacterial, aggregation by the interaction between the bacteria and microalgae or by the compounds produced by bacteria are the method that can be applied to harvest the microalgae.

#### 2.6.13 Microalgae/fungal co-palletization

Bio flocculation of algae has been performed by employing suitable microbial partners e.g., algaealgal-fungal interaction<sup>164</sup>. Microalgae algae. algal-bacterial and harvesting using microalgae/fungi co-palletization where microalgae cells are attracted to filamentous fungus and form co-pellets during their co-culture is cost effective and sustainable and was influenced by the pH and calcium ion concentration <sup>165</sup>. Microbial flocculation had been practiced in processes such as wastewater treatment and fermentation and was also suggested for harvesting technique for microalgae <sup>166</sup>. Both microalgae and fungi have low negative zeta-potentials regardless of the pH of the bulk solution (<-10 mV). Fungi have positive electric charge at low pH (pH=3). Different values of pH and the ionic strength of the medium, surface charges of the fungi and microalgae in addition to the magnitude of the zeta potential values influences the co-palletization process.<sup>165</sup>. Rapid flocculation of 99% was developed by ration of the algae and fungi, agitation speed and temperature for Chlorella pyrenoidosa cells with Aspergillus fumigatus within 3 h<sup>164</sup>.

#### 2.6.14 Magnetic flocculation

Magnetic separation is typically used for concentrating magnetic materials and for removing magnet stable particles from air and liquid streams. Magnetic separation processes for ferromagnetic materials or ferrous particles have been used extensively in the past <sup>167</sup>. This technique has also been successfully applied in solid waste separation and recycling, ultrapurification of many fine chemicals, nano and biotechnological products etc. In addition, various types of magnetic particles have been synthesized and investigated to remove magnetic particle tagged algal and microalgae from lakes based on the intrinsic paramagnetic movement<sup>167</sup>.

## 2.6.15 Electrical based harvesting

Micro algal cells are negatively charged and hence the application of electrical field to culture broth can facilitate the separation process by precipitation on the electrodes (electrophoresis) or by accumulation on the bottom of vessel. Sacrificial and Non sacrificial electrodes used and are considered less toxic than the chemical coagulants<sup>25,121</sup>.

# 2.6.16 Flocculation by electromagnetic particles

Electro-coagulation-flotation (ECF) is the technique allows flotation of microalgae cells for surface harvesting<sup>168</sup>. ECF process contains successive stages: (i) formation of coagulants by electrolytic oxidation of the "sacrificial electrode"; (ii) destabilization of the contaminants, particulate suspension, and breaking of emulsions; and (iii) aggregation of the destabilized phases to form flocs <sup>159</sup>. The application of this technique can lead to microalgae to exhibit a negative charge in the presence of proton-active carboxylic, phosphoric, phospho-diester, hydroxyl, and amine functional groups. Then, the anode and cathode in the ECF release metal ions to neutralize the negativity of microalgae and to produce the flocculated activity of microalgae in water or the culture medium <sup>168</sup>.

# 2.7 Conclussions

Microalgal cells accumulate lipids in the form of triacylglycerol during the adverse growth conditions are used for the biodiesel production. The species and its genetics regulates the lipid biosynthesis are also greatly influenced by the environmental and nutritional conditions. The stresses like pH, temperature and salinity are known to induce the lipid biosynthesis. The application of the stress to induce the lipid production reduces the cell growth cycle resulting in the reduced biomass production. Hence, the nutrient management during the growth stage for the improvement of biomass and lipid production is key to enhance the lipid productivity. Fed batch cultivation has the ability to control the feeding rates of the nutrients and the studies on these directions are important. The energy intensive harvesting techniques increases the biomass cost. The simple rapid coagulation- flocculation process has the ability to separate the algal cells from the media, which depends on the type, dosage and composition of the media.

# Chapter 3

# **3** Materials and Methods

In this chapter, the microalgae species, growth media preparation, culture conditions, experimental set up, analytical methods used for the study are described.

# 3.1 Growth media

The synthetic media were used for the cultivation of *N. gaditana* in the experiments. Windsor Safe T-Salt (Windsor Salt Ltd. Pointe-Claire (Québec) a common material used for the deicing the driveway was dissolved in the deionized water was used for the preparation of road salt run-off. The artificial seawater is the control media for the marine microalgae was prepared by dissolving the known quantity of reef salt (Instant Ocean salt, Petsmart, London,ON, Canada) in deionized water.

The solutions (3.5 % w/v) of the above noted reef and road salt were filtered through 0.45  $\mu$ m filters and the 50 ml of filtered samples were submitted to Biotron (Western University, London, Ontario) for the full compositional analysis by inductively coupled plasma atomic absorption spectrometry/mass spectrometry (ICP-MS). The results of the elemental compositions were obtained as mg/L.

# 3.2 Algal culture

The marine microalgae *N. gaditana -L526-* was used for the study. This culture was obtained from national culture collection of marine phytoplankton (National center for marine microalgae and microbiota NCMA), Mane, USA). The cultures were grown and sub-cultured in synthetic seawater media. The fresh cultures (7 days old) containing the biomass of approximately  $500 \pm 50$  mg/L were used at 10 % (v/v) for the inoculation were used for the experiments

# **3.3** Algal growth media and cultivation

The two growth media were used for the cultivation of microalgae in the study. The simulated seawater (reef salt water) and road salt run-off media were prepared by dissolving the above reef and road salts at 3.5 % (w/v) in deionized water. Sodium nitrate as nitrogen source (65mg/L) and

potassium phosphate (5mg/L) was added to the media. The F/2 Gaillard's marine water enrichment solution 50x (Sigma Aldrich) as a source of trace minerals and vitamins were added to the reef salt solutions. The initial nitrate concentration was adjusted 100 m g /L in the micro algal growth media for the experiments. The pH of the growth media in all the experiments were adjusted to  $8.0 \pm 0.05$  using 0.1M HCl and or 0.1 M NaOH, the 40 mM Tris-HCl (Sigma Aldrich) were used as the buffer to restrict the changes in the pH of the media. Synthetic greenhouse effluents were prepared by dissolving the sodium nitrate and sodium phosphate as source of nitrate and phosphate were added as 75 mg NaNO<sub>3</sub>, and 5.65 mg NaH<sub>2</sub>PO<sub>4</sub>. 2H<sub>2</sub>O. All the media were sterilization by 20 minutes of autoclaving at 15 psig.

The marine microalgae *N. gaditana -L526-* cultures were cultivated using the above synthetic reef salt and road salt media. The 2 L conical flask bioreactors with 1.5L of working volumes equipped with sparger were used for batch cultures. The fresh cultures (7 days old) containing the biomass of approximately  $500 \pm 50$  mg/L were used at the rate of 10 % (v/v) for the inoculation All the cultivation studies were conducted under photoautotrophic mode by supplying the carbon dioxide at 1 vvm (volume of air /volume of culture /minute) by aeration. The fluorescent light bulbs were used to provide the light of  $100\pm20$  micromole photon /m<sup>2</sup>/sec intensity on the reactor surface to support the cell growth. The light dark cycles of 12:12 hours were maintained and cultivation was continued for the duration of 21 days and was sub-cultured to maintain the viable cell counts in the culture media.

#### **3.3.1** Microalgae growth media for the fed batch studies

The growth media was prepared by dissolving 26 gram/L of salt (Windsor Safe T-Salt (Windsor Salt Ltd. Pointe-Claire (Québec) in deionized water. Sodium nitrate as a nitrogen source (100 mg/L) and potassium phosphate for the phosphate source (5mg/L) was added to the salt media. The pH of the media was adjusted to 8.0+/-0.1 using 0.1M NaOH solutions. For experiments, exploring the effect of feeding used the stock solution of the nitrate solutions were prepared by dissolving 100 mg NaNO<sub>3</sub> and 7.5 mg NaH<sub>2</sub>PO<sub>4</sub>. 2H<sub>2</sub>O (Sigma Aldrich) in the 1 L deionized. All prepared solutions were autoclaved for 20 minutes at 15-psig prior to use and applied for the cultivation.

The NaNO<sub>3</sub> (Sigma Aldrich) solutions were prepared in the deionized water were added at the rate of 50,75 and 100 mg/L to the road salt growth media for the initial nitrate concentrations in the

batch culture to stimulated the vegetable greenhouse effluents conditions to determine the influence of initial nitrate levels on biomass growth and lipid concentration of *N. gaditana*.

#### **3.3.2** Experimental set up for the batch cultures

Batch cultivation systems used in the study explained below. The simulated seawater (reef salt water) and road salt run-off media were prepared by dissolving various amounts of the above reef and road salts ranging between 2.6, 3.5 % and 4.4 % (w/v) in deionized water. Sodium nitrate as nitrogen source (65mg/L) and potassium phosphate (5mg/L) was added to the media. The F/2 Gaillard's marine water enrichment solution 50x (Sigma Aldrich) as a source of trace minerals and vitamins were added to the reef salt solutions. The initial nitrate concentration were adjusted to 37.5, 75, 100 g /L in the micro algal growth media for the experiments of the investigation of the influence of initial nitrate concentration studies. The pH of the growth media in all the experiments were adjusted to 8.0 +/-0.05 using 0.1M HCl and or 0.1 M NaOH, the 40 mM Tris-HCl (Sigma Aldrich) were used as the buffer to restrict the changes in the pH of the media. Synthetic greenhouse effluents were prepared by dissolving the sodium nitrate and sodium phosphate as source of nitrate and phosphate were added as 75 mg NaNO<sub>3</sub>, and 5.65 mg NaH<sub>2</sub>PO<sub>4</sub>. 2H<sub>2</sub>O. All the media were sterilization by 20 minutes of autoclaving at 15 psig.

The marine microalgae *N. gaditana -L526-* cultures were cultivated using the above synthetic reef salt and road salt media. The 2 L conical flask bioreactors with 1.5L of working volumes were used for batch cultures. The fresh cultures (7 days old) containing the biomass of approximately  $500 \pm 50$  mg/L were used at the rate of 10 % (v/v) for the inoculation All the cultivation studies were conducted under photoautotrophic mode by supplying the carbon dioxide at 1 vvm (volume of air /volume of culture /minute) by aeration. The fluorescent light bulbs were used to provide the light of  $100\pm20$  micromole photon /m<sup>2</sup>/sec intensity on the reactor surface to support the cell growth. The light dark cycles of 12:12 hours were maintained and cultivation was continued for the duration of 21 days.

#### 3.3.3 Experimental set up and cultivation conditions for fed-batch cultures

The stock cultures for the study were cultivated in batch in 4 L Erlenmeyer flasks containing 3L of media and the fed batch culture studies were conducted in 6 L Erlenmeyer flasks containing initially 2 L of media under the photoautotrophic growth conditions. Air was injected at 1% vvm to media to provide the carbon dioxide (CO<sub>2</sub>) and agitation and the flow was controlled with rotameters (Multi-tube rotameters and gas mixer, Omega, Stamford, USA). The photoperiod of 12 hour light and 12-hour dark were used for the study at room temperature (23±2 C) with the average light intensities of 100±10 and 200±10 µmol photons/m<sup>2</sup>/s) were provided by white fluorescent lamps.

Fresh microalgae stock cultures containing average  $550 \pm 50$  mg dry weight biomass per liter were used (10 % (v/v) as the inoculum for the fed-batch system. Fed-batch cultivations were initiated by feeding the nutrient feed media to the batch cultures (initial nitrate concentration average 100 mg/L) on the eighth day when the nitrate concentration was decreased below 10 mg/L. Pulse (intermittent) feeding was carried out as follows. On the eight day, a known volume of nitrate feed (pulse) (from 100 mL to 200 mL at 100 mg/L nitrate) was added to the 6 L flask every two days at a fixed time till day 18. Five feeding rates (10, 12.5, 15, 17.5 and 20 mg/day) were investigated for this mode at two different light regimes. Continuous feeding was done by pumping nitrate feed (500 ml (50 mg/L)) were added to the microalgae culture on Day 5, 10 and 15. This staged feeding mode was repeated for initial nitrate concentrations of 50,75 mg/L. The growth experiments were conducted in duplicates and the analytical methods were applied in triplicates and the averages of the values were used.

#### **3.3.4** Flocculation experiments

The algal culture suspensions from the batch culture were used from the above condition (100mg/L nitrate) was used for the flocculation experiments. The pH induced flocculation methods for the harvesting were conducted by adding 0.1M NaOH (Sigma Aldrich) solutions as flocculating agent to obtain the desired pH in the media.

The micro algal suspensions of 250 ml were placed into each of the 500 mL flasks, and the pH of the media were adjusted to 9.0,9.5, 10,10.5, 11 11.5,12 and 12.5 experiments otherwise the pH was kept at  $8.0 \pm -0.05$ 

The stock solution of  $\text{FeCl}_3(10\text{g/L})$  prepared by dissolving (of  $\text{FeCl}_3$  (Sigma Aldrich) in deionized water used as inorganic flocculants. The stock solutions of three chitosan (1g/L) prepared by dissolving chitosan in 1% acetic, 1% citric and 1% HCl acid solutions by continuous agitation until a clear solution was obtained was used as the organic flocculants for the treatments. The pHs of the chitosan solutions were approximately below pH 2.

The conductivity of the road salt media was and culture suspension were measures in conductivity meter (even compact 230 conductivity meter, Mettler Toledo, Canada) and the values of 32.7 mS/cm for the culture suspension for the media and 38.7 mS/cm were obtained

## **3.4** Analytical Methods

#### 3.4.1 Cell count and biomass concentration quantification

The biomass cell densities were measured by cell count using a heamocytometer. Microalgal samples were drawn using syringe every two days and were placed to the counting chamber of the heamocytometer (Hausser-Scientific, Horsham, PA, USA) and counted using microscope (Leica Microsystems, Weizlar, Germany) with 40x magnification. The growth experiments were conducted in duplicates and the analytical methods were applied in triplicates and the averages of the values were used. Biomass dry weights were calculated gravimetrically at the end of the cultivation period, by vacuum filtering the 50 ml volume of the samples using pre-weighed 0.2  $\mu$ m pore size filters. Filter papers were dried in vacuum oven at 70<sup>o</sup> C for 24 hours. Mass differences were used to quantify the algal dry weight (m) and biomass concentrations were calculated using the following equation.

Biomass X 
$$\left(\frac{g}{L}\right) = m.\frac{1000}{50}$$
 (3.1)

Where m, is the algal cell dry weight (g) in 50 ml of growth media

Specific growth ( $\mu$ ) rate was determined by the slope of the trend line of the plot of the natural logarithm of the cell count versus the cultivation time (days).

$$\mu(/day) = \left(\frac{\ln \frac{Xt}{X0}}{t0 - t}\right)$$
(3.2)

#### 3.4.2 Nitrate measurements

The nitrate concentrations in the growth media were measured using HACK kits. The Nitrate high range Test"N tube (0 to 30.0 mg/L NO<sub>3</sub>-N) HACH kits and spectrophotometer DR 2800 (HACH, Loveland, Colorado, USA) were used. In this a known volumes of growth media from the cultivation reactors were drawn and the cells were filtered using syringe filters (0.45  $\mu$ m). The cell free media were analyzed for the residual NO<sub>3</sub>-N using HACH kits and HACH spectrophotometer-DR 2800 (HACH company, Loveland, Colorado, USA).

## 3.4.3 Chlorophyll quantification

Chlorophyll "a" was estimated following the method described by Porra et.all (<sup>169,170</sup>).In this method 10 ml sample of algal suspensions in the test tubes were centrifuged at 5000 rpm for 10 minutes. The supernatant was discarded and 10 ml of 95% methanol was added to the cell pellet. The test tubes were covered with aluminum foil to reduce the photo oxidation and placed in a water bath at 60<sup>o</sup>C for 60minutes to ensure complete chlorophyll extraction. The tubes were then centrifuged again at 5000 rpm for 10 minutes to separate the cell debris. The absorbance of the supernatant containing extracted chlorophyll "a" in methanol was measured at 652 nm and 665 nm using a Varian Cary 50 Bio- UV visible spectrophotometer (Varian, Palo Alto, California). Chlorophyll "a "concentration was calculated using the following formula.

Total Chlorophyll ( $\mu g/ml$ ) = Chlorophyll 'a' ( $\mu g/ml$ ) (3.3)

Chlorophyll 'a' 
$$(\mu g/ml) = 16.29 \text{ xAb665-8.54 x Ab652}$$
 (3.4)

#### 3.4.4 Lipid quantification

Biomass was harvested at the end of the cultivation by centrifugation (Sorvall ST40, Thermo Scientific<sup>TM</sup> (Life Technologies Inc. Burlington, ON, Canada) for 20 minutes at 3400 rpm and 4°C. Pellets collected were freeze dried using "Free Zone" (Labconco, Kansas City, MO, USA) vacuum freeze dry system. Freeze dried biomass were used for the evaluation of cellular lipid concentration using gravimetrical method<sup>171</sup>. A known amount of dried biomass was

homogenized with10 ml of hexane -isopropanol (3:2 v/v) extraction solvent by vortexing for 30 seconds. These mixtures were placed in test tubes covered with aluminum foil and kept overnight for the complete lipid extraction. A known amount of sodium sulphate solution added to remove the phospholipids from the mixtures. The mixtures were then vacuum filtered and the filtrate of the lipid extracts were mixed with 0.2 times its volume of water and allowed to separate into two phases by centrifugation at 3500 rpm for 3 minutes. The upper phase was containing total lipid were carefully removed and placed in pre-weighed aluminum pans for the evaporation of solvent. Lipid content was calculated gravimetrically by the weight differences and the percentage of lipid determined by using the following equation.

Lipid % = 
$$\left(\frac{\text{Lipid mass (g)}}{\text{Dry mass of micro algae(g)}}\right) \times 100$$
 (3.5)

#### **3.4.5 GC-FAME analysis**

Lipid profiles were estimated by the GC method for the quantification of the triacylglyceride (TAG) by Fatty acid methyl esters (FAME). Lipids extracted from the above methods were subjected for the FAME analysis. This included two steps of process lipid trans-esterification and FAME analysis as described below.

Lipid extracted from microalgae (~100 mg) dissolved in 10 ml of hexane and 100µlof methanolic 2N KOH (11.2g of KOH in 100 ml methanol) in a test tubes. The samples were mixed for 30 seconds using vortex for the trans- esterification process and the samples were allowed to settle the pigments and non-lipid compounds. 1mL of clear samples from the top was carefully transferred in to GC vials and was stored at -20C to prevent evaporation. FAME was analyzed by injecting 2µL samples into an Agilent 7890A GC-flame ionization detector equipped with a 30mX0.32mmX0.25µm J&W HP-5 column. Oven temperature was kept at 80°C for 2 minutes, then heated up to 140°C at the rate of 20°C/min, and then to 260°C at the rate of 4°C/min. Temperature was maintained at 260°C for 10 minutes. Values obtained are used to determine the lipid profile and quantification of TAG<sup>172</sup>.

# **3.5 Determination of flocculation efficiency**

Flocculation experiments were conducted using the Jar test in 50 ml Falcon test tubes and 100ml measuring cylinders. In the first step, 250 ml of the algal cultures from the reactors were

distributed in to the 500 ml of flasks for the pH adjustments and addition of various amounts of flocculants. The flasks were agitated rigoursly at 200 rpm for 2 minutes and were subjected for the slow agitation at 100 rpm for another 5minutes. These suspensions (each 45 ml of the mixtures) were immediately transferred to the 50 ml of Falcon test tubes. The 100 ml of the suspensions were transferred to the 100ml measuring cylinders. The suspensions were left to settle for 15, 30, 45 and 60 minutes without agitation in the falcon test tubes and measuring cylinders. At the end of the settling time, the optical density (OD) of the supernatant from half the height of the clarified layer were measured using spectrophotometer (Varian Cary 50 Bio- UV visible spectrophotometer (Varian, Palo Alto, California) at 682nm. Reference blanks were used for every flocculating agent (growth media + used flocculants concentration) in order to take in to the influence of the dissolved flocculants on absorbance.

The harvesting efficiency (HE) was defined as the ratio of the mass of cells recovered to the total mass of cells was calculated using the following equation

Harvesting efficiency (HE %) = 
$$(A - B/A) \times 100$$
 (3.6)

Where A represents the OD 682 of the control algal culture without the added flocculants and B is the optical density of the supernatant flocculants after the treatments are calculated below

A = OD 682 of the initial culture

B = OD 682 of the supernatant

The samples of the measuring cylinders were used for measuring the concentration factors. The concentration factor was the ratio of the final product concentration to the initial concentration<sup>173</sup>. Concentration factor were calculated as below.

Concentration factor (CF) = 
$$\frac{Vi}{Vi - Vs}$$
 (3.7)

Where,  $V_i$  is the volume of micro algal suspension before flocculation, and  $V_s$  is the volume of microalgal supernatant after flocculation.

# 3.6 Statistical methods

All the experiments were conducted in duplicates and triplicate samples were taken each time for the analysis. The average values and standard deviation were determined and used in the results.

# Chapter 4

# 4 Investigation of phyco-remediation of road salt run-off with marine *N. gaditana*

Summary: Phyco-remediation is an environmental friendly method, which involves the application of beneficial microalgae to treat wastewater-containing pollutants for a diverse range of conditions. Several industrial processes generate hyper saline wastewater, which is a significant challenge for conventional wastewater treatment, and the disposal of saline waters also has a significant negative impact on the environment. Road salt run-off is one such saline wastewater stream not currently treated and one that contributes significant to negatively impacting receiving bodies of water. In the present study, phyco-remediation using the halophilic marine microalgae *N.gaditana* was applied to investigate the removal of nutrients from road salt run-off. The biomass of these algae has high lipid content and they are therefore considered suitable feedstock for the biofuel production. In this study, N.gaditana microalgae were able to assimilate >95 % of the nitrates within 8 days in road salt concentrations ranging from 2.6 to 4.4% under photoautotrophic cultivation method. Biomass yield of 1.1 g/L of culture obtained with the maximum lipid of 22% (g/g) biomass in the road salt media. Fatty acid composition analysis of the obtained lipid indicated C16 and C18 as the major components (~ 45% of FAME) further confirmed the suitability of biomass generated could be used for biofuel feedstock. This study established that the use of road salt run off containing nitrate and phosphate nutrients will support the growth marine micro algae for remediation of waste water system that are the concern at winter prevalent regions.

# 4.1 Introduction

The application of road (de-icing) salts (e.g. sodium chloride) is carried out on roads and highways in certain areas of Canada and northern United States during the winter months. This allows for ice and snow to melt due to lowered freezing points of the mixture. The subsequent brine solution drains as road salt runoff into underlying soil or surface waters or storm water management ponds (SWMPs) and sewer systems. The partial replacement of de-icing salts with biodegradable materials adds COD, nitrates and phosphates to the salt run-off hence may increase eutrophication of receiving bodies of water. A large amount of run-off can end up in catchment
basins and SWMPs. Nitrogen- and phosphorus pollution also arises from urban runoff from roads and highways, residential areas and grassland, vehicular exhaust, or overuse of fertilizers. Phosphorus is often found to be the limiting nutrient for freshwater aquatic ecosystems, but can increases in storm water detention ponds with time. A significant issue with phosphorus accumulation is production of toxins such as microcystins during harmful algal (cyanobacterial) blooms (HABs) which can occur in the SWMPs due to the high phosphorus levels<sup>174</sup>. The HAB also lead to eutrophication due to oxygen depletion.

Microalgae are a diverse group of single celled photosynthetic organisms grows in wide range of aquatic environments from fresh to saline water using carbon dioxide and sunlight. Phycoremediation applies beneficial microalgae for the removal of nutrients from the waste water <sup>5</sup>. The produced micro algal biomass is rich in lipid content and has considerable potential for biofuel production or nutraceuticals<sup>15,29,36,37,175</sup>. Microalgae during photosynthesis utilize solar energy and several essential nutrients (C, N, P, S, K, Fe etc.) to synthesize biomass compounds and to multiply their cells. Typical cell composition of microalgae undergoing photosynthetic growth requires the addition of nutrients in the so-called Redfield ratio of N: P: 16:1 <sup>28,176</sup>. Light and nutrient supply controls the growth kinetics of microalgae<sup>177</sup>. Variation in the growth conditions and nutrient shifts induce stress that diverts the metabolic pathways directing the synthesis of high value secondary metabolites like lipid and pigments <sup>58,95,178</sup>.

*N.gaditana* a species of marine micro algae has been reported to have an optimal growth in salinity ranging from 0.5 M to 1.0 M NaCl. Biomass of this algae is rich in lipid and pigments like astaxanthin which are of great commercial interest <sup>18</sup>. *Nannochloropsis* sp. also exhibit a diverse fatty acid composition due to their physiological and eco-physiological variabilities <sup>179</sup>. Marine microalgae are extremophiles are robust organisms producing stable enzymes, and are often able to tolerate changes in environmental conditions<sup>180</sup>. If biodiesel is to become an economically viable resource, more efficient novel sources of oil, such as microalgae as well as from extremophiles organisms, need to be researched. The advantage of using extremophile microalgae would be to minimize contamination within the photo bioreactors, which tends to be problematic in outdoor cultures <sup>181</sup>. Although its cultivation using salt media like road salt run off not previously reported.

Microalgae biomass have low cellulose content compared to the terrestrial crops and thus the nutrient demand is much higher<sup>64</sup>. Mass production of microalgae for biofuel requires massive quantity of nutrients. In large-scale cultivation systems, large amount nutrients like nitrogen, phosphorous and carbon are required to supplement the cultivation medium to the needs of micro algal growth under the specific environmental conditions, in order to achieve high yields. Many studies evaluated the potential of various micro algal monocultures for nutrient removal and biomass production in various types of wastewaters. Availability of the major and minor nutrients affects the cellular biochemical reactions and production of biomass and metabolites were noted. Since, wastewaters differ in their physicochemical characteristics, which directly or indirectly have an effect on algal growth<sup>64</sup>. The present study examined the growth kinetics of *N.gaditana* on synthetically prepared (simulated) road salt run off media and reef salt media (artificial seawater) to investigate its suitability for phyco-remediation. The generated biomass was further evaluated for lipid content to determine its suitability as feedstock for the production of biofuel production by determining the fatty acid profiles.

### 4.2 **Results and Discussion**

Road salt run-off is highly variable wastewater and the composition of nutrients (nitrate and phosphate) present in the run-off depends on location, time of year and weather. Various nutrients are required to carry various biochemical reactions to support the growth of microalgae. Nitrogen is the critical nutrient required for the production of nucleic acid, protein, chlorophyll and enzymes. Phosphorous required for the synthesis of energy transfer system of the cells such as ATP, NADPH and phospholipids. Other nutrients like Ca, Mg activates the acetyl CoA and NADPH and fatty acid synthesis <sup>182</sup>. Iron (ferric) one of the most important elements required by most microalgae. Ferric ion involves in fundamental enzymatic reactions, photosystem II (PS II) ,nitrogen consumption and chlorophyll synthesis in the algal cells <sup>183</sup>.

The compositional characteristics of 3.5 % solution of road salt and reef salt solution were first analyzed for the presence of major and trace metals and are shown in Table. 4.1. All elements (major and minor nutrients) found in seawater were identified in the road salt solution with varied concentration. The concentration of sodium ions noted higher in road salt media (5339 mg/L) than in the reef salt media (2229 mg/L). Sodium and chloride are the key salinity providing elements which support the growth of the halophilic microalgae was present in ~2.4x and calcium about

1.4x concentrations in road salt solution. Apart from these, sulphur, magnesium, ferrous and cobalt are also present in a higher concentration. Elemental compositional analysis identified the presence of all most all the ions that would support the growth of marine microalgae in the simulated road salt indicated the feasibility to use road salt run off as base media for the cultivation. Various concentrations of salt and nitrates were used to mimic the field conditions to determine N removal efficient for its application to the phyco-remediation.

Elements in the media (mg/L)	2.6% Road salt (w/v)	3.5% Road salt (w/v)	4.4% Road salt (w/v)	2.6% Reef salt (w/v)	3.5% Reef salt (w/v)	4.4% Reef salt (w/v)
Na	3964	5336	6708	1655	2228	2801
Mg	5	6	8	4722	6357	7991
Al	0.3	0.3	0.4	0.1	0.2	0.2
Si	< 0.01	< 0.01	< 0.010	< 0.01	< 0.01	< 0.01
Р	0.0	0.0	0.0	0.0	0.0	0.0
S	32	42	53	167	225	283
Κ	3	4	5	169	227	285
Ca	102	138	173	147	198	249
Mn	0.0	0.1	0.1	0.0	0.0	0.0
Fe	< 0.010	< 0.010	< 0.010	< 0.010	< 0.010	< 0.010
Со	0.0	0.0	0.0	0.0	0.0	0.0
Cu	0.1	0.1	0.1	0.0	0.0	0.0
Zn	0.1	0.1	0.1	0.0	0.0	0.0
As	0.4	0.5	0.6	0.0	0.0	0.0
Sr	2	3	3	6	8	10
Ba	0.0	0.1	0.1	0.0	0.0	0.0

Table 4-1 Elemental concentration of reef and road salt media analyzed by the ICP- MS

# 4.2.1 Effect of media and salt concentration on growth kinetics of *N.gaditana*

The growth kinetics of marine algae was studied by analyzing the cell density and pigment concentration. The cell multiplication time profile was represented by the growth curves. The various stages in growth curves identify the status of the cells in the media, which are specific to species, and are influenced by the environmental and nutrient conditions of the media. The typical growth curve includes the lag, exponential, stationary and decline stages for microalgae. Experiments were conducted to understand the growth kinetics of algal cell production for 21 days

of cultivation time at three salt levels of media. The results are shown for road salt (Figure 4.1) and reef salt (Figure 4.2). It was observed that *N.gaditana* cells rapidly adapted to the simulated growth media as noted by the short lag period at all levels of salt concentrations. The active growth phases (exponential growth) from day two showed a linear increase in cell density. This growth phase continued without reaching the stationary phase (in 21 days) in 2.6% reef salt. In 3.5% reef salt media showed a decline in growth phase on 16<sup>th</sup> day and maintained the stationary phase (in 21<sup>th</sup> day).

Although, all these culture media started from the similar concentration of substrate nitrate (65mg/L) and initial cell concentration (~ 35-45  $10^5$  cells /ml), two noticeable apparent growth phases, an active growth phase and the stationary phase, were observed. The highest cell density (2.93 x  $10^{7/}$ ml) and (3.18 x  $10^{7/}$ ml) in 3.5% reef and in 4.4 % road salt respectively were observed during the active growth on the 14th day of the cultivation.

This variation in the cell density can be attributed that the value of salt concentration particularly the sodium ion in the media affected the algal growth due to the variation in osmotic concentration and its effect on permeability of other elemental compounds to cytoplast for the cellular biochemical reaction that shifted the cell growth cycles.

Cells started to grow at relatively higher concentration in road salt media than in the reef salt at elevated media concentrations. In general road salt media showed increased cell counts than the reef salt media even though the initial N and P concentration, light and carbon were maintained at the same level in all the experiments. Many species of microalgae exhibit positive growth under nutrient sufficient conditions with optimum environmental conditions like light and carbon dioxide supply were provided. N and P are the key nutrients promote growth significantly <sup>184</sup>, Iron, copper were also noted as influencing elements that at specific concentration increases the cell density.



Figure 4:1 Growth curves by the cell count of the N.gaditana in road salt media



Figure 4:2 Growth curves by the cell count of the N.gaditana in reef salt media

Copper was reported to increase the chloroplast and lipid content of the cells <sup>126</sup> by promoting esterase activity. Since the amount of trace metals were noted higher concentration in road salt media than in the reef salt media that directly influenced the cellular activity and generations.

Further observation on the varied salt concentration demonstrated that higher salinity resulted in lower cell concentration. Optimum salinity for the growth of the marine microalgae is in the range of seawater but the halophilic microalgae have the varied salt tolerance with the tendency to grow in the wide range of salt concentration. The cell count data of this study showed steady increment of cells reaching the maximum of (3.18 x  $10^7$ /ml) in 4.4% road salt base media in  $14^{th}$  day as the optimum level. But these reactors noted the salt precipitation indicating the concentration of this level is too high to apply to the field condition where there would be a possibility of evaporation, which might lead to osmotic shock and hinder the cell growth if the cultivation period prolonged to 21 days.

The pigments play vital role in photosynthesis to capture the light energy for the conversation to chemical energy in autotrophs. *Nannochloropsis* species of genera *Eustigmatophyceae* consists only chlorophyll "a" (Chl-"a") pigment with several kinds of carotenoids<sup>179</sup>. The level of Chl-"a" in the algal cells indicate the physiological state of the culture <sup>103</sup>. In order to explain the difference in the growth profile between the reef and the road salt media the Chl- "a" concentration measured during the various growth stages and the results shown in Figure.4.3 and 4.4



Figure 4:3 Cellular concentration of chlorophyll "a" in biomass of road salt media



Figure 4:4 Cellular concentration of chlorophyll "a" in biomass of reef salt media

Varied salinity levels affected the Chl-"a" content of cell mass. In general, steady increment of Chl-"a" concentration observed as the cultivation time progressed. The decline in Chl-"a"-content observed when cells reached to the stationary level in all the experimental conditions. The highest concentration of Chl-"a" (7.0 $\mu$ g /ml) noted in 3.5%

reef salt media and the road salt media provided only 2.1  $\mu$ g /ml (~1/3rd) in 4.4% media. The higher reef salt media concentration (4.4%) resulted in a maximum pigment concentration in day 18. Elevated road salt (2.6% and 4.4%) resulted in lower pigments of 1.67 and 1.49 $\mu$ g /ml in day 18 and 20 respectively. However, the lowest concentrations of Chl-"a" were detected in both the lower (2.6%) salt levels.

Chlorophyll production is the key indicator of the ability of photosynthesis and cellular growth. Chlorophyll is the green pigment responsible for the transfer of light energy and its conversion in autotrophic green algae and land plants <sup>179</sup>. Chlorophyll structure consists of N and composed of four pyrrole group forming around the magnesium atom<sup>185</sup> indicating the requirement of N and magnesium for the biosynthesis of chlorophyll pigment. Many studies also noted the environmental factors including pH, light and temperature are known to influence cellular physiological characteristics such as the nutrient removal and the yield of pigments <sup>185</sup>.

Meanwhile, Chl-"a" concentration followed the cell growth pattern and was steadily increased when the cells reached to exponential growth phase indicating the photosynthesis correlation of pigment and the cell growth. It is known from the previous studies that the chlorophyll and some carotenoids can absorb light energy and collect in the core reaction of photosynthetic process. The cells attempts to increase the bioenergetics yield by increased biosynthesis of chlorophyll to absorb more light <sup>186</sup>. The total chlorophyll production also always depends on the cellular concentration and nitrogen sources in addition to light energy.

The pigment Chl-"a" concentration in the cells from different cultivation time as function of media concentration, increased as the cultivation period progressed in both media even after the nitrate depletion until the stationary growth stage indicating that the Chl-"a" production is growth dependent. Secondary carotenoid accumulation known to be promoted by nitrogen starvation in aged cultures of many microalgae species, indicating the cessation of chlorophyll for major carotenoid synthesis <sup>179</sup>. The study on *Chlamydomonas* and Coccomyxa suggested during the nutrient depletion the turnover of nitrogen-rich compounds such as proteins might provide carbon and energy for TAG

biosynthesis for the nutrient deprived cells<sup>12</sup> where the left over nitrogen from the protein structures further utilized for the generation pigments illustrating the continued chlorophyll production in the nitrate depleted media<sup>103</sup>. Increased cell density during the late stages also limits the light availability for the photosynthesis. In conclusion, variation in the Chl-"a" concentrations of *Nannochloropsis* were observed in the two tested media with the highest in reef salt media than the road salt media due to the compositional variations observed in these media. Cell count and Chl-"a" concentration directly affected by the level and type of base media used were noted in this study.

## 4.2.2 Effect of type of media and its salt concentration on nitrate consumption of *N. gaditana*

In order to understand the application of phyco-remediation technique to the road salt run off, the growth kinetics of the cells in related to the substrate nitrate, time course profiles of nitrogen concentration in the media during the cultivation period were analyzed. Kinetic profiles of nitrate consumption in the studies determined by analyzing the residual nitrogen-nitrate (N) levels of the cell free media at the fixed time periods during the growth cycle were plotted in the Figures 4.5 and 4.6. Initial N level in all the media were maintained around 65mg /L. Complete consumption of the nitrogen nitrate in culture media recorded when the readings were below the detectable limits of the test method. Concentration of the N level decreased as the cell growth progressed in all the experimental conditions. Faster reduction of nitrate observed in reef salt than in road salt media. Nitrate levels decreased to nil (100% removal) and 5 mg/L (92 % removal) in 3.5% and 4.4% at day 8 respectively, while 2mg/L (97% removal) in 2.6% in 6 days in reef salt media. Nitrate concentration depleted to 3.4 mg/L (95% removal) and 2.7 mg/L (96 % removal) in 2.6% and 3.5% and to 0 mg/L (100% removal) in 4.4% road salt media at 8 and 6 days respectively indicating that *N.gaditana* microalgae were effective for removal of nitrate from the reef and road salt media.



Figure 4:5 Concentration of Nitrate -Nitrogen versus cultivation in road salt media



Figure 4:6 Concentration of Nitrate -Nitrogen versus cultivation in reef salt media

Analysis of nitrate in cell free media during the cultivation period confirmed progressive N depletion with increase cell count indicating the consumption of the N by the cell for

the cellular biochemical reaction to generate the biomass. In general, time taken to deplete the N in reef salt media varied from the road salt media due to the concentration of the elemental composition also varied. The highest salt concentration (4.4%) reported longer time duration (8days) for the depletion of nitrate and the fastest depletion to the lowest value (~97%) was observed in day 6 noticed in reef salt media at lowest salt concentration (2.6%). Increased salt concentration decreased the nitrate assimilation rate in both the salt media indicating that the cellular nitrate assimilation further affected by the media type, concentration and composition but supported the cell growth indicating wide range of salt tolerance of the marine halophilic microalgae. Marine microalgae have the optimum growth at the salt level of 3.5 % that is equal to the salinity of seawater. Elevation from this condition induces a stress environment to the growth hence modifies the metabolism rate <sup>187</sup> thus the demands of the nitrate assimilation in to the cytosol. Nitrogen including phosphorus and carbon are the key elements that directly support the cell growth rate. Cells increase metabolism rate during the initial growth stage when the nutrients in the media are in abundant supply. Photosynthetic metabolism of microalgae is also supported by the availability of light and chlorophyll pigment<sup>187</sup>. Nitrate consumed at day 8 in all the media during which the cell concentration increased progressively to the exponential growth stage. The work on fresh water micro algae -Chlorella vulagaris provided similar trends of nitrate consumption profile at varied salt concentration with the maximum assimilation of nitrates found at 2 to 4 days of cultivation where the cells are in exponential growth phase<sup>188</sup>. This result indicates that the increasing salt concentration reduces and nitrate assimilation in reef salt while the optimum reef salt concentration of 3.5 % and 2.6 % road salt provided the fastest rate of nitrate consumption and indicating the optimal media concentration to consider for the phyco-remediation using the marine microalgae. The mechanism behind the effect of N depletion believed to be associated with decrease in intracellular chlorophyll, which play very important role in the cell division rate in addition to the other N containing cellular components. Algae use organic and inorganic N for the synthesis of amino acids and nucleic acid<sup>189</sup> and its demand is highest at the during the active growth stage. Hence the highest N assimilation is noted when the cells were in the exponential growth stage. However, the cell growth continues after the exhaustion of media nitrogen indicating the

nitrogen supply from intracellular storage such as RuBisCo supports the survival of cells. Nutrient starvation triggers the secondary metabolism leading to biosynthesis of lipid was also reported (30).

However the cellular growth rates were affected by the environmental and physical conditions of media. Variation in the salt concentration considered as the stress, which alters the cellular metabolic activities and multiplication rate. Thus the cellular nutrient demands in the stress induced media were different than the optimal growth conditions. In the study varied N depletion rates were noticed. Increased salt concentration showed the reduced N depletion rate and vice versa at the lower salt concentration of the media.

# 4.2.3 Specific growth rates of *N.gaditana* at various media salt concentration

Specific growth rate determined were presented for the both media at three level of media concentration (Table 4.2). Road salt at 4.4% and 2.6% provided the highest growth rate with ( $\mu = 0.26$ /day) while the least growth rate ( $\mu = 0.13$  /day) was noted in the lowest reef salt media (2.6%).

Table 4.2: Specific growth rates ( $\mu$ ) and the doubling time of *N.gaditana* in different cultivation media at 2.6, 3.5 and 4.4% (w/v) salt concentrations.

Growth media	Media salt concentration (%w/v)	Specific growth rate µ (/day)	Doubling time (Td) (days)
Road Salt			
	4.4	0.26	2.7
	3.5	0.15	4.6
	2.4	0.26	2.7
Reef Salt			
	4.4	0.21	3.3
	3.5	0.24	2.9
	2.4	0.13	5.3

Higher specific growth rate resulted in higher biomass production. The change in the external salt concentration disturbing the ion homeostasis in cytoplasm and the varied cell capacity for the regulation of the Na<sup>+</sup> homeostasis during cell adaptation changing the growth rate were reported <sup>190</sup> and was attributed to the results of in varied specific

growth rates. As noted earlier various environment factors (light and carbon source in autotrophs) and concentration of major and minor nutrients in the growth media and its bioavailability to the cellular reactions were responsible for regulating the rate of cell reproduction and thus shifting the specific growth rate <sup>191</sup>

### 4.2.4 Effect of media, salt and initial media nitrate concentration on biomass production

Determination of biomass production is important in cultivation study of microalgae for its feasibility of its application to biofuel feed stock. Biomass quantities generated were determined at the end of cultivation (Figure 4.7). Changing the growth conditions directly affect the cellular production rate and thus biomass yield. The highest biomass of 1.8 g/L was obtained in the reef salt media at 3.5% salt concentration and lowest of 0.77 g/L biomass was noted in 4.4 % reef salt media. Biomass yield of 0.97, 0.99 and 0.88 g/L were obtained in road salt media at 4.4, 3.5 and 2.6 % salt concentration respectively. The most favourable salt concentration noted was 3.5% reef salt yielded the highest biomass. Increase the salt concentration by 25% to 4.4% reduced the biomass yield by 57% (0.77 g/L) and decrease the media concentration to 2.6% reduced the biomass by 55% and produced 1g/L of biomass. Biomass yield in road salt media obtained was ~ 1 g/L. Increase in the concentration of media increased the osmotic pressure. Osmotic up shocks generally cause severe temporary inhibition of photosynthesis in microalgae, however, osmotic down shocks have little effect on photosynthesis in microalgae with strong cell walls, while wall-less species indicated higher sensitivity. Rapid glycerol synthesis takes place in response to increased external salinity in *Chlamydomonas pulsatilla* were reported (29, 35, 34) indicating the cell protection mechanism than the cell multiplication on the salt stress conditions.



Figure 4:7 Biomass dry weight of N. gaditana in road and reef salt growth media

Biomass yield depends on the supply of the major and trace nutrients. In general for biomass growth (consisting of 40–50% carbon) microalgae depend on a sufficient supply of a carbon source and light to carry out photosynthesis<sup>29</sup> in addition to the abundant supply of Nitrogen and Phosphorous. *Nannochloropsis sp.* also could adapt to a relatively wide range of salinity and other growth conditions <sup>192</sup> was the reason to obtain the similar level of biomass in all the conditions studied.

# 4.2.5 Effect of media, salt and nitrate concentration on lipid production and fatty acid profile of *N. gaditana*

The biomass accumulations in microalgae also depended on the availability of sufficient number of nutrients in specific ratio for the cellular activity with the growth supporting environmental conditions. As previously noted the nitrate plays major role in the cellular biochemical reaction. The influence of initial nitrate concentration on biomass production further studied using 2.6% road salt media at three external nitrate concentrations (37.5, 75, 100mg/L) to understand the influence of the substrates N in the on the biomass and lipid production in *N.gaditana* (Table 4.2) with an attempt to mimic the variability of the field condition for the feasibility of the phyco-remediated biomass for the biofuel feedstock production. In general, biomass production rate varied with the

change in the initial media nitrate concentration. Increasing the initial nitrate level in the media from 37.5 to 100mg/L increased the biomass to ~2.8x indicated the biomass generation is substrate dependent. Increase in the initial nitrate concentrate to100mg/L changed the biomass production to 1.78 from 0.7 g/L. The highest biomass of 2.0 g/L obtained at 75 mg/L nitrate containing media. The lowest biomass production noticed at the lowest nitrate concentration (37.5mg/L) indicating this level of nitrate is below the critical limit for the optimum growth. Since the limitation of the other elements like P and C in increased N (100 mg/L) might have resulted in slight reduction of the biomass from the maximum 2.0 to 1.78g/L. This study demonstrated the correlation resulting with variation in the biomass yields in road salt media and to the substrate N level as noted in the previous studied <sup>180</sup>. Effect of media, salt and nitrate concentration on lipid production and fatty acid profile of *N. gaditana* 

#### 4.2.6 Lipid productivity and salinity

Many microalgae exhibit the ability to enhance lipid content during the stress conditions. Environment and nutrient starvation are the common stress condition. Varying the salt concentration widely applied as cheap stress inducer to trigger the secondary metabolism and to enhance the lipid content of cells <sup>193</sup>. Lipid and biomass to sodium conversion in the tested salt concentration of both growth media were determined to understand the influence of the fluctuations of the salt as stress inducer on the lipid production (Table 4.3). Lipid production of 53.4% and 56.2% (w/w) was observed in the biomass obtained from the 2.6 and 4.4 % and the average of 18.6% (w/w) in 3.5% reef salt media. As the road salt level increased the lipid content in the cells also steadily increased from 17.0 to 21.6% (w/w). Lipid production changed from 23 % (w/w) to 28% (w/w) when the nitrate level elevated from 75 mg/ L to 100 or 37.5 mg/L indicating substrate stress changed lipid production within the tested ranges on nitrate.

The salt stress and substrate starvation affected the lipid production in the study. The lipid content increased from 17% to 38% as the concentration of NaCl increased from 0 to 400mmol/L were noted in previous studies on freshwater microalga *Chlamydomonas mexicana* grown on municipal wastewater with different levels of salinity<sup>194</sup>.

Salinity	Media	Na concentration (ICP) g/L	Lipid (% w/w) in biomass	Biomass production (g/L)	Biomass / Na+	Lipid / (biomass / Na+)
4.40%		2.8	56.2	0.8	0.3	0.16
3.50%	Reef salt	2.2	18.6	1.8	0.8	0.15
2.60%		1.6	53.4	1.0	0.6	0.32
4.40%		6.7	21.6	0.9	0.1	0.03
3.50%	Road salt	5.3	20.9	1.0	0.2	0.04
2.60%		4.0	17.0	0.9	0.2	0.04

 Table 4-2 Biomass and lipid to sodium conversation

This phenomenon was widely studied and explained noting the cellular lipid production is formed by the secondary metabolism, and is typical considered as the protective mechanism that all microalgae undertake during the various stress conditions<sup>195</sup>. During the substrate nitrate source depleted in the media cells derive nitrogen from the cellular storage like protein and leading to increase the cellular carbon levels. This increased level of cellular carbon used for the formation of the fatty acid and thus the lipid formation. Additionally, cells triggers the protective mechanism and the formation of lipid in response to the unfavorable environmental condition like changes in temperature, pH, osmotic level etc. <sup>196</sup> were also reported. Stressing the micro algal cell through salt may cause additional lipid accumulation metabolism than stressing through nutrients limitation were also noted <sup>89</sup>. Through our study we noted cultures experienced increased salt concentration from 2.6% to 4.4% exhibited increased total lipid accumulation (Figure 4.8). The mechanism of increased demand for nitrogen keeps healthy growth of cells under high salinity conditions to prevent cell membranes from being destroyed by changing osmotic pressure also result in lipid accumulation <sup>197</sup>. Salt stressing used in the fresh water based microalgae cultivation but is not reported for the marine microalgae. Typically the salinity of the seawater in the ocean ranges due to the evaporation and dilutions from 2.6 to 3.5 %. This fluctuation also anticipated during the open pond cultivation of micro algae. Salinity fluctuation plays a positive trend on cellular lipid accumulation and predator control were also noted <sup>187</sup>. Our study suggests marine

*microalgae N.gaditana* biomass production for biofuel might work best if the algae grown at lower salinity than the higher salinity and this technique allowed for the higher lipid accumulation rates. Varying the type of the base media reduced the biomass yield and lipid content maintained at 20.6% of the biomass and is in consistent to the studies reported of 19.3% <sup>198</sup> for the *Nannochloropsis* species.



Figure 4:8 Lipid concentrations (%w/w) of N. gaditana in the biomass

# 4.2.7 Effect media, salt and nitrate concentration on fatty Acid profile

Composition of fatty acid in the lipid predominantly determines the quality of biodiesel and its suitability to use as biofuel. Therefore, the lipids extracted from the cells were profiled by using the GC- FAME analysis and the fatty acids identified listed (Figure 4.9)

The highest percentage of FAME (31% and 39% of lipid recovered) observed in 3.5% reef and 4.4% road salt growth media when the biomass yield was highest. Elevating the media concentration from 3.5 to 2.6% decreased the FAME from 31% to approximately 28% in reef salt. While the steadily increased percentage of FAME from 35% to 39% observed as the concentration of road salt media increased from 2.6 to 4.4%. The

predominant fatty acids identified in the analysis were palmitic acid (16:0), palmitioleic acid (16:1) irrespective of the media concentrations. Long chain fatty acid (>C16) accounted over 90% with less than 5% relatively short chain fatty acids (< C14) of the total fatty acid in both media. Over 90 % of C 16 and around 3- 4% of C18 were detected in the biomass obtained by the cultivation in the reef salt media. Around 67% of C16 and around 20 % of C18 were detected in the biomass obtained 18 levels are used as the indicator for the selection of the biomass for the feed stock for biofuel <sup>199</sup>



Figure 4:9 Fatty acid profile of FAME in lipids of *N. gaditana* 

Percentages of palmitic acid (C16) tended to increase with elevating the media salt concentration from 3.5% in both the growth media; however, arachidic acid (C20) showed the contrary and was observed highest (28%) in 3.5% road salt media. Elevating the salt in media enhanced the percentage of oleic acid and arachidic acid. The road salt

media provided complementary condition for the production of the long chain fatty acid (>C16). Increased the media salt concentration increased the long chain fatty acid in both the media while the road salt accounted overall increased amount of long chain fatty acid and FAME indicating more favorable for the biofuel point of view.

The effect of initial media nitrate concentration on lipid production and fatty acid profile were plotted in Table 6 were evaluated on the biomass obtained from the varied initial nitrate concentration in 2.6 % road salt based media at 37.5, 75 and 100 mg/L sodium nitrate. Lipid profile of these cells illustrated the lower the nitrate level had the higher the long chain (>C16) fatty acid. Overall fatty acid profile comprised 92% of long chain fatty acid in 75 and 100 mg/L of sodium nitrate, while 94% of long chain fatty acid in lowest nitrate (37.5 mg/L) containing medium.

Factors such as temperature, irradiance and nutrient availability affect not only lipid content but also fatty acid composition in many algal species. Large amounts of neutral lipids, mainly including triacylglycerol (TAG), diacylglycerol (DAG), monoacylglycerol (MAG), sterol ester were noted in the N-stressed cells often.<sup>103</sup>. Nitrogen deficiency and salt stress induced the accumulation of C18:1 to some extent C20:5 were noted <sup>29</sup>. The study confirmed the suitability of the generated biomass for the biofuel production and the varied concentration on salt and nitrate resulted with increased level of long chain fatty acids.

#### 4.3 Conclusions

Phycoremediation using marine micro algae addressed in this study to provide the solution to treat the road salt run off efficiently as environmental friendly way with the generation of biomass which were influenced by the concentration of salt and nitrates in the growth media. Road salt run off enriched with leaching of soil fertilizer nutrients could be ideal for the marine microalgae cultivation to mitigate the negative effect of road salt run off on the environment. Nutrient removal of water run-off using marine microalgae will be the cost effective technique, which has the potential to generate biomass that can be further used as feedstock for biofuel. *Nannochloropsis gaditana* exhibited growth in all the tested road salt resulted with the production of  $\sim 1g/L$  biomass

and over 90 % nitrate reduction with in the 6 days. Increased level of salt concentration also increased the lipid accumulation and lipid profile. Results indicates that phycoremediation is one of the processing way that can treat the road salt run off and can be effectively used to generates valuable biomass in addition to removal of nitrate based nutrients from the waste water system that are difficult in the hyper saline waste water systems. Alternatively, road salt contains sodium chloride, which is the essential salinity provider to replace seawater. The limitation of marine microalgae cultivation only in seawater are eliminated and by introducing viability of these species to utilize in road salt run off and the biomass production from road salt run off system will provide economic feasibility of production of biofuel. The two salinity based approach in the experimental study provided valuable insight for an ecological approach to overcome the limitation of water for cultivation of microalgae for use in algal biofuel by producing valuable biomass and pigments with removal of pollutants from wastewater treatment system deeming it as a novel green and sustainable environmental biotechnology.

### Chapter 5

### 5 Investigation of three fed batch feeding strategies for biomass and lipid production of *N.gaditana* in Fed-batch cultivation

Improvement of lipid productivity in microalgae can be enhanced by production of biomass with higher concentration of lipids. Nutrient management in the culture system during the growth period is an important strategy to improve both the biomass and lipid productivity. Nitrogen limitation in the culture medium can increase the lipid synthesis in microalgae but reduces the biomass. The effect of controlled supply of nitrogen in fedbatch cultivation to improve the biomass and lipid production in N.gaditana was investigated in this study. Three modes of nutrient feeding in fed-batch cultures i.e., staged, pulse and continuous were investigated for their effects on the cell growth and lipid synthesis. The influence of two average light intensities (100  $\pm$ 10 and 200  $\pm$ 10  $\mu$ mol photons/m<sup>2</sup>/s) was also investigated in auto phototrophic cultivation. The maximum biomass of 2.08 g/L obtained by the feeding of 20 mg /d of nitrate at light intensity of  $100 \pm 100 \pm 100$  µmol photons/m<sup>2</sup> s with nitrogen to biomass conversion rate of 8.32 mg nitrate /mg biomass. The supply of nitrate by the continuous feeding mode did not improve the biomass yield in fed batch cultivation. A maximum lipid productivity of 46.84 mg/L/d was obtained when the nitrate supply of 10 mg/L fed at higher light intensity of 200  $\mu$ mol. photons/ m<sup>2</sup> /s in fed-batch. The feeding mode influenced the fatty acid compositions and increased the content of two long-chain fatty acids of palmitic acid (C16:0) and palmitolieic acid (C16:1) over the 70 % FAME

### 5.1 Introduction

The sustainable biofuel production systems from microalgae require high growth rate and lipid yields to replace the fossil fuels<sup>26,36</sup>. The algal lipid productivity determines the unit cost of the biofuel and is affected by the biological potential of the species and environmental conditions<sup>31,58</sup>. The choice of species and the design of culture system to provide the correct condition are the key factors plays important role in the cultivation. However the optimization of lipid productivity with the maximized growth rate and lipid

content is challenging due the contrasting effects of nutrients on biomass and lipid production <sup>58,101,200</sup>.

Microalgae exhibit considerable metabolic versatility and flexibility. The metabolic capability and environmental adaptability of the algal cells are modulated by the availability of nutrients, light, temperature and pH in the growth system. The choice of species and the design of culture systems provide suitable conditions to enhance biomass and lipid production for the biofuel production. Among the many reported studies, two species of green algae, i.e. *Chlorella* and *Nannochloropsis* have been shown to have the capability to produce higher lipid content ranging from 20-60% of the biomass weight and are considered as the most suitable species for the applications to biofuel production <sup>85,201</sup>.

Microalgae cultivation can be carried out in photoautotrophic, heterotrophic and mixotrophic modes based on the energy and carbon supply. Light and nutrient mediated growth enhancement for biomass productivity in photoautotrophic mode is considered as the most inexpensive strategy suitable for large-scale outdoor cultivation <sup>53</sup>. The demand of light energy and nutrients varies with algal growth stages. A lower level of light intensity is adequate during the initial stages of growth, whereas, a higher level of light is needed for moderate-high density cultures to avoid mutual shading of cells and to enhance photosynthesis efficiency <sup>101</sup>.

Nutrient nitrogen (N), phosphate (P) and carbon(C) are required at the specific ratio to the growth, biomass generation and lipid synthesis<sup>202</sup>. Nitrogen is a major component of proteins, enzymes, chlorophyll and pigments which regulates the cellular reaction and its concentration is regarded as the key factor to control the biosynthesis of lipids and its manipulation can leads to remarkable changes in lipid content and the fatty acid profile of microalgae<sup>203</sup>. Thus, a positive correlation exists between the nitrogen concentration of the growth media and the production of biomass which results in the reduced lipid accumulation<sup>93</sup>. The nitrogen limitation decreases the production of the growth promoting components and the cell utilizes nitrogen rich cellular components like proteins and carotenoids for the nitrogen supply.

Higher concentration of the nutrients in the initial growth phases therefore increases the biomass generation and the decreased nutrient levels at the later growth phase shifts metabolic cycles for the synthesis of the secondary metabolites such as lipids<sup>204</sup>. The microalgae cells respond to deprivation and limitation of nutrients by decrease in cell division, photosynthesis and respiration and an increased accumulation of storage compounds (starch, oil) and/or secondary metabolites <sup>58,203</sup>. Even though, nutrient deprivation conditions promotes the lipid accumulation in micro algal cells the reduced growth rate, reduces the overall lipid yield.<sup>9,204</sup>.<sup>95</sup>.

In microalgae, the structural membrane lipids are phospholipids synthesized during the active growth stage and these serve as carbon stocks during the limitation of nutrients in the growth stages <sup>31</sup>.The storage lipids are the neutral lipids which are accumulated as triacylglycerol (TAG) and are composed of long chain fatty acids suitable for biofuel(biodiesel)production. These are synthesized in the late growth stage using the carbon stocks from the structural lipids and other degraded biogenic components (Protein, chlorophyll, etc.,) <sup>36,95</sup>. Production of TAG is enhanced due to stress by adverse growth conditions<sup>53</sup>. Single variable substrate nitrogen control has been shown previously to manipulate the biomass and lipid production in autotrophic microalgae <sup>33</sup>. The supply of nutrient to provide different nutritional situation like repletion, sufficiency, limitation and deprivation conditions varies the composition of biomass with the balanced growth <sup>42,58,202</sup>.

A fed-batch cultivation system has the flexibility to control the chosen nutrient concentration by varying the flow rate and addition regimes using intermittent or continuous feeding during the growth stage to support the requirements of micro algal growth<sup>109,205</sup>. The continuous and pulse feedings of urea as a nitrogen source in fed-batch cultivation were used for the cultivation of *Spirulina platensis* provided the 1231  $\pm$  86 mg/l) and 1145 $\pm$ 52 mg/l biomass<sup>115</sup>. By employing low feed rates, the fed-batch system can operate with cells in the stationary phase of growth. Thus, a quasi-steady state culture system can be developed and these conditions support growth and trigger lipid production in the fed- batch cultivation system <sup>36,105,109</sup>. Fed-batch systems can operate using diverse nutrient strategies, such as periodic, continuous or pulsing nutrient addition.

Only limited studies have been carried out on different feeding modes and their effect on lipid accumulation in microalgae. The study proposes for the first time to investigate alternative feeding strategies for fed-batch cultivation of *N. gaditana*.

### 5.2 **Results and Discussion**

Improvement of lipid productivity in the microalgae can be enhanced by maximized production of biomass with higher concentration of lipid. Nutrient management in the culture system during the growth period is essential to improve the biomass and lipid productivity.

# 5.2.1 Effect of cultivation methods and feeding rates nitrate conditions for media on nitrate uptake and biomass

#### 5.2.1.1 Batch cultivation

The nitrate uptake process by the cells during the growth time were determined by the residual media nitrate concentration (Figure 5.1and 5.2) to identify the nitrate-depleted states in the media at two light intensities of  $100 \pm 10$  and  $200 \pm 10\mu$ mol. photons/ m<sup>2</sup>/s<sup>)</sup> for the production of biomass and lipid. Microalgae are predominantly known for their capability of rapid assimilation of nitrogen even at low concentration. Nitrate consumption by the cells progressively increased during the growth period with the reduction to < 10 mg/L in the media within the 8 days of cultivation. Increased supply of light of 200 µmol. photons/ m<sup>2</sup>/s<sup>1</sup> improved the nitrate utilization rate reaching the 100% depletion in the media. However increasing the initial nitrogen were observed in the reactors with lower nitrate concentration (75 and 50 mg/L) at 200 µmol. photons/ m<sup>2</sup>/s<sup>1</sup> light intensity. The batch culture cultivation continued for 20 days after the nitrate depleted conditions to investigate the impact of the nitrate depleted condition on the biomass and lipid production.



Figure 5:1 Media nitrate concentration in batch culture media at 100  $\mu$ mol photons/ m<sup>2</sup>/s light intensity



Figure 5:2 Media nitrate concentration batch culture media at 200  $\mu$ mol photons/ m<sup>2</sup> /s light intensity

The effects of different nitrate concentration in the batch culture on the production of biomass were investigated by determining the biomass dry weight (DW) during the growth (Figure 5.3 and 5.4).

The increases in the biomass observed from the day 2 of the cultivation time indicated the increased assimilation of nitrates supported the cellular growth. The growth curves showed a very short lag phase and followed the dynamic growth between 4 to15 days in all initial nitrate levels. The continued productions of biomass were observed even after the nitrate depletion conditions (<10 mg/L). The maximum biomass of 1.39, 1.17, and 1.045 g/L at the end of the cultivation were obtained in the media with 100, 75, 50 mg/L initial nitrate N concentration respectively under the 100 light intensities. The increased light intensity (200) in the batch cultivation increased the biomass to 1.59 and 1.275g/L in the media with 75 and 100mg/L initial nitrate concentration. The lowest biomass of 0.90 g/L obtained in reactor with lowest nitrate N concentration (50mg /L) at the highest light intensity 200 µmol. photons/ m<sup>2</sup>/s.



Figure 5:3 Biomass concentrations of 100- $\mu$ mol photons/m<sup>2</sup>/s light intensity in batch

culture



Figure 5:4 Biomass concentrations of 200  $\mu$ mol photons/m<sup>2</sup>/s light intensity in batch culture

The increases in the biomass observed from the day 2 of the cultivation time indicated the increased assimilation of nitrates supported the cellular growth. The growth curves showed a very short lag phase and followed the dynamic growth between 4 to15 days in all initial nitrate levels. The continued productions of biomass were observed even after the nitrate depletion conditions (<10 mg/L). The maximum biomass of 1.39, 1.17, and 1.045 g/L at the end of the cultivation were obtained in the media with 100, 75, 50 mg/L initial nitrate N concentration respectively under the 100 light intensities. The increased light intensity (200) in the batch cultivation increased the biomass to 1.59 and 1.275g/L in the media with 75 and 100mg/L initial nitrate concentration. The lowest biomass of 0.90 g/L obtained in reactor with lowest nitrate N concentration (50mg /L) at the highest light intensity 200 µmol. photons/ m<sup>2</sup> /s. The Lipid content of the biomasses was analyzed at the end of the cultivation. The lower initial nitrate increased the lipid content but the lower light intensity resulted with the lower lipid contents in the cells. The lipid of 17, 20.6 and 22.0 and 35.7, 39.7 and 42.7(%DW) were obtained in 100, 75 and 50 mg/L of initial nitrate concentration at 100 and 200  $\mu$ mol.photons/ m<sup>2</sup> /s respectively. The increased light intensity doubled the cellular lipid concentration.

The nitrogen is the key nutrient for the production of various biogenic components to support the growth in microalgae. Three Nitrogen treatments with nitrate concentration levels (50, 75 and 100mg/L) of greenhouse effluents to provide the nitrate depletion conditions for *Nannochloropsis gaditana* growth model were studied first in the batch culture using road salt run off.

#### 5.2.1.2 Fed batch cultivation Microalgae Culture and media

#### 5.2.1.2.1 Pulse feeding Fed batch cultures

Nitrate consumption by the cells of *N. gaditana* was observed as the concentration of the nitrate in the media nitrate decreased rapidly from day one. The reduction of 99% nitrate observed in day 8 in batch cultures were used to initiate the Fed-batch culture systems. On day  $8^{th}$  the nitrate repletion conditions were introduced to the growth media by the feeding of nitrates. The pulse feeding of the nitrates were operated in two sets of 100 and 200 µmol. photons/ m<sup>2</sup>/s light intensities in five 6L photo bioreactors with steady level of

initial nitrate concentration (100mg/L) at 5 feeding rates of nitrate concentration ranging from 10, 12.5, 15, 17.5 and 20 mg/day using stock solution of 100 mg/L Nitrate N containing stimulated greenhouse effluents by following the alternative days of feeding regimes. The nitrate consumption rates were evaluated (Figure 5.5 and 5.6 by analyzing the residual nitrate N concentration of the growth media.



84









Figure 5:6 Nitrate concentrations in pulse feeding fed batch culture at 200  $\mu mol$  photons/  $m^2$  /s light intensity

Switching the batch reactors to the pulse feeding mode resupplied nitrates to the nitrate depleted media and established the nitrate replete conditions to enhance the biomass production. The increased nitrate N concentration approximately to 8-12 mg/L provided nitrate replete conditions. The oscillating of the nitrate N level in the media were observed with the feeding time indicated the continued nitrate assimilation process by the starved cells with the rapid decline in the nitrate concentration prior to the pulse feeding. The increased supply of light intensity of from 100 to 200  $\mu$ mol. photons/ m<sup>2</sup>/s<sup>1</sup> further improved the process of nitrate assimilation processes in the cells with an approximate nitrates levels oscillated between <10 mg/L. The introduction of pulse feeding to maintain the nitrogen replete condition promoted the biomass growth (Figure 5.7 and 5.8).



Figure 5:7 Biomass concentrations in pulse feeding at  $100\mu$ mol photons/m<sup>2</sup>/s light intensity

The increasing the rate of pulse feeding increased the nitrate levels in the growth media increased the cellular growth and the biomass of 1.56,1.69,1.91,1.96 and 2.08mg/L were obtained at 100  $\mu$ mol photons/ m<sup>2</sup> /s<sup>1</sup>. The increased supply of light generated slightly lower biomass 1.365, 1.445, 1.625, 1.88 and 1.925 g/L biomass at 200  $\mu$ mol photons/ m<sup>2</sup>

/s<sup>1</sup> at 10, 12.5, 15, 17.5 and 20 mg/day feeding rates of Nitrate N concentration. The increase in the feeding rate increased the biomass but the increased light of 200  $\mu$ mol photons/ m<sup>2</sup>/s resulted the lower total biomass in comparison to the biomass at 100  $\mu$ mol photons/ m<sup>2</sup>/s at the entire nitrate feeding rates.



Figure 5:8 Biomass concentration in pulse feeding fed batch culture at  $200\mu$ mol photons/m<sup>2</sup>/s light intensity

Lipid contents of 25.0,22.9,21.36.20.6, 20.2(%DW) were obtained at 100 light  $\mu$ mol photons/ m<sup>2</sup>/s at 10,12.5,15,17.5 and 20 mg/day feeding rates of Nitrate N concentration. The increased light supply of 200 light with introduction of pulse feeding increased the

lipid content to approximately 2 times with total of 58.3,52.5,50.4,49.8 and 45.1(%DW) were obtained 10,12.5,15,17.5 and 20 mg/day feeding rates of Nitrate N concentration (Table 5.4).

#### 5.2.1.2.2 Continuous feeding fed batch cultures

In the third set of experiments under continuous feeding fed batch cultivation induced the nitrate sufficient conditions. The experiments in continuous mode at above feeding rates ranging were performed by operating two sets of reactors with average light intensities of 100 and 200  $\mu$ mol.photons/ m<sup>2</sup> /s<sup>1</sup> in five reactors simultaneously to provide the nitrate sufficient conditions. The continuous feeding of the stock solutions was introduced at 8<sup>th</sup> day similar to the pulse feeding when the nitrate concentrations were diminished to < 10 mg/L. The concentrations of nitrogen in alternative days (Fig 5.9 and 5.10) were analyzed to identify the nitrate sufficient condition in the growth media. The nitrate concentration in the media continuously decreased as the cultivation time progressed in all the feeding rates. The concentrations below 10mg/L rate were noted in all the reactors.

The higher residual nitrate concentrations were observed in the reactors that are supplier with higher light in comparison to the reactors with the low light intensity. The residual nitrate concentration of <10 mg/L observed at the end of the 20 day cultivation indicating there were steady nitrate sufficient conditions were maintained for the cells.

The nitrate concentration in the media continuously decreased in all the feeding rates. The concentrations below x rate were noted in all the reactors. The higher residual nitrate concentrations were observed at the higher light supply in comparison to the reactors with the light intensity. The residual nitrate concentration of <10 mg/L observed at the end of the 20 day cultivation indicating there were steady nitrate sufficient conditions were maintained for the cells.'



Figure 5:9 Growth media nitrate concentration in continuous fed FB culture at 100  $\mu$ mol photons/ m<sup>2</sup> /s light intensity




Figure 5:10 Nitrate concentration during the continuous fed FB batch culture at 200  $\mu$ mol photons/ m<sup>2</sup>/s light intensity

The generation of biomass decreased as the feeding rates increased in the continuous feeding fed batch cultures and with increased light intensities (Figure 5.11and 5.12). The highest biomass of 1.42 g/L generated at the lowest feeding rate (10 mg/d) and with 100  $\mu$ mol photons/ m<sup>2</sup>/s<sup>1</sup>. The increased light to 200  $\mu$ mol photons/ m<sup>2</sup>/s<sup>1</sup> resulted in the slight reduction of biomass to 1.39 g/L.

The increased supply of nitrate with feeding increased the biomass levels were observed at the lower light irradiation. The lowest values of biomasses of 1.1 g/L at 17.5 g/ d feeding rates at the lower supply of light and at 20 mg/L with higher supply light were observed.



Figure 5:11 Biomass concentration in continuous fed FB batch culture at 100  $\mu$ mol photons/m<sup>2</sup>/s light intensity



Figure 5:12 Biomass concentration in continuous fed FB batch culture at 200  $\mu$ mol photons/m<sup>2</sup>/s light intensity

The nitrates dose of 17.5 mg/ L resulted with the lowest (1.09 mg/L) generation of biomass, which was below the total biomass concentration obtained in the batch cultivation. The lipid concentration in the cells reduced from 18.6% to 9.62 % as the feeding rates increased from 10 to 20 mg/L by the continuous feeding modes (Table 5.4).

#### 5.2.1.2.3 Stage feeding fed batch culture

In the last part of this study, a stage feeding strategy for the fed batch was performed for *N. gaditana* cells to induce the prolonged conditions of nitrate depletion prior to the nitrate replete conditions to influence the lipid production. This experiment was performed in the two sets with average impinging irradiance 100 to 200  $\mu$ mol. photons/m<sup>2</sup>/s<sup>1</sup> in three 6L photo bioreactors with initial volume of 2L with feeding 500 ml of stock solution containing 100 mg/L nitrate at 5,10 and 15 days. The reactors set up with three initial nitrate concentrations of 100, 75 and 50 mg/ L similar to the levels of the batch cultivation reactors. The initial levels of nitrate concentration and supplementation with stage feeding affected the availability nitrogen to the cells in the growth media. Nitrate concentrations of the media are noted in the Figure 5.13 and 5.14



Figure 5:13 Media nitrate concentration in stage fed FB batch culture at 100  $\mu$ mol photons/m<sup>2</sup>/s light intensity



Figure 5:14 Media nitrate concentration in stage fed FB batch culture at 200  $\mu$ mol photons/m<sup>2</sup>/s light intensity

The average residual nitrate concentration of <2 mg/L were noted prior to the stage feeding at day 10 and 15 and the concentration bounced to 7- 10 mg/L levels indicating the nitrate depleted and replete conditions were established in the growth media.

The reactor that supplied low light intensity resulted slow nitrate intake rates and the low initial nitrate N concentration further reduced the uptake rate with the residual nitrate concentration of 4-5 mg/L remained at the end of the cultivation in the reactor with <10mg/L. The stage feeding to the batch cultivated *N.gaditana* at three initial nitrates levels of 50, 75and100 mg/L in the media reintroduced the nitrates.

The increased biomass productions with the growth were observed (Figure 5.15 and 5.16). The stage feeding accompanied by the lower light intensity resulted the maximum production of biomass (1.5 g/L) when the initial nitrate concentration was at highest level (100mg/L). The increased light intensity reduced the biomass and the lowest of 0.7g/L obtained at lowest (50 mg/L) initial nitrate concentration.

The reintroduction of nitrates at the various growth stages influenced the lipid production when the nitrate concentration was higher (100mg/L) at the low light intensity. The other sets of the reactor not showed in the improvements in the lipid concentrations except in the initial nitrate 75 and 50 mg/L slightly decreased lipid concentrations observed with the introduction of the stage feeding to the nitrate depleted cells.

The lipid concentration varied from 18-18.5 (%DW) in the low supplied light and lipid content increased with the increased light to 35-42 (%DW) at all the initial nitrate concentrations (Table 5.4). Improvement of lipid productivity in the microalgae can be enhanced by maximized production of biomass with higher concentration of lipid. Nutrient management in the culture system during the growth period is essential to improve the biomass and lipid productivity.



Figure 5:15 Biomass concentration in stage fed batch FB culture at 100- $\mu$ mol photons/m<sup>2</sup>/s light intensity



Figure 5:16 Biomass concentration in stage fed FB culture at 200- $\mu$ mol photons/m<sup>2</sup>/s light intensity

#### 5.2.2 Effect of culture conditions on biomass

The goal of the study was to optimize the biomass productivity with increased lipid contents by managing the nitrogen concentration in the growth media using various culture systems. The change in the supply of light and the initial nitrate concentrations provide the varied the nitrate conditions in the growth media during the growth period. The continued reduction of nitrate in all the conditions can be attributed to the requirement of nitrogen by the cells for synthesis of various biogenic compounds to support the cell replication process to enhance the growth. The nitrate assimilation processes for the use age of cellular activities were noticed in all the cultures. Nitrogen uptake are influenced by the various factors like, cell physiology, metabolic flux, concentration of the available nitrate and the light irradiation in the photoautotrophic cultivation system. The stoichiometry ratio of N/P and N/C in the growth media significantly affects the growth.

Based on the trends displayed in the batch cultivation, the biomass growth continued after the cells experienced the nitrate depletion in the media indicating the use of containing components are utilized as cellular survival mechanism. The rate of nitrate depletion in the media with the increased supply of the light intensity in the autotrophic cultivation mode increased the production of biomass confirms the positive correlation of cellular nitrate and the biomass growth. The nitrate-depleted condition attained at day 8 and the concentration of 100 mg/L in the batch culture was considered as optimum levels were used further for the investigating the feeding strategies.

In general maximum biomass concentration as dry weight achieved in fed-batch was higher than in batch culture mode, as shown on Table 5.2. In terms of feeding strategy pulse feeding in fed-batch system resulted in the maximum biomass of 2.08 g/L. However stage feeding was efficient in achieving the biomass of 1.505 g//l and was further depended on the higher initial nitrate concentrations and the reduction in the initial nitrate concentration (50 mg/l) generated the lowest biomass of 0.7 g/L below than the batch culture mode at the higher (200  $\mu$ mol. photons/ m<sup>2</sup>/s) light intensity.

Increase in the nitrate concentration and light intensity increased the biomass production in the batch cultivation only. Increasing the light intensity increased the cell mass to 1.14 fold and the 1.6 fold increment observed with increasing the initial nitrate concentration from 50 to 100 mg/L indicating both the nitrate and lights were efficient in promoting the cell growth. Light is the source of energy enhances the photosynthetic carbon fixation in the autotrophic organisms with the abundant supply of N for the syntheses of biogenic compounds. The reduced the initial nitrate concentration to 50 mg/L produced the lowest biomass and the influence of increased light intensity was not efficient due to the low availability of nitrogen limited to support the synthesis of components to improve the biomass production. The insufficient availability nitrate resulted in inefficient use of light to promote the cell growth. Unavailability of sufficient nitrogen concentration in the cytosol hindered the biochemical reactions to produce the nitrogen containing auxiliary cellular light harvesting pigments like chlorophyll to harvest the light energy might be the reason for the reduced generation of biomass.

The trends of increased biomasses were attained with the introduction of pulsed feeding fed batch cultivation. The increased feeding levels in parallel increased the biomass with maximum biomass of 2.08 g/L were obtained at the 20 mg/day feeding rate of nitrate. The influence of light intensity was not as efficient in the fed batch cultivation as in the batch cultivation since the dilution of the culture media by the pulse feeding improved the light availability to the cells. The continuous feeding strategy with the increased feeding rate reduced the biomass concentration in both light intensities. The decrease in the biomass concentration can be attributed to the continued increment in the dilution was sufficient to enhance the light penetration to the cells to carry the photosynthesis the increased light intensity did not resulted with increased cell growth. Similarly, stage-feeding strategy resulted with trends of reduced biomass with increased light. However, the increased initial nitrate concentration promoted the cell growth and biomass production.

In these studies, the higher amounts of biomasses were produced when the media fed with higher nitrogen concentration. This indicates that the application of nitrate nitrogen by the fed-batch culture mode with promoted the cell divisions and biomass accumulation. Among fed-batch treatments, pulsed feeding result the highest biomass concentration. Results could indicate that a good cultivation strategy to obtain high biomass concentration will be to use pulsed feeding to induce nutrient replete conditions and feeding of 20 mg /L should be aimed to return nutrients like nitrogen to maintain the cell growth.

# 5.2.3 Effect of feeding on nitrogen to cell conversion factor

To understand the effect of nutrient feeding in the *N.gaditana*, cell to nitrogen conversion factor (Y=X/N) were calculated. The results presented in Table 5.1 indicate that highest cell to nitrogen conversion factor (Y=X/N) of 20.9 where accounted when lower total mass of nitrate (50 mg/L) were added per unit volume values and at high light intensities in the batch cultivation.

In other words, the cell to nitrogen conversion factor increased with the increase in the light intensity and decreased with the increase in the total mass of nitrate. In general, the increased supply of light with decreased amount of nitrate feeding reduced the generation of cell mass. While, in batch cultures, at low light intensity, the increase in the nitrate led to a decreased cell to nitrogen conversion because the cellular growth is limited by the light intensity. The overall cell to nitrogen conversion factors were low in the fed batch cultivation systems since the feeding regimes increased the total mass of the nitrate added which did not increase the biomass proportionately. The cell to nitrogen conversion factor ranged from 6 to 8.5 in the pulse fed batch, while the 4 -6 and 6-8.5 were obtained in the continuous and stage fed batch respectively.

Table 5-1 Biomass to substrate nitrogen conversion rates in the batch and fed batch cultivations at light 200  $\mu$ mol photons/ m2/s

Cultivation mode	Feeding rate (ml/day)	Initial nitrate concentration (mg/L)	Total nitrate (mg/L)(N)	Biomass (g/L)(X)	Biomass to substrate conversion ratio (Y=X/N)
		100	100	1.39	13.90
Batch		75	75	1.18	15.67
		50	50	1.05	20.90
Pulse- Feeding Fed batch	100	100	233.33	1.56	6.69
	125	100	238.46	1.69	7.07
	150	100	242.86	1.91	7.84
	175	100	246.67	1.96	7.93
	200	100	250	2.08	8.32
Continuous feeding fed batch	100	100	233.33	1.42	6.09
	125	100	238.46	1.30	5.43
	150	100	242.86	1.24	5.11
	175	100	246.67	1.10	4.44
	200	100	250	1.15	4.60
Stage feeding * (500ml/5 days)	100*	100	242.86	1.51	6.20
	100*	75	192.86	1.30	6.74
	100*	50	142.86	1.21	8.47

## 5.2.3.1 Specific growth rates in batch and fed-batch cultures

Specific growth rate is a useful index to evaluate culturing methods based on the initial and terminal cell biomass. *Nannochloropsis* has a very slow growth rate. The specific growth rates were ranged from 0.03- 0.09/day in the studied growth conditions (Table 5.2). Batch cultivation with increased nitrate concentration and light intensity resulted with maximum specific growth rates of 0.08/day and pulse fed batch with maximum of 0.12 /d in increased feeding rate of 20mg/L of nitrate concentration in the lower light intensity. Similarly increased nitrate feeding rate with increased light resulted with reduction of the specific growth rates in continuous and stage feeding fed-batch cultivation. In the stage feeding the initial nitrate concentration is important to increase the specific growth rate. The maximum specific growth rate of 0.7 /d at the 100 mg/L

nitrate concentration and the least of 0.03 /d were obtained in the 50 mg/L initial nitrate concentration.

		Light intens	ity -100µmol	Light intensity -		
			photons/ m <sup>2</sup>	/s	200µmol photons/ m <sup>2</sup> /s	
Cultivation mode	Feeding rate (ml/day)	Initial nitrate concentration ( mg/L)	Biomass (g/L)	Specific growth rate (per day)	Biomass (g/L)	Specific growth rate (per day)
Batch		100	1.39	0.07	1.59	0.08
		75	1.18	0.06	1.28	0.06
		50	1.05	0.05	0.90	0.04
	100	100	1.56	0.08	1.37	0.06
Dulsa Easding Ead	125	100	1.69	0.09	1.45	0.06
batch	150	100	1.91	0.11	1.63	0.07
	175	100	1.96	0.11	1.88	0.09
	200	100	2.08	0.12	1.93	0.09
Continuous feeding fed batch	100	100	1.42	0.05	1.39	0.03
	125	100	1.30	0.04	1.32	0.04
	150	100	1.24	0.04	1.18	0.03
	175	100	1.10	0.03	1.12	0.03
	200	100	1.15	0.05	1.08	0.03
Stage feeding * (500ml/5 days)	100*	100	1.51	0.07	1.32	0.06
	100*	75	1.30	0.06	0.90	0.04
	100*	50	1.21	0.06	0.70	0.03

Table 5-2 Specific growth rate and Biomass concentration in various cultivation methods

Increased light supply did not correlate with the increasing of the specific growth rates in all the experiments. The specific growth rates were reduced in the pulse, continuous and stage fed FB cultivation with increased light supply as these treatments also provided lower biomass. In most of the studies thus far, the reported specific rates were usually below 0.4/day for the *Nannochloropsis* species and the studied treatments were provided the lower specific growth rates. The ample supply of light and nitrate might be an effective way to boost photosynthetic efficiency to promote the weight of the biomass accumulation but the not the cell reproduction (daughter cells) as the specific growth rates were derived by the cell counts. In batch cultivation, the decreasing light intensity available per cell as a result of the increase in cellular biomass often leads to light limitation for algal growth. However, when cells are grown in continuous cultivation,

growth can be controlled by the fixed operational conditions that provide a steady supply of nutrients and allow for continuous cell density and light distribution. The dilution factors directly reduce the cell densities of the culture media, which increases the incident lights to the cell surfaces. The higher light intensity cause the photo inhibition due to the damages to the photo systems thereby the cell growth is reduced.

The amount and timing of N feeding in fed batch culture had significant effect on the biomass productivity. The nitrate supply increased the biomass production in the pulse fed batch cultures when compared to the batch. The process of photosynthesis in auto phototrophic microalgae can explain the influence of nitrogen supply on the biomass growth. The cellular responses vary with the availability nitrogen. Nitrogen is an essential nutrient and is critical for the synthesis of proteins, nucleic acids, and chlorophyll molecules. The green pigment chlorophyll "a" is the only pigment in *Nannochloropsis* species plays a central role in photosynthesis for absorbing and transferring light energy. Chlorophyll accumulation was higher in autotrophic cultivation and lower in heterotrophic cultivation explained as the more degree of chlorophyll needed for efficient light absorption when light is the sole energy source. Since the treatments in the cultivation provided sufficient supply of nitrogen and the synthesis of pigments were not disturbed. The sufficient nitrogen conditions in the all the cultivation systems improved the use of light energy and enhanced the growth and biomass generation.

Although light is important for algal growth, levels that are too low or too high limit or inhibit the photosynthetic process. Production of chlorophyll depends on the availability of the nutrients especially the nitrogen in the media. Nitrogen deficiency reduces the chlorophyll contents in microalgae which declines the growth rate and chlorophyll content were reported for *C. vulgaris*<sup>178</sup>.

The nitrogen-starved cells responses by the cessation synthesis or gradual degradation of chlorophyll during the nitrogen depleted conditions as noticed in the batch cultures. Apart from the supply of nitrogen, the availability of micronutrients and phosphorus affect the chlorophyll concentration, since the phosphorous is essential nutrient for many phosphorylation-dependent reactions like ATP synthesis and the Calvin cycle and its deficiency affects chlorophyll synthesis and cell growth and metabolism. The reduction in chlorophyll content of N-starved micro algal cultures associated with the lack of nitrogen atoms that form part of the chlorophyll structure, which is composed of four pyrrole groups forming a ring around a magnesium atom, which is stabilized by four nitrogen atoms<sup>98</sup>.

# 5.2.4 Effect of cultivation methods on lipid production

It is possible to change the biochemical composition of microalga by modifying the cultivation mode. Lipid production kinetics determined with the growth at day 10, 15 and 20 days of batch cultivation in the biomass obtained from the initial nitrate concentration of 100 mg/L (Table 5.3). As noted in the table 5.3, the lipid content increased as the growth progressed from 12.30 to 27.05 (%DW) in the cells, as the nitrate-depleted conditions were prevalent in the media in the batch cultures.

The increased light intensity enhanced the lipid production during the growth stage (Table 5.4). Fed batch cultures were supplied with nitrate feeding below20 mg/din all the treatments in order to maintain the nitrate depleted conditions (<10 m/L) to promote the lipid synthesis process.

	Light 100µmol photons/ m /s	Light 200µmol photons/ m2 /s		
Cultivation days	Biomass (mg/L)	Lipid (%DW)	Biomass (mg/L)	Lipid (%DW)
Day 10	0.52	12.3	0.88	17.65
Day 15	0.88	17.39	1.09	23.79
Day 20	1.36	27.05	1.59	35.73

Table 5-3 Lipid and biomass production in various cultivation stages in the batch culture with starting nitrate concentration of 100 mg/L

In order to determine the effects of cultivation and feeding regimes on lipid productivity the biomass (X), lipid content (P), volumetric lipid concentration ( $P_{Vol}$ ) and Lipid productivity ( $Q_P$ ) and lipid yield on nitrate (Table 5.4) were compared.

The lower intensity of light resulted with lower concentration of the cellular lipids in all the cultivation modes. The increased light intensity improved the lipid content in the cells irrespective of nitrogen availability.

The maximum concentration of lipid ranging from 45 to 58 (%DW) obtained under the pulse feeding in combination with increased light. The pulse fed-batch cultures with increased light yielded the highest volumetric lipid concentration ranging from 796 to 937mg/l was able to increase the lipid productivity to the ranges of 38 to 47 mg/L/day since the lipid contents in these treatments increased 2 fold. All the other culture treatments were showed the lipid productivity below 25 mg/L/day. The trends of overall increased lipid concentration were also observed as the availability of nitrates was reduced severely in the culture media except the continuous feeding fed batch cultures where the lowest biomasses were produced.

The continuous feeding fed batch cultures maintained the lipid concentration at 19-20(%DW) at the low supply of light and varied to 18- 35 (%DW) at the higher light intensity. The timing of nitrate feeding by the stage feeding (during log, linear and decline phase) in the fed batch culture system affected the lipid content that depended on the starting concentration of the nitrates with lipid yields and productivities were in comparable to the continuous fed batch cultures.

The yield of lipid on nitrate (g/g) ranged from 0.2 to 0.8 in the fed batch cultures and the yield of lipid on nitrate were reached the maximum of 7.68 in the batch cultures when the light supply was highest due to the total nitrate input to the culture system was at the lowest levels in the batch cultures.

# Table 5-4 Comparison of difference cultivation regimes for biomass (X), Lipid content (P), volumetric lipid concentration (Pvol.) and lipid productivity (QP) and yield per g of nitrate

Cultivation	Light	Biomass	Lipid	$P_{Vol}$	$Q_P$	Yield of
	(µmol photons/	(X)(g/L)	(P)	(mg/L)	(mg/L/d	Lipid
	m2 /s)		(%DW)		ay)	on nitrate
						(g/g)
Batch at 100 mg/L Nitrate	100	1.39	17	236	12	2.36
Batch at 75 mg/L Nitrate	100	1.18	21	242	12	3.22
Batch at 50 mg/L Nitrate	100	1.05	22	230	12	4.61
Batch at 100 mg/L Nitrate	200	1.59	36	568	8	5.68
Batch at 75 mg/L Nitrate	200	1.28	40	506	25	6.75
Batch at 50 mg/L Nitrate	200	0.9	43	384	19	7.68
Fed batch -pulse-10mg/day nitrate	100	1.57	25	391	20	0.39
Fed batch -pulse-12.5mg/day nitrate	100	1.69	23	386	19	0.31
Fed batch -pulse-15mg/day nitrate	100	1.91	21	407	20	0.27
Fed batch -pulse-17.5mg/day nitrate	100	1.96	21	402	20	0.23
Fed batch -pulse-20mg/day nitrate	100	2.08	20	420	21	0.14
Fed batch -pulse-10mg/day nitrate	200	1.37	58	796	40	0.80
Fed batch -pulse-12.5mg/day nitrate	200	1.45	53	759	38	0.61
Fed batch -pulse-15mg/day nitrate	200	1.63	50	819	41	0.55
Fed batch -pulse-17.5mg/day nitrate	200	1.88	50	937	47	0.54
Fed batch -pulse-20mg/day nitrate	200	1.93	45	868	43	0.29
	100	1.40	10	263	12	0.00
Fed batch -continuous-10 mg/day nitrate	100	1.42	19	203	13	0.26
red batch -continuous-12.5mg/day	100	1.20	20	261	12	0.21
Ead batch continuous 15 mg/day nitrate	100	1.50	20	241	15	0.21
Fed batch continuous-15 llig/day littrate	100	1.24	19	241	12	0.10
nitrate	100	1 10	10	205	10	0.12
Fed batch _continuous_20mg/day nitrate	200	1.10	19	202	10	0.12
Ted batch -continuous-zonig/day intrate	200	1.15	10	202	10	0.10
Fed batch -continuous-10 mg/day nitrate	200	1.39	32	440	22	0.44
Fed batch -continuous-12.5mg/day	200	1.00	24	454	22	0.04
nitrate	200	1.32	34	416	23	0.36
Fed batch -continuous-15 mg/day nitrate	200	1.18	35	416	21	0.28
riteste	200	1.12	20	331	17	0.10
Intrate	200	1.12	50	201	17	0.19
ed batch -continuous-20mg/day nitrate	200	1.08	27	294	15	0.15
Fed batch - stage feeding ( 3x 50 mg/L						
and initial 100 mg/L nitrate	100	1.51	25	377	19	0.24
Fed batch - stage feeding (3x 50 mg/L				240		
and initial 75mg/L nitrate)	100	1.20	29	340	17	0.22
Fed batch - stage feeding (3x 50 mg/L				116		
and initial 50 mg/L nitrate)	100	1.30	32	410	21	0.26
Fed batch - stage feeding (3X50 mg/L				162		
and initial 100 mg/L nitrate)	200	1.32	35	405	23	0.29
Fed batch - stage feeding (3 x 50 mg/L				347		
and initial 75mg/L nitrate)	200	0.90	38	342	17	0.21
Fed batch - stage feeding ( x 50 mg/L				204		
and initial 50 mg/L nitrate)	200	0.70	42	294	15	0.18

## 5.2.4.1 Lipid productivity

The best culture strategy in terms of volumetric lipid productivity ( $P_{Vol}$ ) and yield ( $Q_P$ ) and yield on nitrate (table 5.5) was pulsed fed batch culture. The light intensity of 200 µmol photons/ m2 /s provided the highest productivity, yield and yield of lipid on nitrate and the light intensity of 100 µmol photons/ m2 /s in the same cultivation mode provided the second best condition. The batch cultures provided the highest yield of lipid on nitrate with the lowest lipid productivity ( $P_{Vol}$ ) and yield ( $Q_P$ ) values. N replete condition maintaining by the continuous feeding with lower nitrate feeding and lower intensity were not efficient cultivation methods to increase the lipid productivity ( $P_{Vol}$ ) and yield ( $Q_P$ ) and yield on nitrate despite of having the moderately higher biomass, the continuous supply of nitrate to the media did not promote the conditions for the lipid synthesis.

The increased trend of lipid production observed with increased supply of light intensity. The excessive light energy promotes the lipid production to avoid photo-oxidative damage in the algal cells were reported<sup>41</sup>. The optimal light intensity for algal growth, product accumulation, and other applications varies significantly among different algal species ranges from 62.5 to 2000- $\mu$ mol m<sup>2</sup>/s. The study conducted by increasing the light intensity to 200 $\mu$ mol  $\mu$ mol m<sup>2</sup> s<sup>1</sup> in the reactors noticed the influence able impact on the lipid concentration in N. gaditana. Algal light absorption and conversion efficiency is species specific and can significantly vary due to inherent properties such as the pigment profile, cellular architecture and chloroplast arrangements. Various factors such as light spectra, irradiance level and the nature of microalgal photochemical machinery affect the photosynthetic responses and the accumulation of organic matter in various microalgae. These variables govern the biomass production and overall biochemical concentration of algal cells. The lipid biosynthesis is the secondary metabolisms that are triggered during the adverse growth conditions as the survival mechanism to protect the cells from light and nutrient starvation. Since the lipid synthesis reaction consumes higher NADPH (58), which serve as electron sink which relaxes the electron transport fluxes that were occurred by the increased supply of light as electron. Also, the lipid molecules serve as energy and carbon reserve to the cells during the adverse growth conditions.

The phenomenon of increased lipid content related to the ratio of the carbon/nitrogen were reported in the studies were not provided in the treatments. Only the light and the nitrates were studied since the increased carbon supply generally switches the metabolic pathways to starch synthesis which increases the carbohydrate contents of the cells but not the lipid. Fed-batch cultures with high glucose concentrations (50g/L) and intermittent nitrate feeding promoted specific carbohydrate accumulation. There are theories on the shifts of the metabolic pathway indicating the enzyme potentially hydrolyzes starch chains to channel the carbon flow to lipid synthesis in *N.oleoabundans* cultivation under prolonged nitrogen starvation conditions <sup>189</sup> were the possible reason for the increased concentration of the lipid in the study.

Secondly the studies reported were noted, depending on the intracellular N concentration, the enzymes modulate the activities. Nitrogen-limited conditions trigger the overproduction of proteins to scavenge internal and external nitrogen <sup>189</sup> and to activate the lipid synthesis pathways were can be explained to increased lipid production in the treatments.

# 5.2.4.2 Fatty acid composition of micro algal lipid

The study aimed to utilize the biomass of *N. gaditana* to the production of biofuel. The biomass obtained in the various feeding rates of fed batch cultures were collectively further analyzed for the fatty acid profiles. The lipids extracted from biomass from four cultivation strategies were trans esterified to fatty acid methyl ester (FAME) and analyzed by gas chromatography. The compositions of the fatty acids are summarized in Figure 5.17.



5:17 Percentage of fatty acids in FAME of the lipids

The total FAME content varied between 31- 39% of the total lipids. The major constituents were the two long-chain fatty acids of palmitic acid (C16:0) and palmitolieic acid (C16:1) and these two fatty acids contributed to more than 60% of FAME. The concentration of palmitic and palmitolieic acid accounted highest of 75 and 77.4% in the FAME of the lipid obtained from the biomass of pulse feeding and continuous feeding fed batch respectively. Interestingly the concentration (>35% of FAME) of the palmitolieic acid (C16: 1) was noted in the lipids and this fatty acid is known as omega 7 fatty acid which is considered to have the benefits also in human health for maintaining health cholesterol and blood sugar. The increased amount of the C16 indicates the continuous / intermittent supply of the nitrate exposes the cells to the nitrate deplete conditions and continuously triggers the lipid synthesis process.

These results are consistent with other reported studied that Nannochloropsis sp. accumulated a high proportion of saturated fatty acids (SFA) and monounsaturated fatty acids (MUFA) to >70% of the total lipid content. Compared to the commonly used soybean oil and rapeseed oil as feed stocks for biodiesel, the biodiesel derived from

microalgal lipid in this study contained SFA and MUFA which are more saturated and hence will provide a higher cetane number (CN), lower NOx emissions, shorter ignition delay time, and higher oxidative stability

# 5.3 Conclusions

Three fed batch-feeding regimes were compared in this study to see the effects of different nitrate dose in combinations with light on the productions of biomass and lipid. Overall, it can be safely summarized that fed batch culture treatments were efficient in enhancing the growth and lipid yield of *N. gaditana*. In this study the combination of the higher light intensity and nitrates efficiently improved the production of biomass. The data obtained from different fed batch cultivation imply the biomass production efficiency was highest in the pulse feeding. The biomass could be improved by developing pulse feeding culturing method to target the nitrate concentration in the growth media with sufficient light intensity. The continuous feeding data indicated the increased concentration of nitrate concentration has the very little influence on the biomass production.

The maximum biomass of 2.08 g/L obtained using the pulse feeding fed batch culture with the specific growth rate of 0.12 per day achieved was superior to the other cultivation. Furthermore the use of feeding regime and fed batch cultivation can be used to enhance the biomass and lipid production. Marine species of *N. gaditana* are capable of producing higher lipid when the pulse feeding was applied with sufficient light supply the lipid content increased to >50%DW and the higher light was found to have more effectiveness for lipid production. The lipid contents were doubled with increased light in all the treatments. The biomass contained the lipids that were rich with palmitic acid (C16:0) and palmitolieic acid (C16: 1).

The method developed to use the nutrient nitrate rich vegetable greenhouse effluents has the potential to integrate with road salt run off for the application to generate the biomass which would reduce the nutrient and water footprint of the biofuel production.

# Chapter 6

# 6 Investigation of harvesting efficiency of *N. gaditan* a from phycoremediated road salt media by flocculation and settling

Harvesting of microalgae is a necessary step where the microalgae are physically separated from the spent cell growth media. Investigations were conducted to evaluate various flocculation methods to separate microalgae from spent growth media and these results are described in this chapter.

The recovery and harvesting of biomass is the major hurdle in the production of bio-fuel from the microalgae. The rapid, efficient, and economic technologies for harvesting microalgae are currently a subject of research and are essential for successful applications. This study aims to investigate the harvesting of marine microalgae from the phycoremediated road salt media by using flocculation as a unit operation. Chemical based flocculation using inorganic metal FeCl<sub>3</sub>; organic polymer chitosan or pH change by alkali addition was examined and compared. The optimal coagulant dosage, sedimentation time and the effectiveness of pH on both the harvesting efficiency and concentration factor of algal suspensions were determined.

The results of the study showed that increasing the pH of the road salt-based growth media helped to induce the aggregation and sedimentation of *N.gaditana* cells. An increase of pH above 10.5 led to rapid flocculation of cells with harvesting efficiency of 89.2% at pH 11 and 91.5% at pH 11.5 with the rapid settling in less than 15 min. Raising the pH to 11 (noted as a critical point) promotes the flocculation of the cells by the alkaline flocculation method. In the chemical flocculation study, the inorganic salt FeCl<sub>3</sub> was the most efficient as a coagulant/flocculant and showed the 95% harvesting efficiency at the higher dosage of 80 mg/L. However, the settling rate was slower (60 minutes to achieve the complete (95%) harvesting efficiency). The concentrations of cells using alkaline induced flocculation were lower than the FeCl<sub>3</sub> method. A 16.2 concentration factor was obtained at pH 11.5 and a concentration factor of 23.6 was

found when the  $\text{FeCl}_3$  were used at the dosage of 60 mg/L at the end of 60 minutes of settling time

The use of a bio coagulant, i.e., chitosan polymer for the application of flocculation was not found to be efficient. A solution of chitosan in 1 % HCl at the dosage of 10-100 mg/L only provided below 5% of the flocculation efficiency. Thus, the alkaline and FeCl<sub>3</sub> methods for the flocculation are more appropriate for the pre-concentration of the cells.

# 6.1 Introduction

The use of microalgae for the biofuels is a promising alternative to replace the fossil fuels for the production of sustainable, renewable clean energy. Microalgae have the ability to fix carbon dioxide by photosynthesis and utilize nutrients from wastewater <sup>60,206</sup>. The micro algal systems for the biofuel productions are preferred due to their characteristic of high growth rate, the shorter growth time than the terrestrial plants and demands for less land usage <sup>36,100,101</sup>. However, the high energy input for the production and recovery of the biomass in the downstream processing is a key limitation for the economical commercial cultivation <sup>29,207</sup>. Harvesting of biomass is an important factor and the major bottleneck for the commercial use of algal biomass <sup>25,139</sup>. The cost of the dewatering process has been estimated around 25-65 % of the algal biomass cost itself <sup>4,139,208</sup>. The development of methods for the recovery of biomass from the growth media are challenged by the smaller size, negative surface charges of the microalgae, low cell concentration and the similarity in density of algal cells to the growth media <sup>32,209–211</sup>.

The methods applied for the harvesting of the biomass included the traditional separating techniques like filtration and centrifugation. Both of these techniques have disadvantages including high energy input, fouling of membranes (filtration) and hence are not considered economically viable at the present time (17–20). Thus, more effective, simple, environmentally friendly harvesting methods would be important for commercial biodiesel production from microalgae.

Flocculation techniques are applied in the waste water treatment as, low coat, effective and convenient process that could be potentially applied for the treatment of large volumes of microalgal cultures<sup>213</sup>. Flocculation of cells attained by the use of physical, chemical and biological (microbial) methods conducted by the two –step process contain coagulation/sedimentation and dewatering<sup>24</sup>.

Depending on the pH, microalgae cell surfaces are negatively charged and are suspended in the media by electrostatic repulsion forces. The application of flocculants aggregates the cells by destabilizing the net negative charges by charge neutralization, electrostatic patches (sweeping) and by the bridging mechanisms (netting). The aggregated cells sediment by gravity as sludge and the sludge removed or dewatered by filtration<sup>214</sup>.

Inorganic coagulants or flocculants like chloride and sulphates of ferric and aluminum and organic polymers (chitosan, starch, moringa flour etc.) are used to harvest microalgae <sup>173,215</sup>. The polyvalent metal, electro coagulant and polymers have been investigated for their efficiency to sediment microalgae to determine the optimal conditions to harvest the biomass from various culture media<sup>164,215,216</sup>. Among these, chitosan, i.e., d-glucosamine and N-acetyl-d-glucosamine deacetylated from chitin (the second- most-abundant natural polymer in the world), contain primary amino groups is one of the promising flocculants emerged to use it in the harvesting of microalgae <sup>217</sup>. However, the practical application of chitosan is limited due to its insolubility in water at neutral or higher pH<sup>218</sup>.

The high dosage of organic based coagulants like starch and chitosan are required for the flocculation adds cost to the treatment <sup>148,219</sup>. The dosage of chemical coagulants contaminate the biomass with metals limit the application of recovered biomass in the environment<sup>220</sup>. The other reported techniques include physical flocculation using ultrasound, electro-flocculation. These are cleaner processes but consume more energy and requires specialized equipment and may be difficult to scale up<sup>221</sup>. Bio-based purified bio flocculants and co-flocculation using microbes are considered more safe and environmentally friendly, but the cost is higher in comparison to the chemical flocculants<sup>162</sup>.

Another approach to destabilize the negative charges is by adjusting the pH of the media leading to cell self-flocculation was reported recently for the harvesting of microalgae<sup>146</sup>. The cell self-flocculation and the auto-flocculation by increasing the pH can be more

economical. Especially if the use of increased pH can be achieved by the depletion of  $CO_2$  by the photosynthetic activity of the cells itself then this could be a simple method that can be easily adopted to the culture systems <sup>121</sup>.

The negatively charged carboxyl and/or sulphate groups of cell surface stabilizes the microalgal suspensions <sup>142</sup>. The fact of flocculation of microalgae at a high pH is therefore surprising, since the surface charge of micro algal cells are expected to become more negative at a high pH and thus flocculation should be inhibited. But in contrast to the stabilization, the increase in pH above 10, algae are reported to auto flocculate due to electrostatic interactions between anionic algae and divalent cat ions with the formation of calcium or magnesium precipitates <sup>25,127,222–224</sup>. Some reports noted that the autoflocculation did not occur in waters poor in calcium and magnesium and addition of lime or with high concentration of phosphates induce the flocculation <sup>154,223</sup>.

Several questions have emerged which applying flocculation as a unit operation for microalgae. First, the majority of work done on microalgae flocculation has been reported mostly on freshwater species. The reports noted the inhibition of flocculation efficiency of polymers and the requirement of increased dosage of inorganic flocculants in the high ionic strength media<sup>121</sup>. Secondly, the effect of salinity and ionic strength of the growth media for the microalgae grown in a high ionic strength matine environment on the flocculation efficiency were not reported for the all the growth media and species. Each growth medium has unique electrical charge properties depending on types of ions and composition; therefore, even identical species being cultured in different culture medium might also have a specific threshold pH that triggers the flocculation. Finally, the intrinsic characteristics of the media and algal strains impacts on the harvesting efficiency with varied pH ranges are not reported for marine cultures cultivated in the high saline waste water systems.

Since the marine microalgae in the study is cultivated in road salt containing media, the ionic strength of the media is high with relatively higher concentration of metal ions and sodium, it was hypothesized that the increasing pH of the media would potentially induce the destabilization of the algal surface charges and promote the cell aggregation.

However, the high ionic could pose difficulties in harvesting the biomass with the use of chemical induced coagulation. Hence this study investigated the harvesting efficiencies (HE %) (See equation 3.6) using three separate approaches: inorganic ferric chloride (chemical), chitosan (bio coagulant) and pH to determine the effectiveness on N. *gaditana* flocculation on spent media. The degrees of pre-concentration were determined by applying the concentration factors as parameters (See Equation 3.7).

# 6.2 **Results and Discussion**

The flocculation experiments were performed in three sets using the above-mentioned three flocculants. The spent media culture suspension of *N. gaditana* cells are obtained from the growth experiments at end of the growth stage (21 days). Chitosan was prepared as in the materials and methods 1 % acetic, 1%citric and 1% hydro chloric acids were used .The flocculation efficiency (FE) referred in the study as harvesting efficiency (HE) expressed by the percentage of the sediment of algal sludge to the initial algal cell mass.

# 6.2.1 Effect of pH on sedimentation of *N.gaditana* cells with alkaline flocculation

The growth media of cultures was at pH (8+/-0.05) at the end of the growth stage. Hence the harvesting efficiency was studied using 0.1M NaOH to increase the pH of media. The previous studies reported noted that the potassium hydroxide and calcium hydroxide induced flocculation at the same pH as sodium hydroxide but at higher concentrations  $^{224}$ .

The harvesting efficiencies (%HE) were determined in four sedimentation time intervals (15, 30, 45 and 60 minutes) for the microalgal cells by increasing the pH of the media by 0.5 units from 9 to 12.5. The increase in the pH of the suspension with the addition of sodium hydroxide caused the settling of cells to the bottom of the test tubes with the formation of visibly noticeable interface between the clarified solution and the concentrated algae cells.

Figures 6.1 and 6.2 show the harvesting efficiencies of the *N.gaditana* with four settling times. As shown in the figures, the two separate harvesting efficiency (HE) conditions

were noted with the increased pH of media as a function of time. In the first condition, the slower settling of cells observed at pH below 11.



Figure 6:1 Harvesting efficiency versus the pH of media at various settling time

The HE was below <32% until the 45 minutes of settling time in the pH range of 9.5- to 11 were observed. And the second conditions at Above 11 pH showed rapid HE of 90%. The zones of the cell aggregation observed in the pH ranges were further showed by the harvesting efficiency versus the settling time at four time intervals (15 minutes increment) to investigate the effect of pH on the settling of the cells as a function of time. The harvesting efficiency increased as the settling time progressed from 15 minutes 60 minutes and with the increased of the pH of algal suspension media from 9 to 10.5 ranges. The settling time of 45 minutes were required to flocculate <40% of the cells and the 60 minutes flocculated 70 to 86 % of the cells at the pH 9.5 10.5 range.

In the pH 11 and above settling of the cells were faster and maximum cells were sedimented to the bottom of the test tubes with the HE of above 85% within the 15 minutes were obtained when the cells were exposed to higher pH by the increasing media pH. The two clear zones of settling rates of cells with pH were obtained. The Figure 6.2 was depicted to see the differences by plotting the percentage HE as a function of time. Hence the pH 11 was considered as threshold pH in the media, which differentiated the

settling speed of the algal cells in the experiments. The Figure 6.3 and 6.4 were plotted further to differentiate the two zones of pH to understand the sedimentation process during the increased pH of the media.



Figure 6:2 Harvesting efficiency as a function of settling time at various pH of media

As seen in Figure 6.2, the lower settling rate was noted below the threshold pH (pH=11) a. At settling times of 60 minutes in region of pH of 9.5, 10, and 10.5 provided the harvesting efficiency was 19.6%, 74.7%, 74.2% and 86% for the cells respectively. The further increased the pH by 0.5 units from 10.5 to 11 caused rapid settling of the cells resulting with 1.8x increased harvesting efficiency from 32.3 % to 89.2% noticed with in

the 15 minutes of settling time. This zone is referred as the fastest settling pH range (second zone pH=>11). Above pH 11 no increase in HE is noted at the end of 60 minutes. The harvesting efficiency above 90% was achieved by increasing the pH from the threshold (pH=11) to pH 11.5 and 12.0 at the 15 minutes of settling time. Further increase in the pH to 12.5 showed slight decrease to 89% in the harvesting efficiency. The trends of the HE were further described by the linear equation of the trend line obtained.



Figure 6:3 Harvesting efficiency of *N.gaditana* as a function of time in the pH ranges (9 to 10.5)



Figure 6:4 Harvesting efficiency of *N. gaditana* as a function of time in the pH ranges (11 to 12.5)

An empirical linear relationship in the following equation were observed between the harvesting efficiency versus the settling time (0-60 minutes) (see the below equations 6.4 to 6.7)

HE % = 
$$0.34 \text{ x} + 2.4 (\text{R}^2 = 0.92) \text{at pH 9}$$
 (6.4)

HE % = 
$$1.17x + 2.4$$
 (R<sup>2</sup> = 0.92)at pH 9.5 (6.5)

HE % = 
$$1.23 \text{ x} + 2.4 (\text{R}^2 = 0.81) \text{at pH } 10$$
 (6.6)

HE % = 
$$1.17 \text{ x} + 2.6 (\text{R}^2 = 0.81) \text{at pH } 10.5$$
 (6.7)

Where x is the sedimentation time and the HE (%) is the harvesting efficiency as per the equation 3.6.

The increasing the pH of the media from 9 to 10.5 with the NaOH was provided the linear relationship between the harvesting efficiency (HE %) and settling time. The flocculation and settling of the cells were increased with the time (0-60 minutes) .The settling of the cells was steadily increased when the cells were exposed to the pH longer time.

But these trends were not observed when the pH of the media increased from 11 and above. The increased pH in the range of 11 to 12.5 provided the increased harvesting efficiency (HE %) and the faster settling of the cells. The harvesting efficiency of 89-91.5% was obtained at the 15 minutes of the settling time at pH 11, 11.5,12 and 12.5 respectively. The harvesting efficiency (%HE) were 90, 93.1, 91.3 and 90.3% were obtained at pH11, 11.5,12 and 12.5 respectively at exposure of the cells for 60 minutes to these alkaline pH.

Since the majority of the cells experienced the immediate flocculation, the increasing the sedimentation time did not further increased the harvesting efficiency. The steady values between the 90 -93% were obtained at the end of the 60 minutes sedimentation time. The harvesting efficiency (%HE) as function of time were not maintained the linear relationship, since the maximum settling of the cells were completed within the first

15minutes of the sedimentation time. The settling of the cells in the 15 minutes were 5.95x, 6.1x, 6.1x, 5.95x at pH 11,11.5,12 and 12.5 were obtained when the trend line of the harvesting efficiency (%HE) versus the settling time further implying the settling rate is

In this study, a pH-induced flocculation method was applied to harvest marine *N*. *gaditan*a and the degree of the cell aggregation was determined by the flocculation efficiency. The flocculation efficiency was presented as the harvesting efficiency versus the pH values.

The steady increased settling of the cells increasing the harvesting efficiency (%HE) were observed in the 9 to 10.5 pH range with the 70 to 86 % of HE at 60minutes of the sedimentation time. When the pH increased above 10.5, the cell flocculation experienced immediate effect with the change in the pH of the media. The rapid settling of 90% cells noted at the first interval of settling time (15 minutes). But the progressing the settling time did not able to flocculate 100% of the cells since the increased settling time reached an asymptote (Figure 6.4). The pH 11 and above provided the maximum settling of cells with highest HE was noted at the pH of 11.5. Hence the pH of 11 was considered the critical threshold pH that changes the settling behavior of the cells. The low flocculation of the cells at below the pH 10.5 might be accounted to the inherent buffering capacity of salt media.

The effect of flocculation efficiency with change in pH were previously noted in several studies and it has been suggested that calcium, magnesium, phosphate may have a role. Since the cultivation was conducted using the road salt media which has the initial concentration of Mg of 4.6 mg/L and Calcium of 102 mg/L (see Table 4.1 materials previously measured by the ICP-MS). The role of calcium and phosphate precipitation at different pH ranges were noted in the waste water system were previously provided <sup>223</sup> described below for the explanation to the threshold pH and maximum settling. The formation of calcium carbonate precipitates out at pH 9.1- to 9.5 (Equation 6.8 and 6.9) and entraps the suspended and colloidal particles by sweep coagulation as per the below equation.

$$Ca(OH)_2 + H_2CO_3 \rightarrow CaCO_3 + 2H_2O$$
(6.8)

$$Ca(OH)_2 + Ca(HCO_3)_2 \rightarrow 2CaCO_3 + 2H_2O$$
(6.9)

The precipitation of Phosphate takes place at the pH of 10.5 to 11 as per the below equation (Equation 6.10 and 6.12). The microalgae consume the phosphate for the growth and hence the effect of phosphate precipitation will not be applied for the microalgae.

$$3 \operatorname{Ca}(OH)_2 + 2PO_4 \rightarrow \operatorname{Ca}_3(PO_4)_2 + 6 OH^-$$
 (6.10)

$$4 \operatorname{Ca}(OH)_2 + 3PO_4 + H_2O \rightarrow \operatorname{Ca}_4H(PO_4)_3 + 9 OH^-$$
(6.11)

Since the sodium hydroxide is used as base to increase the alkalinity of the media, the increased pH precipitates the calcium and magnesium as per the below equation.

$$Mg(HCO_3)_2 + 2NaOH \rightarrow Mg(OH)_2 + 2Na(HCO)_3$$
(6.12)

$$Ca(HCO_3)_2 + 2NaOH \rightarrow Ca(OH)_2 + Na(HCO_3)$$
(6.13)

Since the media contains magnesium, the precipitation magnesium begins at approximately pH 9.5, becomes significant above pH 10.5, and is essentially complete at pH 11.0–11.5 were noted in the previous studies <sup>223</sup> will be applicable to the media. The metal hydroxide formed above during the increased pH acts as "weighting agent "for the suspending particles induce the settling and described in the waste water treatment process<sup>222</sup>. The increased flocculation at the pH 11 and above is in the agreement of previous reports of Chlorella cultivated in fresh water media provided a noticeably increased flocculation exceeding 95% at pH 11.5and 12, ranges <sup>224</sup> were further explained. The charge neutralization at increased pH was explained by the formation of magnesium hydroxide, which will be most likely mechanism induced the flocculation of the *N. gaditana* cells. The explanation provided previously as the trivalent aluminum and ferric ions if present in the media replaces bivalent magnesium in the crystal structure of magnesium hydroxide provides net positive charges to neutralize the negative surface charges. The effect of cell density and the effectiveness harvesting using the pH were also explained for the marine species defined the amount of alkaline required for the bulk harvesting using different bases <sup>225</sup>.

# 6.2.2 Harvesting efficiency of *N. gaditana* cells with ferric chloride

The harvesting efficiency of *N. gaditana* was determined using the inorganic flocculants ferric chloride at the concentration ranging from 10-to100 mg/L. The harvesting efficiencies were plotted for the four settling times in the figure 6.5. As shown in the figure the increasing the dosage of ferric chloride and settling time increased the harvesting efficiency.

The dosage of 50 mg/L and above could start the immediate aggregation the cells were noticed at 30 minutes. The dosage of 50 mg/L resulted 89 % HE at the end of 30 minutes of settling time. The dosage of 100 mg/l resulted with cell aggregation provided the harvesting efficiency of 45% at the end of 15 minutes accounting approximately for the 50% of the total harvesting efficiency obtained by the study. The dosage of 60 mg/L and above of ferric chloride provided the >90 % harvesting efficiency at the 30 minutes of settling time with maximum of 94.7% harvesting efficiency obtained at the 100 mg/L dosage.

The increased settling time to 45 minutes increased the harvesting efficiency over 90% for the samples added with dosage of 20 mg/L and above ferric chloride. In general at the end of the 60 minutes the harvesting efficiency of 95% obtained in all the samples that are added with 70 mg/L and above ferric chloride. Further increasing the settling time had very minimum effect on the flocculation. The dosage of the ferric chloride varies between the culture conditions were reported previously <sup>215</sup>.



Figure 6:5 Harvesting efficiencies at various settling time versus ferric chloride dosage

To understand the combined effect of change in the pH of the culture system pH=9,10 and 6 on the flocculation of efficiency of ferric chloride were studied at 10,20,30, 40 and 50 mg/L dosages rates with the settling time of 30 minutes. The acidic pH (pH=6) reduced the flocculation efficiency of the *N.gaditana* cells as noticed in the figure 6.6, since the decreased pH to 6 reduced harvesting efficiency and the higher dosage was required higher to flocculate the cells. The 50mg/L dosage provided the HE of 80 % at 30 minutes of settling time. But the increased pH to 9 and 10 showed the increased harvesting efficiency of 90% were obtained at pH 9 and 10 with the dosage of 50 mg/L of ferric chloride flocculants.



Figure 6:6 Harvesting efficiencies with ferric chloride and change in the media pH

The dissociation and availability of ferric ions to neutralize the surface charges of the cells vary with the media composition. Generally, ferric ions can react with hydroxyl to form ferric hydroxides, even at low pH. The change in the pH changes to the alkaline conditions induces the sweeping flocculation by the metal hydroxide as noticed in the alkaline induced treatments. The chemical flocculation by the ferric chloride takes place by the charge neutralization. The combined effect of sweeping and charge neutralization showed a positive impact on the cell flocculation. The amount of flocculants required is highly dependent on the number of charges that need to be neutralized, which are in turn a function of the charge density of the cell surface as well as the surface to volume ratio of the cells, parameters that differ strongly between species were noted <sup>226</sup>. But the slight increased pH assisted in reduction of dosage ofcoagulant/ flocculant ferric chloride in the study.

There were several mechanisms causes flocculation due the reduction in the energy barrier between the cells by increased ionic strength of the media, reduction of negative charges of cell surface or charge neutralization. The multivalent metal cations, such as Mg<sup>2+</sup>, Ca<sup>2+</sup> and Fe<sup>3+</sup> play important roles in destabilizing the negative surface charges in microalgae to promote the flocculation process at higher pH. The hydroxyl ions at the higher pH combine with metal form metal hydroxides ions. The positive charges of the metal hydroxides neutralize the negative surface charges of microalgae. The increased flocculation efficiency hence will be related to the charge neutralization. The increase the pH by adding NaOH ferric hydroxides are formed neutralizes the surface charges of the microalgal cells enhances the rate of the cell aggregation and settling. Formation of ferric hydroxides increases with the increase in pH or dosage of NaOH results in higher availability of ionic form of metal hydroxide to enhance the flocculation process.

The culture parameters define the flocculation properties of the ferric chloride. The flocculation properties of ferric chlorides explained by the formation of ferric hydroxide and interaction of ferric with released polysaccharides from the cells<sup>155,227</sup>. The inhibitory influence of released polysaccharides (RPS) on the recovery of halophillic microalgae the *cyanobacterium A. halophytica GR02* with ferric chloride were reported that the attachment of ferric ions to cell surfaces were reduced with the presence of RPS <sup>227</sup>. But in the study the inhibition of the flocculation were not observed.

### 6.2.3 Harvesting efficiency of *N. gaditana* cells with chitosan

Chitosan is soluble in acidic pH and the solution (10 mg/ml) of chitosan was prepared in acetic, citric and HCl were added at 10-100mg/L concentrations to the algal suspension to investigate the effectiveness of flocculation of the marine microalgae. The chitosan is soluble in acidic pH and its application to the marine microalgae growth media provided the lower efficiencies. Since chitosan is soluble in acidic solution, the effects of three acids solvents were used for the preparation of chitosan flocculants solution. The influence of chitosan on flocculation efficiency was failed to provide the sedimentation of the cells as noticed in the previous two treatments. The clear separations of cells from the media were not observed and the tested conditions provided the HE were less than 10% efficiency.(fig 6.7). Among the three-tested solution of the chitosan solution, the chitosan in the citric acid solution (HE=4.9%). The optical density of the top layer fluctuated in the algal

suspension when the chitosan solution prepared with acetic acids were added and the least HE (>3.5%) based on the optical density were observed at 10-100 mg/L dosage levels of flocculants chitosan.





The optimal working pH for chitosan flocculation is pH 6 or lower and is more efficient cationic flocculants in a weak acidic environment. The chitosan contains a poly-glucosamine backbone and has an isoelectric point around 6.5. More parts of the polymeric chains are positively charged at pH 6 and below than at pH 7 or higher. The decrease of pH below 7 reported to increase the chitosan activity and harvesting efficiency due to the decreasing in the viscosity and the mean surface charge of algal cells <sup>228</sup>.
Among all the tested flocculants, chitosan showed the lowest performance for the flocculation of biomass even at very high dosages (100 mg/L). This was possibly due to the pH of the road salt media were around 8. It is clear that chitosan flocculation conditions depend on the pH, which influences the availability of the active sites of the chitosan for the charge neutralization. The maximum harvesting observed in the HCl based chitosan solution than the citric acid. However the acetic acid showed the lowest flocculation efficiency and the increased inclusion of chitosan did not influence the performance of chitosan on cell flocculation as no clear separation of the cell/sludge formation were noted in the testing tubes.

The mechanism of operating in chitosan–algal cell flocculation described by the reports suggests the combination of partial charge neutralization, static patch effects, and bridging (netting). The structure of the chitosan consist positively charged deacetylated units that are spread along the whole backbone of chitosan chains. This extended arrangement of the polymer maintained by the repulsive forces. The decrease or the no effect in flocculation efficiency of chitosan in the road salt media might be due the high ion concentration of the environment is probably the large amounts of ions in a high ionic medium shield the repulsion of positive charges on chitosan chains, thus causing a reduction in their ability to cause flocculation.

The DLVO theory states that suspensions are stabilized by surface charges of the particles. The balance between the electrostatic repulsion and the Van der Waals attraction can be shifted towards attraction, causing coagulation and flocculation through several mechanism. <sup>229,230</sup>. The alkaline flocculation mechanism is by the sweeping mechanisms while the chemical coagulants is by charge neutralization were previously reported were effectively flocculated the cells.<sup>127,137,215,231,232</sup>. Flocculation induced by high pH is a potentially useful method to pre-concentrate microalgal biomass. However, the method depends on sufficiently increased high pH (>10.5) using sodium hydroxide can be attained over >90 % cell aggregation within 15 minutes.

#### 6.2.4 Concentration factors of algal suspension

Flocculation and concentration factor are the two important factors determine the harvesting efficiencies. The concentration factor (CF) is important factor to evaluate the degree of harvest and estimates the pre concentration of the biomass. The volumes of the clarified media obtained in the flocculation experiments were used to determine the concentration factors. Clarification factor of the pH induced, chemical and chitosan flocculation were plotted in figure 6.8, 6.8 and 6.10. As shown in the figures the concentration factor increased with settling time provides the clear indication of increased sedimentation of the algal cells. The maximum concentration factor of 16.2+/-0.3 obtained with pH 11.5 at the 60 minutes sedimentation time in the pH induced flocculation treatments, which corresponds to the maximum harvesting efficiency.

The flocculants ferric chloride provided the maximum concentration factor of  $23.56 \pm 0.3$  at end of the settling time 60 minutes with the dosage of 60 mg/L of flocculants. Among the tested flocculation methods application of chitosan was failed to separate the cells and hence the lowest concentration factors were noted.

However this highest value of the ferric chloride based flocculation does not corresponds to the highest harvesting efficiency. The highest harvesting efficiency of 95.7 % obtained at the dosage of 100 mg/l of the ferric chloride provided the 12.51 concentration factor. The concentration factors were further compared to the ratios of biomass obtained by the dry weight (Table 6.1). The concentration factor of the biomass obtained does not by the volume differences were noted slightly higher than the dry weight, since the dry weight removes the moisture and did not directly translated to the concentration factor of the volume reduction



Figure 6:8 Concentration factor by pH induced flocculation at 4 sedimentation times

•



Figure 6:9 Concentration factor with ferric chloride at 3 sedimentation times

The concentration to 20x were reported previously in *Chlorella* species with the dosage of 5mM of NaOH High flocculation efficiency of 90%  $^{155}$  and on marine diatom *Phaeodactylum tricornutum* with the alkaline  $^{233}$  were noted in the previous studies . However the present on the *N. gaditana* by the alkaline pH were achieved the maximum of 15.36 concentration factor. But the application of ferric chloride resulted above 20x concentration factor in the study. The pre concentration of this level by the flocculation techniques provided opportunity to apply the flocculation methods for harvesting microalgae from the growth media.





#### 6.2.5 Comparison of concentration factors

While the harvesting efficiency was defined as the ratio of the mass of cells recovered to the total mass of cells and the concentration factor was the ratio of the final product concentration to the initial concentration <sup>234</sup>. The increased concentration factor means to the decreased amount of the sludge, which provided increased quantity of the clarified media in the flocculation methods. The higher the concentration factor lesser the growth media in the sludge, which reduces the further, drying process in the downstream processing. Concentration factor of the three methods were further analyzed by the cell dry weight Table 6.1. Flocculants are attached to the cells, when the flocculation is achieved by the use of flocculants, the amount of the ions attached would contribute to the weight of the biomass.

Treatment method	Initial biomass concentr ation (g/L)	Biomass oncentration by e dry weight ethod (g/L)	CF by the mass difference by dry weight method	CF by the volume difference of flocculati on
pH 11.5 at 60 minutes sedimentation	1.25+/- 0.05	19.2+/-0.23	15.36	16.2+/- 0.03
Ferric chloride @ 60mg/L dosage at 60minutes of sedimentation	1.27+/- 0.05	27.6+/-0.043	21.73	23.56+/- 0.29
Ferric chloride @ 100 mg/L dosage at 60minutes of sedimentation	1.27+/- 0.05	15.2+/-0.217	11.97	12.51+/- 0.26

Table 6-1 Comparison of Concentration factor dry weight method and flocculation volume reduction methods

## 6.3 Conclusions

In conclusion, three chemical flocculants namely alkaline and ferric chloride and chitosan were applied to harvest the cells of *N. gaditana* from road salt run off media for the biomass recovery. The alkaline based flocculation provided the two different settling patterns of the cell, The harvesting efficiencies were in the liner relationship with the setline time at the pH range of 9-10.5 and the rapid settling of the cells were observed the pH range of 11-12.5. Over 89 % of harvesting efficiencies were obtained at the pH 11 and above.

The chemically induced flocculation using ferric chloride is the most effective method for the harvesting the marine microalgae *N. gaditaa* and was dosage dependent. The highest

harvesting efficiency of 95% at the settling time of 60 minutes were obtained when the dosage of 80 to 100 mg/L were applied to the spent media. Averages of 90 % harvesting efficiencies were obtained at the dosage of 50mg/L ferric chloride. The increased pH reduces the dosage of ferric chloride to obtain the >90 % of harvesting efficiency.

The last method of flocculation with the organic flocculant chitosan provided very poor performances of cell harvesting and was considered not effective to harvest the marine microalgae in the road salt media. Chitosan is soluble in acid and the change in the acid solution (citric, acetic, HCl) did not improve the harvesting efficiencies. The harvesting efficiencies obtained were below 6% at the tested dosage range of 10-100 mg/L of chitosan.

Therefore, the alkaline-based flocculation method has important advantages, such as high harvesting efficiency, environment friendly, and low cost, which could be more applicable for harvesting microalgae for biofuel production than other methods.

The concentration factor (CF) is important factor to evaluate the degree of harvest and estimate for the pre concentration of the biomass. The alkaline flocculation at pH 11 and above provided the CF of above 10 and the maximum of 16.2 concentration factor. The ferric chloride was the most efficient method, which provided the maximum concentration of the cells (CF=23.6) in the sludge.

# Chapter 7

### 7 Conclusions and recommendations

The research study investigated phycoremediation using marine microalgae cultivated on simulated media representing road-salt and vegetable Greenhouse industry effluents and the potential to use in biofuel application. Here the overall conclusions and recommendations are provided.

#### 7.1 Conclusions

In this study, a review of the wastewater systems that are limited for the traditional treatment in the local area identified the road salt run off and green house industry effluents as candidates for the phycoremediation processes. The use of low cost/zero cost effluents was chosen to provide a competitive advantage to biofuel production in microalgae. *Nannochloropsis* was the microalgal species of choice as it is halophilic and accumulates high content of lipids. First batch growth studies were carried out to evaluate a range of parameters including salt concentrations and nitrate loading on cell growth. Simulated nutrient media, which represented road-salt effluent mixed with vegetable Greenhouse effluents, were investigated in batch and fed-batch bioreactors under different light regimes. Overall it is concluded that the study showed that phycoremediation is a useful unit operation, which can add value to the bioprocess by generating lipids for applications as biofuels.

The investigation of fed batch cultivation methods for the continuous supply of the nitrate nutrients to the growth media is an effective method for simultaneously increasing the biomass yield and lipid content. This method was efficient was able to increase the biomass yield to 2.08 g/L and lipid content to 58% DW with the application of light production. The results of these studies are summarized with the nitrogen to cell conversation rate and volumetric lipid yield (mg/L) and the lipid yield on nitrate, which can be used for the cost efficiencies

Finally the approaches of flocculation methods to harvest the biomass were investigated using the alkaline, chemical flocculants ferric chloride and chitosan and the results showed the increased the pH of the media induce the flocculation of the cells in the road salt media .The effective threshold pH 11 and above provided the 90% of the cell aggregation and sedimentation . The ferric chloride is the most efficient flocculants resulted with h 95 % of harvesting efficiency at the dosage of 70mg/L with the settling time of 60 minutes. The chitosan solution were solublised using three acids were failed to coagulate the cells in road salt media with in the settling time of 60 minutes

The biomasses obtained were subjected for the fatty acid analysis. The fatty acid profile of the lipid defines the suitability for use as biofuel to meet the standards of ASTM. The fatty acid profile of the lipids extracted from the biomass dominated the C16 and C18 indicated the monounsaturated and saturated lipid were contributed to76 % of the FAME

### 7.2 **Recommendations**

From the experience obtained after the completion of this thesis the following suggestions are done for future work:

- Chapter 4: The vegetable greenhouse were used for the study, but it would be interesting to conduct growth study to on other types of wastewaters and industrial effluents to analyze the biomass productivity and lipid yield for the field conditions and to examine productivities in other marine and halophillic microalgae species..
- Chapter 5. The details study of the feeding regimes on three-fed batch cultivation for the microalgae cultivation provided the direction of its efficiency on increasing the biomass and lipid yield. Further studies could focus on scale up on a larger scale under field conditions.
- In chapter 6 briefly flocculation/harvesting approaches were compared for algal removal from spent media. This approach could be further compared with other

traditional and non-traditional methods. Methods for the recovery of the cells from the growth media. It will be useful to discuss and explore other types of bio and chemical coagulants and also determine economic and life cycle analyses of the processes.

• Care should however be taken that the high pH does not destroy the microalgal cells, as this may result in loss of useful bio products from the biomass. The chemical flocculation and the pH-induced method are sustainable. Further research regarding the impact of pH induced flocculation on cell viability and recycling of growth media and effect on the biomass and upstream processing like lipid extraction and fatty acid ethyl ester are needed for the for the consideration of application of biomass for the biofuel production.

### References

- Shurtz BK, Wood B, Quinn JC. Nutrient resource requirements for large-scale microalgae biofuel production: Multi-pathway evaluation. *Sustain Energy Technol Assessments*. 2017;19:51-58. doi:10.1016/j.seta.2016.11.003.
- Costa JAV, de Morais MG. The role of biochemical engineering in the production of biofuels from microalgae. *Bioresour Technol*. 2011;102(1):2-9. doi:10.1016/j.biortech.2010.06.014.
- Patel A, Gami B, Patel P, Patel B. Microalgae: Antiquity to era of integrated technology. *Renew Sustain Energy Rev.* 2016;71(December 2016):535-547. doi:10.1016/j.rser.2016.12.081.
- Dickinson S, Mientus M, Frey D, et al. A review of biodiesel production from microalgae. *Clean Technol Environ Policy*. 2016:1-32. doi:10.1007/s10098-016-1309-6.
- Renuka N, Sood A, Prasanna R, Ahluwalia AS. Phycoremediation of wastewaters: a synergistic approach using microalgae for bioremediation and biomass generation. *Int J Environ Sci Technol*. 2015;12(4):1443-1460. doi:10.1007/s13762-014-0700-2.
- Gong Y, Jiang M. Biodiesel production with microalgae as feedstock: From strains to biodiesel. *Biotechnol Lett.* 2011;33(7). doi:10.1007/s10529-011-0574-z.
- Bhaskar S, Guldhe A, Rawat I, Bux F. Towards a sustainable approach for development of biodiesel from plant and microalgae. *Renew Sustain Energy Rev.* 2014;29:216-245. doi:10.1016/j.rser.2013.08.067.
- Suali E, Sarbatly R. Conversion of microalgae to biofuel. *Renew Sustain Energy Rev.* 2012;16(6). doi:10.1016/j.rser.2012.03.047.

- Abdelaziz AEM, Leite GB, Hallenbeck PC. Addressing the challenges for sustainable production of algal biofuels: I. Algal strains and nutrient supply. *Environ Technol*. 2013;34(13-14):1783-1805. doi:10.1080/09593330.2013.827748.
- Xie Y, Ho SH, Chen CNN, et al. Phototrophic cultivation of a thermo-tolerant Desmodesmus sp. for lutein production: Effects of nitrate concentration, light intensity and fed-batch operation. *Bioresour Technol.* 2013;144:435-444. doi:10.1016/j.biortech.2013.06.064.
- Singh A, Nigam PS, Murphy JD. Renewable fuels from algae: An answer to debatable land based fuels. *Bioresour Technol*. 2011;102(1):10-16. doi:10.1016/j.biortech.2010.06.032.
- Msanne J, Xu D, Konda AR, et al. Metabolic and gene expression changes triggered by nitrogen deprivation in the photoautotrophically grown microalgae Chlamydomonas reinhardtii and Coccomyxa sp. C-169. *Phytochemistry*. 2012;75:50-59. doi:10.1016/j.phytochem.2011.12.007.
- Bachet Q, Shilton A, Guieysse B. Modeling the effects of light and temperature on algae growth: State of the art and critical assessment for productivity prediction during outdoor cultivation. *Biotechnol Adv.* 2013;31(8). doi:10.1016/j.biotechadv.2013.08.014.
- Slade R, Bauen A. Micro-algae cultivation for biofuels: Cost, energy balance, environmental impacts and future prospects. *Biomass and Bioenergy*. 2013;53. doi:10.1016/j.biombioe.2012.12.019.
- Farooq W, Suh WI, Park MS, Yang JW. Water use and its recycling in microalgae cultivation for biofuel application. *Bioresour Technol*. 2015;184:73-81. doi:10.1016/j.biortech.2014.10.140.
- Chiu SY, Kao CY, Chen TY, Chang Y Bin, Kuo CM, Lin CS. Cultivation of microalgal Chlorella for biomass and lipid production using wastewater as nutrient

resource. *Bioresour Technol*. 2015;184. doi:10.1016/j.biortech.2014.11.080.

- Craggs RJ, Heubeck S, Lundquist TJ, Benemann JR. Algal biofuels from wastewater treatment high rate algal ponds. *Water Sci Technol*. 2011;63(4):660-665. doi:10.2166/wst.2011.100.
- Jiang L, Luo S, Fan X, Yang Z, Guo R. Biomass and lipid production of marine microalgae using municipal wastewater and high concentration of CO2. *Appl Energy*. 2011;88(10):3336-3341. doi:10.1016/j.apenergy.2011.03.043.
- Manirafasha E, Ndikubwimana T, Zeng X, Lu Y, Jing K. Phycobiliprotein: Potential microalgae derived pharmaceutical and biological reagent. *Biochem Eng* J. 2016;109:282-296. doi:10.1016/j.bej.2016.01.025.
- 20. Rangel-Yagui CDO, Danesi EDG, De Carvalho JCM, Sato S. Chlorophyll production from Spirulina platensis: Cultivation with urea addition by fed-batch process. *Bioresour Technol*. 2004;92:133-141. doi:10.1016/j.biortech.2003.09.002.
- 21. Xu L, Weathers PJ, Xiong XR, Liu CZ. Microalgal bioreactors: Challenges and opportunities. *Eng Life Sci.* 2009;9(3):178-189. doi:10.1002/elsc.200800111.
- 22. Borowitzka MA. High-value products from microalgae-their development and commercialisation. *J Appl Phycol*. 2013;25(3). doi:10.1007/s10811-013-9983-9.
- Odjadjare EC, Mutanda T, Olaniran AO. Potential biotechnological application of microalgae: a critical review. *Crit Rev Biotechnol*. 2015;8551(November):1-16. doi:10.3109/07388551.2015.1108956.
- Barros AI, Gonçalves AL, Simões M, Pires JCM. Harvesting techniques applied to microalgae: A review. *Renew Sustain Energy Rev.* 2015;41. doi:10.1016/j.rser.2014.09.037.
- 25. Heaven S. A review of the harvesting of micro-algae for biofuel production. *Rev Environ Sci Biotechnol.* 2013;12(2). doi:10.1007/s11157-012-9301-z.

- Suganya T, Varman M, Masjuki HH, Renganathan S. Macroalgae and microalgae as a potential source for commercial applications along with biofuels production : A biore fi nery approach. *Renew Sustain Energy Rev.* 2016;55:909-941. doi:10.1016/j.rser.2015.11.026.
- Juang YJ, Chang JS. Applications of microfluidics in microalgae biotechnology: A review. *Biotechnol J.* 2016;11(3):327-335. doi:10.1002/biot.201500278.
- Brennan L, Owende P. Biofuels from microalgae-A review of technologies for production, processing, and extractions of biofuels and co-products. *Renew Sustain Energy Rev.* 2010;14:557-577. doi:10.1016/j.rser.2009.10.009.
- Mata TM, Martins A a., Caetano NS. Microalgae for biodiesel production and other applications: A review. *Renew Sustain Energy Rev.* 2010;14:217-232. doi:10.1016/j.rser.2009.07.020.
- Banerjee C, Singh PK, Shukla P. Microalgal bioengineering for sustainable energy development: Recent transgenesis and metabolic engineering strategies. *Biotechnol* J. 2016;11(3):303-314. doi:10.1002/biot.201500284.
- Williams PJLB, Laurens LML. Microalgae as biodiesel & biomass feedstocks: Review & analysis of the biochemistry, energetics & economics. *Energy Environ Sci.* 2010;3(5):554. doi:10.1039/b924978h.
- 32. Tredici MR. Photobiology of microalgae mass cultures: understanding the tools for the next green revolution. *Biofuels*. 2010;1:143-162. doi:10.4155/bfs.09.10.
- Lam MK, Lee KT, Ho S-H, et al. Microalgae biofuels: A critical review of issues, problems and the way forward. *Biotechnol Adv.* 2012;32(8):673-690. doi:10.1016/j.biotechadv.2014.09.002.
- Lane K, Derbyshire E, Li W, Brennan C. Bioavailability and Potential Uses of Vegetarian Sources of Omega-3 Fatty Acids: A Review of the Literature. *Crit Rev Food Sci Nutr.* 2014;54(5):572-579. doi:10.1080/10408398.2011.596292.

- Khozin-Goldberg I, Iskandarov U, Cohen Z. LC-PUFA from photosynthetic microalgae: Occurrence, biosynthesis, and prospects in biotechnology. *Appl Microbiol Biotechnol.* 2011;91(4):905-915. doi:10.1007/s00253-011-3441-x.
- Chisti Y. Biodiesel from microalgae. *Biotechnol Adv.* 2007;25(3):294-306. doi:10.1016/j.biotechadv.2007.02.001.
- 37. Rawat I, Ranjith Kumar R, Mutanda T, Bux F. Dual role of microalgae: Phycoremediation of domestic wastewater and biomass production for sustainable biofuels production. *Appl Energy*. 2011;88(10):3411-3424. doi:10.1016/j.apenergy.2010.11.025.
- Abdel-Raouf N, Al-Homaidan AA, Ibraheem IBM. Microalgae and wastewater treatment. *Saudi J Biol Sci.* 2012;19(3):257-275. doi:10.1016/j.sjbs.2012.04.005.
- Farrelly DJ, Everard CD, Fagan CC, Mcdonnell KP. Carbon sequestration and the role of biological carbon mitigation \_ Areview.pdf. 2013;21:712-727.
- Zaimes GG, Khanna V. The role of allocation and coproducts in environmental evaluation of microalgal biofuels: How important? *Sustain Energy Technol Assessments*. 2014;7:247-256. doi:10.1016/j.seta.2014.01.011.
- 41. Mutanda T, Ramesh D, Karthikeyan S, Kumari S, Anandraj a., Bux F.
  Bioprospecting for hyper-lipid producing microalgal strains for sustainable biofuel production. *Bioresour Technol*. 2011;102(1):57-70. doi:10.1016/j.biortech.2010.06.077.
- Kharati-Koupaei M, Moradshasi A, Benavente-Vald??s JR, et al. Nitrate repletion strategy for enhancing lipid production from marine microalga Tetraselmis sp. *Bioresour Technol*. 2016;205(3):274-279. doi:10.1016/j.biortech.2016.01.045.
- Benavente-Vald??s JR, Aguilar C, Contreras-Esquivel JC, M??ndez-Zavala A, Montaez J. Strategies to enhance the production of photosynthetic pigments and lipids in chlorophycae species. *Biotechnol Reports*. 2016;10:117-125. doi:10.1016/j.btre.2016.04.001.

- Griffiths M. J, Van Hille RP, Harrison STL. The effect of nitrogen limitation on lipid productivity and cell composition in Chlorella vulgaris. *Appl Microbiol Biotechnol.* 2014;98(5). doi:10.1007/s00253-013-5442-4.
- Ho S, Ye X, Hasunuma T, Chang J, Kondo A. Perspectives on engineering strategies for improving biofuel production from microalgae — A critical review. *Biotechnol Adv.* 2014;32(8):1448-1459. doi:10.1016/j.biotechadv.2014.09.002.
- Klok AJ, Martens DE, Wijffels RH, Lamers PP. Simultaneous growth and neutral lipid accumulation in microalgae. *Bioresour Technol*. 2013;134:233-243. doi:10.1016/j.biortech.2013.02.006.
- Abomohra AEF, Jin W, Tu R, Han SF, Eid M, Eladel H. Microalgal biomass production as a sustainable feedstock for biodiesel: Current status and perspectives. *Renew Sustain Energy Rev.* 2016;64:596-606. doi:10.1016/j.rser.2016.06.056.
- 48. Tamayo-Ordóñez Y.J, Ayil-Gutiérrez BA, Sánchez-Teyer FL, et al. Advances in culture and genetic modification approaches to lipid biosynthesis for biofuel production and *in silico* analysis of enzymatic dominions in proteins related to lipid biosynthesis in algae. *Phycol Res.* 2017;65(1):14-28. doi:10.1111/pre.12162.
- Andrade JE, Pérez A, Sebastian PJ, Eapen D. A review of bio-diesel production processes. *Biomass and Bioenergy*. 2011;35(3):1008-1020. doi:10.1016/j.biombioe.2010.12.037.
- Yousuf A, Khan MR, Islam MA, Wahid ZA, Pirozzi D. Technical difficulties and solutions of direct transesterification process of microbial oil for biodiesel synthesis. *Biotechnol Lett.* 2016:1-11. doi:10.1007/s10529-016-2217-x.
- Knothe G. Analyzing Biodiesel: Standarts and other Methods. *Jaocs*. 2006;83(10). doi:10.1007/s11746-006-5033-y.
- 52. Taleb A, Pruvost J, Legrand J, et al. Development and validation of a screening procedure of microalgae for biodiesel production: Application to the genus of

marine microalgae Nannochloropsis. *Bioresour Technol*. 2015;177. doi:10.1016/j.biortech.2014.11.068.

- Pruvost J, Van Vooren G, Le Gouic B, Couzinet-Mossion a., Legrand J. Systematic investigation of biomass and lipid productivity by microalgae in photobioreactors for biodiesel application. *Bioresour Technol*. 2011;102(1):150-158. doi:10.1016/j.biortech.2010.06.153.
- Ríos SD, Torres CM, Torras C, Salvadó J, Mateo-Sanz JM, Jiménez L. Microalgae-based biodiesel: Economic analysis of downstream process realistic scenarios. *Bioresour Technol*. 2013;136:617-625. doi:10.1016/j.biortech.2013.03.046.
- Huang G, Chen F, Wei D, Zhang X, Chen G. Biodiesel production by microalgal biotechnology. *Appl Energy*. 2010;87(1):38-46. doi:10.1016/j.apenergy.2009.06.016.
- 56. Maclean C, Cembella AD, Quilliam MA. Effects of Light, Salinity and Inorganic Nitrogen on Cell Growth and Spirolide Production in the Marine Dinoflagellate Alexandrium ostenfeldii (Paulsen) Balech et Tangen 1. 2003;46(Braarud 1945):466-476.
- Ruangsomboon S. Effect of light, nutrient, cultivation time and salinity on lipid production of newly isolated strain of the green microalga, Botryococcus braunii KMITL 2. *Bioresour Technol*. 2012;109:261-265. doi:10.1016/j.biortech.2011.07.025.
- Procházková G, Brányiková I, Zachleder V, Brányik T. Effect of nutrient supply status on biomass composition of eukaryotic green microalgae. *J Appl Phycol*. 2014;26(3):1359-1377. doi:10.1007/s10811-013-0154-9.
- Packer A, Li Y, Andersen T, Hu Q, Kuang Y, Sommerfeld M. Growth and neutral lipid synthesis in green microalgae: A mathematical model. *Bioresour Technol*. 2011;102(1):111-117. doi:10.1016/j.biortech.2010.06.029.

- Rawat I, Bhola V, Kumar RR, Bux F. Improving the feasibility of producing biofuels from microalgae using wastewater. *Environ Technol*. 2013;34:1765-1775. doi:10.1080/09593330.2013.826287.
- Cai T, Park SY, Li Y. Nutrient recovery from wastewater streams by microalgae: Status and prospects. *Renew Sustain Energy Rev.* 2013;19. doi:10.1016/j.rser.2012.11.030.
- Park JBK, Craggs RJ, Shilton a. N. Wastewater treatment high rate algal ponds for biofuel production. *Bioresour Technol*. 2011;102(1):35-42. doi:10.1016/j.biortech.2010.06.158.
- Mittal GS. Treatment of wastewater from abattoirs before land application A review. *Bioresour Technol*. 2006;97(9):1119-1135. doi:10.1016/j.biortech.2004.11.021.
- 64. Markou G, Vandamme D, Muylaert K. Microalgal and cyanobacterial cultivation: The supply of nutrients. *Water Res.* 2014;65:186-202. doi:10.1016/j.watres.2014.07.025.
- Sohsalam P, Englande AJ, Sirianuntapiboon S. Seafood wastewater treatment in constructed wetland: Tropical case. *Bioresour Technol*. 2008;99(5):1218-1224. doi:10.1016/j.biortech.2007.02.014.
- 66. 203\_diwtitmpiar.pdf.
- Jais NM, Mohamed RMSR, Al-Gheethi AA, Hashim MKA. The dual roles of phycoremediation of wet market wastewater for nutrients and heavy metals removal and microalgae biomass production. *Clean Technol Environ Policy*. 2016:1-16. doi:10.1007/s10098-016-1235-7.
- 68. Seckbach J. The Algae World.; 2015. doi:10.1007/978-94-017-7321-8.
- 69. Meriano M, Eyles N, Howard KWF. Hydrogeological impacts of road salt from Canada's busiest highway on a Lake Ontario watershed (Frenchman's Bay) and

lagoon, City of Pickering. *J Contam Hydrol*. 2009;107(1-2):66-81. doi:10.1016/j.jconhyd.2009.04.002.

- B??ckstr??m M, Karlsson S, B??ckman L, Folkeson L, Lind B. Mobilisation of heavy metals by deicing salts in a roadside environment. *Water Res*. 2004;38(3):720-732. doi:10.1016/j.watres.2003.11.006.
- Huber M, Hilbig H, Badenberg SC, et al. The influence of temperature and NaCl on cadmium, copper and zinc partitioning among suspended particulate and dissolved phases in an urban river. *Water Res.* 2012;46(8):521-532. doi:10.1016/j.watres.2015.04.019.
- 72. Tixier G, Rochfort Q, Grapentine L, Marsalek J, Lafont M. Spatial and seasonal toxicity in a stormwater management facility: Evidence obtained by adapting an integrated sediment quality assessment approach. *Water Res.* 2012;46(20):6671-6682. doi:10.1016/j.watres.2011.12.031.
- Interlandi SJ, Crockett CS. Recent water quality trends in the Schuylkill River, Pennsylvania, USA: A preliminary assessment of the relative influences of climate, river discharge and suburban development. *Water Res.* 2003;37(8):1737-1748. doi:10.1016/S0043-1354(02)00574-2.
- Wu J, Stewart TW, Thompson JR, et al. Optimum winter road maintenance : effect of pavement types on snow melting performance of road salts. *Sci Total Environ*. 2015;18(4):219-229. doi:10.1016/j.scitotenv.2015.06.034.
- 75. Green SM, Machin R, Cresser MS, Petranka JW, Doyle EJ. Effects of road salts on the composition of seasonal pond communities: Can the use of road salts enhance mosquito recruitment? *Aquat Ecol.* 2009;153(1):435-448. doi:10.1007/s10452-009-9286-z.
- Kelting DL, Laxson CL, Yerger EC. Regional analysis of the effect of paved roads on sodium and chloride in lakes. *Water Res.* 2012;46(8):2749-2758. doi:10.1016/j.watres.2012.02.032.

- Judd JH. Lake stratification caused by runoff from street deicing. *Water Res.* 1970;4(8):521-532. doi:10.1016/0043-1354(70)90002-3.
- 78. Werner E, DiPretoro RS. Rise and fall of road salt contamination of water-supply springs. *Environ Geol.* 2006;51(4):537-543. doi:10.1007/s00254-006-0350-x.
- 79. Sun H, Alexander J, Gove B, et al. Mobilization of arsenic, lead, and mercury under conditions of sea water intrusion and road deicing salt application. *Algal Res.* 2016;16:317-329. doi:10.1016/j.jconhyd.2015.07.002.
- Saxena P, Bassi A. Removal of nutrients from hydroponic greenhouse effluent by alkali precipitation and algae cultivation method. *J Chem Technol Biotechnol*. 2013;88(5):858-863. doi:10.1002/jctb.3912.
- Park JBK, Craggs RJ, Sukias JPS. Treatment of hydroponic wastewater by denitrification filters using plant prunings as the organic carbon source. *Bioresour Technol.* 2008;99(8):2711-2716. doi:10.1016/j.biortech.2007.07.009.
- Suzuki K, Tanaka Y, Kuroda K, et al. Removal and recovery of phosphorous from swine wastewater by demonstration crystallization reactor and struvite accumulation device. *Bioresour Technol.* 2007;98(8):1573-1578. doi:10.1016/j.biortech.2006.06.008.
- Suzuki K, Tanaka Y, Kuroda K, Hanajima D, Fukumoto Y. Recovery of phosphorous from swine wastewater through crystallization. *Bioresour Technol*. 2005;96(14):1544-1550. doi:10.1016/j.biortech.2004.12.017.
- Kozminykh P, Heistad A, Ratnaweera HC, Todt D. Impact of organic polyelectrolytes on coagulation of source-separated black water. *Environ Technol (United Kingdom)*. 2016;3330(October):1-10. doi:10.1080/09593330.2015.1130175.
- 85. Mitra M, Shah F, Bharadwaj SVV, Patidar SK, Mishra S. Cultivation of Nannochloropsis oceanica biomass rich in eicosapentaenoic acid utilizing wastewater as nutrient resource. *Bioresour Technol*. 2016;218:1178-1186.

doi:10.1016/j.biortech.2016.07.083.

- Pacheco MM, Hoeltz M, Moraes MS a, et al. Best practices in heterotrophic highcell-density microalgal processes: Achievements, potential and possible limitations. *Appl Microbiol Biotechnol*. 2011;91(6):585-601. doi:10.1080/10934529.2015.994951.
- Singh V, Tiwari A, Das M. Phyco-remediation of industrial waste-water and flue gases with algal-diesel engenderment from micro-algae: A review. *Fuel*. 2016;173:90-97. doi:10.1016/j.fuel.2016.01.031.
- Ebeling JM, Timmons MB, Bisogni JJ. Engineering analysis of the stoichiometry of photoautotrophic, autotrophic, and heterotrophic removal of ammonia-nitrogen in aquaculture systems. *Aquaculture*. 2006;257(1-4):346-358. doi:10.1016/j.aquaculture.2006.03.019.
- Bartley ML, Boeing WJ, Corcoran A a., Holguin FO, Schaub T. Effects of salinity on growth and lipid accumulation of biofuel microalga Nannochloropsis salina and invading organisms. *Biomass and Bioenergy*. 2013;54:83-88. doi:10.1016/j.biombioe.2013.03.026.
- Prakash Rai M, Gautom T, Sharma N. Effect of Salinity, pH, Light Intensity on Growth and Lipid Production of Microalgae for Bioenergy Application. *Online J Biol Sci.* 2015. doi:10.3844/ojbsci.2015.260.267.
- Novovesk L, Zapata AKM, Zabolotney JB, Atwood MC, Sundstrom ER.
   Optimizing microalgae cultivation and wastewater treatment in large-scale offshore photobioreactors. *Algal Res.* 2016;18. doi:10.1016/j.algal.2016.05.033.
- 92. Zheng Y, Li T, Yu X, Bates PD, Dong T, Chen S. High-density fed-batch culture of a thermotolerant microalga chlorella sorokiniana for biofuel production. *Appl Energy*. 2013;108:281-287. doi:10.1016/j.apenergy.2013.02.059.
- 93. Chen Y, Tang X, Kapoore RV, Xu C, Vaidyanathan S. Influence of nutrient status on the accumulation of biomass and lipid in Nannochloropsis salina and Dunaliella

salina. *Energy Convers Manag*. 2015;106:62-72. doi:10.1016/j.enconman.2015.09.025.

- 94. Su Y, Mennerich A, Urban B. Comparison of nutrient removal capacity and biomass settleability of four high-potential microalgal species. *Bioresour Technol*. 2012;124:157-162. doi:10.1016/j.biortech.2012.08.037.
- Ma XN, Chen TP, Yang B, Liu J, Chen F. Lipid production from Nannochloropsis. *Mar Drugs*. 2016;14(4). doi:10.3390/md14040061.
- 96. Tan F, Wang Z, Zhouyang S, et al. Nitrogen and phosphorus removal coupled with carbohydrate production by five microalgae cultures cultivated in biogas slurry. *Bioresour Technol.* 2016;221(422):385-393. doi:10.1016/j.biortech.2016.09.030.
- 97. Hsieh C-H, Wu W-T. Cultivation of microalgae for oil production with a cultivation strategy of urea limitation. *Bioresour Technol*. 2009;100(17):3921-3926. doi:10.1016/j.biortech.2009.03.019.
- da Silva Ferreira V, Sant'Anna C. Impact of culture conditions on the chlorophyll content of microalgae for biotechnological applications. *World J Microbiol Biotechnol.* 2017;33(1):20. doi:10.1007/s11274-016-2181-6.
- Wilson C, Qiu X, Stephenson PG, et al. Photosynthetic pigments in diatoms. *Mar Drugs*. 2015;13(9):5847-5881. doi:10.3390/md13095847.
- 100. Li Y, Han D, Sommerfeld M, Hu Q. Photosynthetic carbon partitioning and lipid production in the oleaginous microalga Pseudochlorococcum sp. (Chlorophyceae) under nitrogen-limited conditions. *Bioresour Technol.* 2011;102(1). doi:10.1016/j.biortech.2010.06.036.
- 101. Chisti Y. Constraints to commercialization of algal fuels. *J Biotechnol*. 2013;167(3). doi:10.1016/j.jbiotec.2013.07.020.
- 102. Wang S-K, Stiles AR, Guo C, Liu C-Z. Microalgae cultivation in photobioreactors: An overview of light characteristics. *Eng Life Sci.*

2014;14(6):550-559. doi:10.1002/elsc.201300170.

- 103. Zhu S, Huang W, Xu J, Wang Z, Xu J, Yuan Z. Metabolic changes of starch and lipid triggered by nitrogen starvation in the microalga Chlorella zofingiensis. *Bioresour Technol.* 2013;152:292-298. doi:10.1016/j.biortech.2013.10.092.
- Suali E, Sarbatly R. Conversion of microalgae to biofuel. *Renew Sustain Energy Rev.* 2012;16(6):4316-4342. doi:10.1016/j.rser.2012.03.047.
- 105. Cerón-García MC, Fernández-Sevilla JM, Sánchez-Mirón a., García-Camacho F, Contreras-Gómez a., Molina-Grima E. Mixotrophic growth of Phaeodactylum tricornutum on fructose and glycerol in fed-batch and semi-continuous modes. *Bioresour Technol.* 2013;147:569-576. doi:10.1016/j.biortech.2013.08.092.
- 106. Chen C-Y, Yeh K-L, Aisyah R, Lee D-J, Chang J-S. Cultivation, photobioreactor design and harvesting of microalgae for biodiesel production: a critical review. *Bioresour Technol.* 2011;102(1):71-81. doi:10.1016/j.biortech.2010.06.159.
- Chen CY, Yeh KL, Aisyah R, Lee DJ, Chang JS. Cultivation, photobioreactor design and harvesting of microalgae for biodiesel production: A critical review. *Bioresour Technol.* 2011;102(1):71-81. doi:10.1016/j.biortech.2010.06.159.
- 108. Suganya T, Varman M, Masjuki HH, Renganathan S. Macroalgae and microalgae as a potential source for commercial applications along with biofuels production: A biorefinery approach. *Renew Sustain Energy Rev.* 2016;55. doi:10.1016/j.rser.2015.11.026.
- 109. Pirt SJ. The theory of fed batch culture with reference to the penicillin fermentation. *J appl Chem Biotechnol*. 1974;24:415-424. doi:10.1002/jctb.5020240706.
- Vieira Costa JA, Colla LM, Duarte Filho PF. Improving Spirulina platensis biomass yield using a fed-batch process. *Bioresour Technol*. 2004;92:237-241. doi:10.1016/j.biortech.2003.09.013.

- 111. Liu J, Mukherjee J, Hawkes JJ, Wilkinson SJ. Optimization of lipid production for algal biodiesel in nitrogen stressed cells of Dunaliella salina using FTIR analysis. J Chem Technol Biotechnol. 2013;88(10):1807-1814. doi:10.1002/jctb.4027.
- 112. Xie Y, Jin Y, Zeng X, Chen J, Lu Y, Jing K. Fed-batch strategy for enhancing cell growth and C-phycocyanin production of Arthrospira (Spirulina) platensis under phototrophic cultivation. *Bioresour Technol.* 2015;180:281-287. doi:10.1016/j.biortech.2014.12.073.
- 113. García-Cañedo JC, Cristiani-Urbina E, Flores-Ortiz CM, Ponce-Noyola T, Esparza-García F, Cañizares-Villanueva RO. Batch and fed-batch culture of Scenedesmus incrassatulus: Effect over biomass, carotenoid profile and concentration, photosynthetic efficiency and non-photochemical quenching. *Algal Res.* 2016;13:41-52. doi:10.1016/j.algal.2015.11.013.
- 114. Anschau A, Xavier MC a, Hernalsteens S, Franco TT. Effect of feeding strategies on lipid production by Lipomyces starkeyi. *Bioresour Technol*. 2014;157:214-222. doi:10.1016/j.biortech.2014.01.104.
- 115. Sánchez-Luna LD, Converti A, Tonini GC, Sato S, De Carvalho JCM. Continuous and pulse feedings of urea as a nitrogen source in fed-batch cultivation of Spirulina platensis. *Aquac Eng.* 2004;31:237-245. doi:10.1016/j.aquaeng.2004.04.003.
- 116. Gerardo ML, Van Den Hende S, Vervaeren H, Coward T, Skill SC. Harvesting of microalgae within a biorefinery approach: A review of the developments and case studies from pilot-plants. *Algal Res.* 2015;11. doi:10.1016/j.algal.2015.06.019.
- Pragya Np, Pandey KK, Sahoo PK. A review on harvesting, oil extraction and biofuels production technologies from microalgae. *Renew Sustain Energy Rev.* 2013;24. doi:10.1016/j.rser.2013.03.034.
- 't Lam GP, Vermuë MH, Olivieri G, et al. Cationic polymers for successful flocculation of marine microalgae. *Bioresour Technol*. 2014;169:804-807. doi:10.1016/j.biortech.2014.07.070.

- Leite GB, Abdelaziz AEM, Hallenbeck PC. Algal biofuels: Challenges and opportunities. *Bioresour Technol*. 2013;145:134-141. doi:10.1016/j.biortech.2013.02.007.
- 120. Torres CM, Ríos SD, Torras C, Salvadó J, Mateo-Sanz JM, Jiménez L. Microalgae-based biodiesel: A multicriteria analysis of the production process using realistic scenarios. *Bioresour Technol*. 2013;147:7-16. doi:10.1016/j.biortech.2013.07.145.
- Uduman N, Qi Y, Danquah MK, Forde GM, Hoadley A. Dewatering of microalgal cultures: A major bottleneck to algae-based fuels. *J Renew Sustain Energy*. 2010;2(1). doi:10.1063/1.3294480.
- 122. Kim J, Yoo G, Lee H, et al. Methods of downstream processing for the production of biodiesel from microalgae. *Biotechnol Adv.* 2013;31(6):862-876. doi:10.1016/j.biotechadv.2013.04.006.
- 123. Lee J, Cho DH, Ramanan R, Kim BH, Oh HM, Kim HS. Microalgae-associated bacteria play a key role in the flocculation of Chlorella vulgaris. *Bioresour Technol.* 2013;131:195-201.
- 124. Dassey AJ, Theegala CS. Harvesting economics and strategies using centrifugation for cost effective separation of microalgae cells for biodiesel applications. *Bioresour Technol.* 2013;128:241-245. doi:10.1016/j.biortech.2012.10.061.
- Wicaksana F, Fane AG, Pongpairoj P, Field R. Microfiltration of algae (Chlorella sorokiniana): Critical flux, fouling and transmission. *J Memb Sci.* 2012;387-388(1):83-92. doi:10.1016/j.memsci.2011.10.013.
- Gerardo ML, Oatley-Radcliffe DL, Lovitt RW. Integration of membrane technology in microalgae biorefineries. *J Memb Sci.* 2014;464:86-99. doi:10.1016/j.memsci.2014.04.010.
- 127. Chen C-YCL, Yeh K-L, Aisyah R, et al. Harvesting microalgae grown on wastewater. *Bioresour Technol*. 2013;102(1):298-303.

doi:10.1016/j.biortech.2014.07.070.

- Rickman M, Pellegrino J, Davis R. Fouling phenomena during membrane filtration of microalgae. *J Memb Sci.* 2012;423-424:33-42. doi:10.1016/j.memsci.2012.07.013.
- Abdelaziz AEM, Leite GB, Hallenbeck PC. Addressing the challenges for sustainable production of algal biofuels: II. Harvesting and conversion to biofuels. *Environ Technol.* 2013;34(13-14):1807-1836. doi:10.1080/09593330.2013.831487.
- 130. Yakup N and WJO. 229\_tsohpa.pdf.
- Lei X, Chen Y, Shao Z, et al. Bioresource Technology Effective harvesting of the microalgae Chlorella vulgaris via flocculation – flotation with bioflocculant. 2015;198:922-925.
- 132. Besson A, Guiraud P. Bioresource Technology High-pH-induced flocculation flotation of the hypersaline microalga Dunaliella salina. *Bioresour Technol*. 2013;147:464-470. doi:10.1016/j.biortech.2013.08.053.
- 133. Velasquez-Orta SB, Garcia-Estrada R, Monje-Ramirez I, Harvey A, Orta Ledesma MT. Microalgae harvesting using ozoflotation: Effect on lipid and FAME recoveries. *Biomass and Bioenergy*. 2014;70:356-363. doi:10.1016/j.biombioe.2014.08.022.
- 134. Ndikubwimana T, Chang J, Xiao Z, et al. Flotation: A promising microalgae harvesting and dewatering technology for biofuels production. *Biotechnol J*. 2016;11(3):315-326. doi:10.1002/biot.201500175.
- Kurniawati HA, Ismadji S, Liu JC. Microalgae harvesting by flotation using natural saponin and chitosan. *Bioresour Technol*. 2014;166:429-434. doi:10.1016/j.biortech.2014.05.079.
- 136. Ravina L, Moramarco N. Everything You Want to Know about Coagulation &

Flocculation.; 1993.

http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:Everything+you +want+to+know+about+Coagulation+&+Flocculation+....#0.

- Vandamme D, Foubert I, Muylaert K. Flocculation as a low-cost method for harvesting microalgae for bulk biomass production. *Trends Biotechnol*. 2013;31(4):233-239. doi:10.1016/j.tibtech.2012.12.005.
- Bernhardt H, Clasen J. Flocculation of micro-organisms. *Aqua London*. 1991;40(2):76-87.
- Udom I, Zaribaf BH, Halfhide T, et al. Harvesting microalgae grown on wastewater. *Bioresour Technol*. 2013;139:101-106. doi:10.1016/j.biortech.2013.04.002.
- 140. Wyatt N.B, Gloe LM, Brady P V., et al. Critical conditions for ferric chlorideinduced flocculation of freshwater algae. *Biotechnol Bioeng*. 2012;109(2):493-501.
- 141. Biggs S, Habgood M, Jameson GJ, Yan Y De. Aggregate structures formed via a bridging flocculation mechanism. *Chem Eng J*. 2000;80(1-3):13-22.
- 142. Aljuboori AHR, Uemura Y, Thanh NT. Flocculation and mechanism of selfflocculating lipid producer microalga Scenedesmus quadricauda for biomass harvesting. *Biomass and Bioenergy*. 2016;93:38-42. doi:10.1016/j.biombioe.2016.06.013.
- 143. Brennan L, Owende P. Biofuels from microalgae-A review of technologies for production, processing, and extractions of biofuels and co-products. *Renew Sustain Energy Rev.* 2010;14(2):557-577. doi:10.1016/j.rser.2009.10.009.
- 144. Alam MA, Vandamme D, Chun W, et al. Bioflocculation as an innovative harvesting strategy for microalgae. *Rev Environ Sci Biotechnol*. 2016:1-11. doi:10.1007/s11157-016-9408-8.
- 145. Tenney MW, Echelberger WF, Schuessler RG, Pavoni JL. Algal Flocculation with

Synthetic Organic Polyelectrolytes. *Appl Microbiol*. 1969;18(6):965-971.

- 146. Alam MA, Vandamme D, Chun W, et al. Bioflocculation as an innovative harvesting strategy for microalgae. *Rev Environ Sci Biotechnol*. 2016:1-11. doi:10.1007/s11157-016-9408-8.
- 147. Roselet F, Vandamme D, Roselet M, Muylaert K, Abreu PC. Effects of pH, Salinity, Biomass Concentration, and Algal Organic Matter on Flocculant Efficiency of Synthetic Versus Natural Polymers for Harvesting Microalgae Biomass. *BioEnergy Res.* 2016. doi:10.1007/s12155-016-9806-3.
- Kandasamy G. Harvesting of the Microalga Nannochloropsis sp . by Bioflocculation with Mung Bean Protein Extract. *Appl Biochem Biotechnol*. 2016. doi:10.1007/s12010-016-2346-7.
- Renault F, Sancey B, Badot PM, Crini G. Chitosan for coagulation/flocculation processes - An eco-friendly approach. *Eur Polym J*. 2009;45(5):1337-1348.
- 150. Lavoie A, de la Noüe J. Harvesting Microalgae with Chitosan. J World Aquac Soc.
  1983;14(1-4):685-694. doi:10.1111/j.1749-7345.1983.tb00122.x.
- 151. Divakaran R, Pillai VNS. Flocculation of algae using chitosan. J Appl Phycol. 2002;14(5):419-422.
- 152. Chen L, Wang C, Wang W, Wei J. Optimal conditions of different flocculation methods for harvesting Scenedesmus sp. Cultivated in an open-pond system. *Bioresour Technol.* 2013;133:9-15.
- Christenson L, Sims R. Production and harvesting of microalgae for wastewater treatment, biofuels, and bioproducts. *Biotechnol Adv.* 2011;29(6):686-702. doi:10.1016/j.biotechadv.2011.05.015.
- 154. Vandamme D, Foubert I, Fraeye I, Meesschaert B, Muylaert K. Flocculation of Chlorella vulgaris induced by high pH: Role of magnesium and calcium and practical implications. *Bioresour Technol.* 2012;105:114-119.

doi:10.1016/j.biortech.2011.11.105.

- 155. Yang F, Xiang W, Fan J, Wu H, Li T, Long L. High pH-induced flocculation of marine Chlorella sp. for biofuel production. *J Appl Phycol.* 2016;28(2):747-756. doi:10.1007/s10811-015-0576-7.
- 156. Schlesinger A, Eisenstadt D, Bar-Gil A, Carmely H, Einbinder S, Gressel J. Inexpensive non-toxic flocculation of microalgae contradicts theories; overcoming a major hurdle to bulk algal production. *Biotechnol Adv.* 2012;30(5):1023-1030. doi:10.1016/j.biotechadv.2012.01.011.
- Clark SE, Pitt R. Targeting treatment technologies to address specific stormwater pollutants and numeric discharge limits. *Water Res.* 2012;46(20):6715-6730. doi:10.1016/j.watres.2012.07.009.
- 158. Harith, Z. T, Ariff, A. B, Yusoff, F. M, Mohamed, M. S, Shariff M, Din M. Effect of different flocculants on the flocculation performance of microalgae, Chaetoceros calcitrans, cells. *African J Biotechnol*. 2009;8(21):5971-5978.
- 159. Shin H, Kim K, Jung J, Bai SC, Chang YK, Han J. Harvesting of Scenedesmus obliquus cultivated in seawater using electro-flotation. 2017;34(3):62-65. doi:10.1007/s11814-016-0251-y.
- 160. Pérez L, Salgueiro JL, Maceiras R, Cancela Á, Sánchez Á. An effective method for harvesting of marine microalgae: pH induced flocculation. *Biomass and Bioenergy*. 2017;97:20-26. doi:10.1016/j.biombioe.2016.12.010.
- 161. Pérez L, Salgueiro JL, Maceiras R, Cancela Á, Sánchez Á. Study of influence of pH and salinity on combined flocculation of Chaetoceros gracilis microalgae. *Chem Eng J.* 2016;286:106-113. doi:10.1016/j.cej.2015.10.059.
- Salim S, Bosma R, Vermuë MH, Wijffels RH. Harvesting of microalgae by bioflocculation. *J Appl Phycol*. 2011;23(5):849-855.
- 163. Aljuboori AHR, Uemura Y, Thanh NT. Flocculation and mechanism of self-

flocculating lipid producer microalga Scenedesmus quadricauda for biomass harvesting. *Biomass and Bioenergy*. 2016;93:38-42. doi:10.1016/j.biombioe.2016.06.013.

- 164. Bhattacharya A, Mathur M, Kumar P, Prajapati SK, Malik A. A rapid method for fungal assisted algal flocculation: Critical parameters & mechanism insights. *Algal Res.* 2017;21:42-51. doi:10.1016/j.algal.2016.10.022.
- Zamalloa C, Gultom SO, Rajendran A, Hu B. Ionic effects on microalgae harvest via microalgae-fungi co-pelletization. *Biocatal Agric Biotechnol*. 2017;9(November 2016):145-155. doi:10.1016/j.bcab.2016.12.007.
- 166. Lee AK, Lewis DM, Ashman PJ. Microbial flocculation, a potentially low-cost harvesting technique for marine microalgae for the production of biodiesel. *J Appl Phycol.* 2009;21(5):559-567. doi:10.1007/s10811-008-9391-8.
- Luo L, Nguyen A V. A review of principles and applications of magnetic flocculation to separate ultrafine magnetic particles. *Sep Purif Technol.* 2017;172:85-99. doi:10.1016/j.seppur.2016.07.021.
- Wong YK, Ho YH, Leung HM, Ho KC, Yau YH, Yung KKL. Enhancement of Chlorella vulgaris harvesting via the electro-coagulation-flotation (ECF) method. 2016. doi:10.1007/s11356-016-7856-x.
- 169. Porra RJ, Thompson W a., Kriedemann PE. Determination of accurate extinction coefficients and simultaneous equations for assaying chlorophylls a and b extracted with four different solvents: verification of the concentration of chlorophyll standards by atomic absorption spectroscopy. *Biochim Biophys Acta -Bioenerg*. 1989;975(3):384-394. doi:10.1016/S0005-2728(89)80347-0.
- 170. Jena U, Vaidyanathan N, Chinnasamy S, Das KC. Evaluation of microalgae cultivation using recovered aqueous co-product from thermochemical liquefaction of algal biomass. *Bioresour Technol.* 2011;102(3):3380-3387. doi:10.1016/j.biortech.2010.09.111.

- Hara A, Radin NS. Lipid extraction of tissues with a low-toxicity solvent. *Anal Biochem.* 1978;90(1):420-426. doi:10.1016/0003-2697(78)90046-5.
- 172. Van Wychen S, Laurens LML. Determination of Total Lipids as Fatty Acid Methyl Esters (FAME) by in situ Transesterification - Laboratory Analytical Procedure (LAP) - http://www.nrel.gov/docs/fy14osti/60958.pdf. 2013;(December):NREL/TP-5100-60958.
- 173. Zheng H, Gao Z, Yin J, Tang X, Ji X, Huang H. Harvesting of microalgae by flocculation with poly (γ-glutamic acid). *Bioresour Technol*. 2012;112:212-220. doi:10.1016/j.biortech.2012.02.086.
- 174. Maclean C, Cembella A, Quilliam M. Effects of light, salinity and inorganic nitrogen on cell growth and spirolide production in the marine dinoflagellate *Alexandrium ostenfeldii* (Paulsen) Balech *et* Tangen. *Bot Mar*. 2003;46(Braarud 1945):466-476. http://www.degruyter.com/view/j/botm.2003.46.issue-5/bot.2003.048/bot.2003.048.xml.
- 175. Vandamme D, Foubert I, Meesschaert B, Muylaert K. Flocculation of microalgae using cationic starch. *J Appl Phycol*. 2010;22(4):525-530.
- 176. Noie J De, Lalibert G, Proulx D. Algae and waste water. 1992:247-254.
- 177. Chauton MS, Olsen Y, Vadstein O. Biomass production from the microalga Phaeodactylum tricornutum: Nutrient stress and chemical composition in exponential fed-batch cultures. *Biomass and Bioenergy*. 2013;58:87-94. doi:10.1016/j.biombioe.2013.10.004.
- Kiran B, Pathak K, Kumar R, Deshmukh D, Rani N. Influence of varying nitrogen levels on lipid accumulation in Chlorella sp. *Int J Environ Sci Technol*. 2016;13(7):1823-1832. doi:10.1007/s13762-016-1021-4.
- 179. Lubián L, Montero O, Moreno-Garrido I, et al. Nannochloropsis (Eustigmatophyceae) as source of commercially valuable pigments. *J Appl Phycol*. 2000;12:249-255. doi:10.1023/A:1008170915932.

- Rocha JMS, Garcia JEC, Henriques MHF. Growth aspects of the marine microalga Nannochloropsis gaditana. In: *Biomolecular Engineering*.Vol 20.; 2003:237-242. doi:10.1016/S1389-0344(03)00061-3.
- Varshney P, Mikulic P, Vonshak A, Beardall J, Wangikar PP. Extremophilic micro-algae and their potential contribution in biotechnology. *Bioresour Technol*. 2015;184. doi:10.1016/j.biortech.2014.11.040.
- 182. Chen Y-H, Walker TH. Biomass and lipid production of heterotrophic microalgae Chlorella protothecoides by using biodiesel-derived crude glycerol. *Biotechnol Lett.* 2011;33(10):1973-1983. doi:10.1007/s10529-011-0672-y.
- Lin Q, Gu N, Lin J. Effect of ferric ion on nitrogen consumption, biomass and oil accumulation of a Scenedesmus rubescens-like microalga. *Bioresour Technol*. 2012;112:242-247. doi:10.1016/j.biortech.2012.02.097.
- 184. Xie YP, Ho SH, Chen CY, et al. Simultaneous enhancement of CO2 fixation and lutein production with thermo-tolerant Desmodesmus sp. F51 using a repeated fedbatch cultivation strategy. *Biochem Eng J.* 2014;86:33-40. doi:10.1016/j.bej.2014.02.015.
- 185. da Silva Ferreira V, Sant'Anna C. Impact of culture conditions on the chlorophyll content of microalgae for biotechnological applications. *World J Microbiol Biotechnol.* 2017;33(1):20. doi:10.1007/s11274-016-2181-6.
- 186. Ajayan KV, Selvaraju M. Reflector based chlorophyll production by Spirulina platensis through energy save mode. *Bioresour Technol.* 2011;102(16):7591-7594. doi:10.1016/j.biortech.2011.05.013.
- PredatorRoberto, Aguilar C, Contreras-Esquivel JC, Méndez-Zavala A, Montañez J. Strategies to enhance the production of photosynthetic pigments and lipids in chlorophycae species. *Biotechnol Reports*. 2016;10:117-125. doi:10.1016/j.btre.2016.04.001.
- 188. Griffiths MJ, Van Hille RP, Harrison STL. The effect of degree and timing of

nitrogen limitation on lipid productivity in Chlorella vulgaris. *Appl Microbiol Biotechnol*. 2014;98:6147-6159. doi:10.1007/s00253-014-5757-9.

- 189. Morales-Sánchez D, Tinoco-Valencia R, Caro-Bermúdez M a., Martinez A. Culturing Neochloris oleoabundans microalga in a nitrogen-limited, heterotrophic fed-batch system to enhance lipid and carbohydrate accumulation. *Algal Res.* 2014;5:61-69. doi:10.1016/j.algal.2014.05.006.
- Strizh IG, Popova LG, Balnokin Y V. Physiological aspects of adaptation of the marine microalga Tetraselmis (Platymonas) viridis to various medium salinity. *Russ J Plant Physiol*. 2004;51(2):176-182. doi:10.1023/B:RUPP.0000019210.59579.6b.
- 191. Wang T-H, Chu S-H, Tsai Y-Y, Lin F-C, Lee W-C. Influence of inoculum cell density and carbon dioxide concentration on fed-batch cultivation of Nannochloropsis oculata. *Biomass and Bioenergy*. 2015;77:9-15. doi:10.1016/j.biombioe.2015.03.014.
- Hu H, Gao K. Response of growth and fatty acid compositions of Nannochloropsis sp. to environmental factors under elevated CO2 concentration. *Biotechnol Lett*. 2006;28(13):987-992. doi:10.1007/s10529-006-9026-6.
- 193. Frumento D, Casazza AA, Al Arni S, Converti A. Cultivation of Chlorella vulgaris in tubular photobioreactors: A lipid source for biodiesel production. *Biochem Eng* J. 2013;81:120-125. doi:10.1016/j.bej.2013.10.011.
- 194. Salama E, Abou-shanab RAI, Rae J, Lee S, Kim S, Oh S. The effects of salinity on the growth and biochemical properties of Chlamydomonas mexicana GU732420 cultivated in municipal wastewater. 2014;35(12):1491-1498.
- 195. N. Kalantaryan, V. Goginyan, L. Saghatelyan BH, Laboratory. Co mpo sition of fatt y acids synthe si ze d b y gree n mi cro algae. 2016.
- 196. Salama E-S, Abou-Shanab R a. I, Kim JR, et al. The effects of salinity on the growth and biochemical properties of Chlamydomonas mexicana GU732420

cultivated in municipal wastewater. *Environ Technol*. 2014;0(12):1-8. doi:10.1080/09593330.2013.871350.

- 197. Cheng J, Feng J, Sun J, Huang Y, Zhou J, Cen K. Enhancing the lipid content of the diatom Nitzschia sp. by 60Co-γ irradiation mutation and high-salinity domestication. *Energy*. 2014;78:9-15. doi:10.1016/j.energy.2014.06.009.
- 198. Cheirsilp B, Torpee S. Enhanced growth and lipid production of microalgae under mixotrophic culture condition: effect of light intensity, glucose concentration and fed-batch cultivation. *Bioresour Technol.* 2012;110:510-516. doi:10.1016/j.biortech.2012.01.125.
- Ma Y, Wang Z, Yu C, Yin Y, Zhou G. Evaluation of the potential of 9 Nannochloropsis strains for biodiesel production. *Bioresour Technol*. 2014;167:503-509.
- 200. Farooq W, Suh WI, Park MS, Yang JW. Water use and its recycling in microalgae cultivation for biofuel application. *Bioresour Technol*. 2015;184:73-81. doi:10.1016/j.biortech.2014.10.140.
- 201. Yeh KL, Chen CY, Chang JS. PH-stat photoheterotrophic cultivation of indigenous Chlorella vulgaris ESP-31 for biomass and lipid production using acetic acid as the carbon source. *Biochem Eng J.* 2012;64:1-7. doi:10.1016/j.bej.2012.02.006.
- 202. da Silva Ferreira V and CSA. The effect of physicochemical conditions and nutrient sources on maximizing the growth and lipid productivity of green microalgae. *Phycol Res.* 2016:3-13. doi:10.1111/pre.12160.
- 203. da Silva Ferreira V, Sant'Anna C. Impact of culture conditions on the chlorophyll content of microalgae for biotechnological applications. *World J Microbiol Biotechnol.* 2017;33(1):20. doi:10.1007/s11274-016-2181-6.
- 204. Liu J, Huang J, Sun Z, Zhong Y, Jiang Y, Chen F. Differential lipid and fatty acid profiles of photoautotrophic and heterotrophic Chlorella zofingiensis: Assessment

of algal oils for biodiesel production. *Bioresour Technol*. 2011;102(1):106-110. doi:10.1016/j.biortech.2010.06.017.

- 205. Sassano CEN, Carvalho JCM, Gioielli L a, Sato S, Torre P, Converti A. Kinetics and bioenergetics of Spirulina platensis cultivation by fed-batch addition of urea as nitrogen source. *Appl Biochem Biotechnol*. 2004;112:143-150. doi:10.1385/abab:112:3:143.
- 206. Batten D, Beer T, Freischmidt G, et al. Using wastewater and high-rate algal ponds for nutrient removal and the production of bioenergy and biofuels. *Water Sci Technol.* 2013;67(4):915-924. doi:10.2166/wst.2012.618.
- Yu N, Dieu LTJ, Harvey S, Lee DY. Optimization of process configuration and strain selection for microalgae-based biodiesel production. *Bioresour Technol*. 2015;193. doi:10.1016/j.biortech.2015.05.101.
- 208. Singh A, Nigam PS, Murphy JD. Mechanism and challenges in commercialisation of algal biofuels. *Bioresour Technol*. 2011;102(1):26-34. doi:10.1016/j.biortech.2010.06.057.
- Neck. B, De Pauw N, Jeurissen B. Potential of electrolytic flocculation for recovery of micro-algae. *Resour Conserv Recycl.* 1997;19(1):1-10.
- 210. Rashid N, Ur Rehman MS, Sadiq M, Mahmood T, Han J-I. Current status, issues and developments in microalgae derived biodiesel production. *Renew Sustain Energy Rev.* 2014;40:760-778. doi:10.1016/j.rser.2014.07.104.
- 211. Sharma AK, Sahoo PK, Singhal S, Joshi G. Exploration of upstream and downstream process for microwave assisted sustainable biodiesel production from microalgae Chlorella vulgaris. *Bioresour Technol.* 2016;216:793-800. doi:10.1016/j.biortech.2016.06.013.
- 212. Gross M, Henry W, Michael C, Wen Z. Development of a rotating algal biofilm growth system for attached microalgae growth with in situ biomass harvest. *Bioresour Technol.* 2013;150:195-201. doi:10.1016/j.biortech.2013.10.016.

- 213. Vandamme D, Pohl PI, Beuckels A, et al. Flocculation properties of several microalgae and a cyanobacterium species during ferric chloride, chitosan and alkaline flocculation. *Bioresour Technol.* 2016;13:271-283. doi:10.1016/j.biortech.2016.08.080.
- 214. Gerardo ML, Hende V Den, Vervaeren H, Coward T, Skill SC. Harvesting of microalgae within a biore fi nery approach : A review of the developments and case studies from pilot-plants. 2015;11:248-262. doi:10.1016/j.algal.2015.06.019.
- 215. Lama S, Muylaert K, Karki TB, Foubert I, Henderson RK, Vandamme D. Flocculation properties of several microalgae and a cyanobacterium species during ferric chloride, chitosan and alkaline flocculation. *Bioresour Technol*. 2016;220:464-470. doi:10.1016/j.biortech.2016.08.080.
- 216. Lee H, Nam K, Yang JW, Han JI, Chang YK. Synergistic interaction between metal ions in the sea salts and the extracellular polymeric substances for efficient microalgal harvesting. *Algal Res.* 2016;14:79-82. doi:10.1016/j.algal.2016.01.006.
- 217. Kurniawati HA, Ismadji S, Liu JC, et al. Cationic content effects of biodegradable amphoteric chitosan-based flocculants on the flocculation properties. *J Environ Sci* (*China*). 2013;24(8):1378-1385. doi:10.1016/j.procbio.2013.04.018.
- 218. Rashid N, Rehman M, Sadiq M, Mahmood T, Han J-I. Current status, isssues and developments in microalgae derived biodiesel production. *Renew Sustain Energy Rev.* 2014;40:760-778.
- Gerchman Y, Vasker B, Tavasi M, Mishael Y, Kinel-Tahan Y, Yehoshua Y.
   Effective harvesting of microalgae: Comparison of different polymeric flocculants.
   *Bioresour Technol.* 2017;228:141-146. doi:10.1016/j.biortech.2016.12.040.
- 220. Das P, Thaher MI, Abdul Hakim MAQM, Al-Jabri HMSJ, Alghasal GSHS. Microalgae harvesting by pH adjusted coagulation-flocculation, recycling of the coagulant and the growth media. *Bioresour Technol*. 2016;216:824-829. doi:10.1016/j.biortech.2016.06.014.
- 221. Wan C, Alm MA, Zhao XQ, et al. Current progress and future prospect of microalgal biomass harvest using various flocculation technologies. *Bioresour Technol.* 2015;184. doi:10.1016/j.biortech.2014.11.081.
- 222. García-Pérez JS, Beuckels A, Vandamme D, et al. Influence of magnesium concentration, biomass concentration and pH on flocculation of Chlorella vulgaris. *Algal Res.* 2014;3(1):24-29.
- 223. Ions M, Ayoub GM. High-pH-magnesium coagulation-flocculation in wastewater treatment. *Adv Environ Res.* 2003;7(2):389-403.
- 224. Vandamme D, Foubert I, Fraeye I, Meesschaert B, Muylaert K. Flocculation of Chlorella vulgaris induced by high pH: Role of magnesium and calcium and practical implications. *Bioresour Technol*. 2012;105:114-119.
- 225. Schlesinger A, Eisenstadt D, Bar-Gil A, Carmely H, Einbinder S, Gressel J. Inexpensive non-toxic flocculation of microalgae contradicts theories; overcoming a major hurdle to bulk algal production. *Biotechnol Adv.* 2012;30(5):1023-1030.
- 226. Gerde JA, Yao L, Lio J, Wen Z, Wang T. Microalgae flocculation: Impact of flocculant type, algae species and cell concentration. *Algal Res.* 2014;3(1). doi:10.1016/j.algal.2013.11.015.
- 227. Chen L, Li P, Liu Z, Jiao Q. The released polysaccharide of the cyanobacterium Aphanothece halophytica inhibits flocculation of the alga with ferric chloride. *J Appl Phycol.* 2009;21(3):327-331. doi:10.1007/s10811-008-9371-z.
- 228. Xu Y, Purton S, Baganz F. Chitosan flocculation to aid the harvesting of the microalga Chlorella sorokiniana. *Bioresour Technol*. 2013;129:296-301. doi:10.1016/j.biortech.2012.11.068.
- 229. Yusa M. Mechanisms of pelleting flocculation. *Int J Miner Process*. 1977;4(4):293-305.
- 230. Mowla D, Tran HN, Allen DG. A review of the properties of biosludge and its

relevance to enhanced dewatering processes. *Biomass and Bioenergy*. 2013;58:365-378. doi:10.1016/j.biombioe.2013.09.002.

- 231. Vandamme D, Pohl PI, Beuckels A, et al. Alkaline flocculation of Phaeodactylum tricornutum induced by brucite and calcite. *Bioresour Technol*. 2015;196:656-661.
- 232. Vandamme D, Muylaert K, Fraeye I, Foubert I. Floc characteristics of Chlorella vulgaris: Influence of flocculation mode and presence of organic matter. *Bioresour Technol.* 2014;151:383-387.
- 233. Şirin S, Trobajo R, Ibanez C, Salvadó J. Harvesting the microalgae Phaeodactylum tricornutum with polyaluminum chloride, aluminium sulphate, chitosan and alkalinity-induced flocculation. *J Appl Phycol*. 2012;24(5):1067-1080.
- 234. Zheng H, Gao Z, Yin J, Tang X, Ji X, Huang H. Harvesting of microalgae by flocculation with poly (??-glutamic acid). *Bioresour Technol*. 2012;112:212-220. doi:10.1016/j.biortech.2012.02.086.

## Appendices

# Appendix A Growth kinetics and nitrate uptake profile of *N. gaditana* in real vegetable greenhouse effluents



Element	2.6% Road salt (mg/L)	3.5% Road salt (mg/L)	4.4% Road salt (mg/L)	2.6% Reef salt (mg/L)	3.5% Reef salt (mg/L)	4.4% Reef salt (mg/L)	F/2 media (mg/L)
Na	3963.811	5335.900	6707.989	1655.309	2228.300	2801.291	4893.617
Mg	4.576	6.160	7.744	4721.971	6356.500	7991.029	4418.188
Al	0.250	0.336	0.422	0.131	0.176	0.221	0.507
Si	< 0.010	< 0.010	< 0.010	< 0.010	<0.010	< 0.010	
Р	0.016	0.022	0.027	0.002	0.003	0.004	
S	31.564	42.490	53.416	166.957	224.750	282.543	107.142
K	2.674	3.600	4.526	168.629	227.000	285.371	383.235
Ca	102.143	137.500	172.857	147.011	197.900	248.789	388.349
V	0.006	0.008	0.010	0.004	0.006	0.007	
Cr	0.001	0.002	0.002	0.001	0.001	0.001	
Mn	0.037	0.050	0.063	0.012	0.017	0.021	
Fe	<0.010	< 0.010	< 0.010	< 0.010	<0.010	<0.010	
Co	0.003	0.004	0.004	0.005	0.006	0.008	
Cu	0.071	0.095	0.119	0.028	0.038	0.048	
Zn	0.063	0.085	0.107	0.003	0.004	0.005	
As	0.374	0.503	0.632	0.022	0.030	0.037	
Sr	1.998	2.690	3.382	5.861	7.890	9.919	
Ba	0.040	0.054	0.068	0.008	0.011	0.014	

# Appendix B Elemental composition of the road and reef salt



# Appendix: C Experimental set up

1: air pump, 2 Moisturizing sterilized air, 3- stands to hold the reactors, 4 microalgae reactors 5) Fluorescent lights

# Appendix: D Photo of the jar experiment





## **Curriculum Vitae**

Name: Roopa Devasya

Post-secondaryUniversity of Agricultural SciencesEducation andBangalore, Karnataka, IndiaDegrees:1987-1992 B.Sc. Dairy TechnologyThe University of Agricultural SciencesUniversity of Agricultural Sciences

1993-1995 M.Sc. Dairy MicrobiologyThe University of Western SydneyHawkesbury, NSW, Australia1995-1998 M.Sc. Honours Food science (incomplete)

## Honours and Awards

1995-1998 AusAid scholarship

#### **Related Work Experience**

- 1) Teaching Assistant The University of Western Ontario- 2013-2016
- Research & Quality System Specialist- Parmalat R&D, London Ontario
  2007- Present
- Quality assurance technologist Cargill Kitchen solutions Etobicoke, Ontario-2005- 2007
- 4) Microbiologist Unilever best foods, Toronto, Ontario, 2003- 2007

## Conferences and presentations

Poster Presentation
 Phycoremediation of Road salt runoff
 Ontario-Quebec Biotechnology meeting-2014
 University of Toronto
 Oral Presentation "Phycoremediation of Road salt runoff using Marine microlagae"
 64th Canadian Chemical Engineering
 Conference Niagara Fall, ON October 19-22, 2014
 Oral Presentation "Phycoremediation of Road salt runoff using Marine microalgae"
 6th International Conference on Algal Biomass, Biofuels and Bio products

San Diego, United States

26–29 June 2016